

OUR LAND AND WATER

Toitü te Whenua, Toiora te Wai



The Matrix of Drivers: 2019 Update

Timothy Driver Professor Caroline Saunders Simon Duff John Saunders

Report for Our Land and Water National Science Challenge Agribusiness & Economics Research Unit (AERU), Lincoln University

Foreword

Enhancing primary sector production and productivity while maintaining and improving our land and water quality for future generations is a key outcome of the National Science Challenge for Our Land and Water. It is therefore important to identify the hierarchy of international and national issues in order to provide an evidence base to guide investment and inform the Challenge Research Strategy. To this end, it was proposed that a small project be conducted, and regularly updated.

This project aims to deliver an overview of international and domestic drivers, as well as issues that are of particular relevance to the New Zealand primary sector and land use. This overview is based on a literature search of the most important issues, followed by a survey of key stakeholders as to their opinion of the most important issues affecting New Zealand land use and land use practice from overseas and domestically. In addition, a review of the level of interest and concern of international consumers on various issues is produced relevant to the primary sector.

This is the third report in this series and provides an updated understanding of the international and national drivers and issues of land use change/practice, and their importance to the primary sector. These drivers will help prioritise where investments in primary sector research based on their relationship to economic growth, social, cultural and environmental interactions. Updates of this research will allow us to understand how drivers and issues change, which will help to assess the impact the Challenge has had as well as future research investment needs. This work also provides a contribution to the Challenge Strategy.

This report is structured as follows: Chapter 1 provides an introduction to this report and its wider context; Chapter 2 presents the results of a survey of primary sector stakeholders regarding their views of the importance of key international and domestic drivers of land use change/practice; Chapter 3 examines future trends and challenges related to land use change/practice (particularly within a New Zealand context); and Chapter 4 concludes the report and provides a summary of its findings.

1. Introduction

Project background

This report is the third in a series providing updated overviews of international and domestic drivers that have the potential to affect land use change and/or practice. This work has been undertaken in order to inform the strategic direction of the Our Land and Water (OLW) component of the National Science Challenge. The OLW challenge mission is to "enhance primary sector production and productivity while maintaining and improving our land and water quality for future generations." As different international and domestic drivers are likely to impact on New Zealand land use change and/or practice in a variety of ways, it is important to quantify to what extent this is likely to occur in order to prioritise key areas of focus for the Challenge.

To meet this requirement, this report presents an academic literature review of the latest research relevant to the international and domestic drivers of land use change and/or practice. The initial literature review undertaken in the first Drivers Project identified a preliminary list of 30 drivers (Saunders et al., 2016b). This was updated in late 2017 to include new arising issues or drivers relevant to land use change/practice (Saunders et al., 2018). The current list of international and domestic drivers is presented in Table 1.1 below. This report has expanded upon previous literature reviews, with an examination of the latest reports produced by key organisations such as the United Nations (including the FAO and IPCC), as well as key academic literature. A summary of each driver and its impact on land use change and/or practice (where possible) has been compiled, and can be accessed digitally by clicking on the links in Table 1.1 below. The updated evidence base used to inform these summaries is also available here.

Table 1.1. Current list of international and domestic drivers likely to impact on land use practice and/or change (as of December 2019)

Agricultural and Trade Policy	<u>Air Quality</u>	Animal Health and Welfare	Authenticity and Traceability		
<u>Biodiversity</u>	Biosecurity	Brand	Chemical Residues		
Climate Change	Country-of-Origin Cultural Value		<u>Demographics</u>		
<u>Digital</u> <u>Communication</u> <u>Systems</u>	Emissions Trading Schemes	Environmental Condition	<u>Extreme Weather</u> <u>Events</u>		
<u>Family and</u> <u>Community</u>	Food Safety	Functional Food	Gene Technology		
<u>Greenhouse Gas</u> (GHG) Emissions	Innovative Products	Local Food/Food <u>Miles</u>	Organic Production		
Pasture-Based Production	Precision Agriculture	Product Quality	Religion		
Social Responsibility and Fair Trade	<u>Soil Quality</u>	Sustainable Supply	Waste and Recycling		
	Water Footprinting and Use	Water Quality			

The literature review identified the key domestic and international drivers that have the potential to affect land use change and/or practice in New Zealand. The review also identified literature that demonstrated how these drivers may change over time drawing on trade modelling, consumer attitudes and behaviour research (see Dalziel et al., 2019).

The domestic drivers were informed by key strategic documents from government agencies such as The Ministry for Primary Industries (MPI), The Ministry for the Environment (MfE) and The Ministry of Foreign Affairs and Trade (MFAT). The strategic documents of regional and local agencies were also reviewed. Where publicly available, key information from sector groups and farmer associations such as Beef+Lamb New Zealand and Fonterra were also considered. Relevant academic literature was assessed. Important legislative and regulatory documents were also considered. This review included voluntary standards such as AsureQuality Organic standards and Sustainable Winegrowing New Zealand standards. International trade agreements, government legislation and reports, retailer requirements, strategic documents, and academic literature helped identify the international drivers. The literature review also looked at future trends that could influence these drivers.

The initial Drivers Project included a broad literature review of studies involving the use of methods such as choice experiments (CE) – an economic valuation method used to assess

willingness-to-pay (WTP) for different attributes of goods and/or services. Purchasing behaviour in markets is often influenced by product attributes such as price, quality and appearance, but also the credence attributes of a product. These are qualities that are not immediately seen or experienced during purchase or consumption, such as food safety, animal welfare, environmental protection, country-of-origin, and sustainability credentials. The CE method requires participants to make trade-offs between attributes by selecting one option from a series of products with multiple attributes, typically with an associated price attribute. This literature review has been updated to include recent CE and other WTP studies relevant to the drivers, covering academic literature published up to 2019. These can be found in Appendix A of this report.

2. New Zealand Primary Industry Stakeholder Survey

The overall aim of this project is to review and cross-reference national and international drivers in order to identify and prioritise areas of importance to the National Science Challenge. To assess the relative importance of the drivers across international regions, a survey addressing issues relating to the drivers was distributed to stakeholders involved in New Zealand's primary industries. In this report, the survey has been redesigned and updated, as presented below.

2.1 Survey methodology

As stated above, the two earlier Drivers reports included a survey of stakeholders. This was repeated for this report with an updated survey, administered in October/November 2019. The overall aim of this survey was to assess the relative importance of the drivers from New Zealand and international regions, with a particular focus on drivers' impact on land use practice/change in New Zealand. The survey was distributed on October 27th 2019 using Qualtrics[™], a web-based survey system. Two rounds of survey participation invitations were distributed – the first were specific invitations to a list of participants selected in consultation with the Science Challenge Directorate based on their experience and expertise in relation to New Zealand's primary industries (n=335). The second was to a list of participants from a database held by the Our Land and Water National Science Challenge (n=1,224). Additional reminder emails were sent following the initial distribution. The survey was distributed to 1,559 participants in total, receiving 226 completed surveys, thereby achieving a 14.5 per cent completion rate.

The survey first asked participants to indicate (unprompted) the three most critical international and domestic issues that they believed could influence New Zealand land use practice/change in a ranked order (e.g. 1 = most critical, in descending order of importance). These responses were then weighted (e.g. 1 (Most Critical) was given a weighting of 3; 3 (Least Critical) was given a weighting of 1) to provide scores of the overall importance of these international and domestic issues.

Participants were then asked to identify from a predetermined list of issues/drivers which of these were likely to have a 'high', 'medium' or 'low' impact on New Zealand land use change/practice. These predetermined drivers were chosen from previous Drivers reports, the literature, and in consultation with the Challenge Directorate.

Participants were also asked to identify their field of expertise and geographical region that they were most familiar with in relation to their work in New Zealand's primary industries. A copy of the survey instrument is in Appendix B of this report. Completed responses were then analysed and are given below.

2.2 Survey results

Survey participants were asked to identify the sector that they were most closely aligned with. As shown in Figure 2.1, 42 per cent of participants identified with the science/research sector, followed by government and other sectors (17 per cent and 10 per cent respectively). The most represented primary sector was dairy (10 per cent), followed by meat (9 per cent) and forestry (4 per cent). Sectors stated within the 'other' category included nutrient management, agribusiness and primary sector advocacy/consultancy, community, environmental management, irrigation, regenerative agriculture, water management, and food manufacturing, as well as work across multiple sectors.



Figure 2.1: Survey participants' alignment with sectors

Participants were also asked to indicate their levels of knowledge regarding particular markets and regions. As shown in Figure 2.2 below, 32 per cent of participants indicated they were 'very knowledgeable' or 'knowledgeable' regarding the United Kingdom market, followed by other markets (30 per cent 'very knowledgeable' or 'knowledgeable) and European Union (28 per cent 'very knowledgeable' or 'knowledageable'). Other markets/regions that participants identified as being familiar with included Australia, India, Latin America, Middle East, New Zealand, Pacific, Russia, South America and Southern Africa.



Figure 2.2: Participant's level of knowledge regarding markets/regions

■ Very knowledgeable ■ Knowledgeable ■ Some knowledge ■ Little knowledge ■ No knowledge

As shown in Figure 2.3 below, participants were also asked to indicate their level of experience in Environmental Policy, International Markets, R&D and Innovation, Trade Policy and 'Other Domestic'. Fifty-five per cent of participants had either 'extensive' or 'moderate' experience in *environmental policy*, followed by *R&D/Innovation* (47 per cent 'extensive' or 'moderate' experience) and *other domestic* (29 per cent 'extensive' or 'moderate' experience).

Figure 2.3: Participants' level of experience in industry fields



Critical international issues

Participants were then presented with an open-ended question that asked them to identify the three most critical domestic issues that would have the potential to influence land use change/practice in New Zealand. This was done to allow participants to identify important domestic issues without being prompted. As shown in Figure 2.4 below, *climate change* was indicated to be significantly more important to participants than any other international issue. Other critical issues identified included consumer preferences, greenhouse gas (GHG) emissions, trade/market access, alternative proteins, water quality/availability, geopolitical effects and changing dietary preferences. The results were consistent with previous survey results, with the main exception being an increase in the importance of climate change (Saunders et al., 2017).





Note: Issues with scores of less than 10 are omitted from this figure.

Critical domestic issues

The survey also asked participants to identify the three most critical international issues that could influence New Zealand land use change/practice. Like the previous question, this was done to allow participants to identify important international issues without being prompted. As shown in Figure 2.5 below, *water quality* was indicated to be important to more participants (followed by *climate change*) than any other domestic issues. Other critical issues identified included water policy, social license to operate, climate policy, government policy in general, GHG emissions and water availability. These results are consistent with the previous survey in which participants identified water-related issues as the most critical domestic issues (Saunders et al., 2018).





Note: Issues with scores of less than 10 are omitted from this figure.

Impact of international drivers/issues on New Zealand land use change/practice

Participants were then presented with a list of 38 international drivers (as identified by previous surveys and extensive literature review) and asked to indicate whether these would have a low, medium, or high impact on New Zealand land use change/practise over the coming decade.

Echoing prior unprompted statements, Figure 2.6 below shows that 83 per cent of respondents identified *climate change* as having a potentially high impact on New Zealand land use change/practice. This was followed by *condition of the environment* (68 per cent *high*, 28 per cent *medium*), *GHG emissions* (70 per cent *high*, 25 per cent *medium*) and water quality (69 per cent *high*, 26 per cent *medium*).

Climate change 3% Condition of the environment 200 4% Greenhouse gas emissions 5% 70% 25 Water quality 69% 269 5% Trade agreements 6% 64% 31% Trade effects 7% 55% Product quality 59% 8% Authentication/Traceability 8% Biosecurity 8% Food safety 9% 55% Sustainable supply 11% /7% 120 Animal health and welfare 11% 46% 170 Extreme weather events 12% Agricultural policy 46% 42% 13% Pasture-based production 35% 51% 13% Nitrate limits 59% 27% 14% Innovative products and services 32% 54% 14% Water footprinting/use 14%Emissions trading scheme (ETS) 16% 40% Country of origin 17% 33% 50% Brand 419 17% י42 **Biodiversity** 18% 22% Chemical residues 30% 20% Local foods/Food miles 50 25% Waste/Recycling 25% 22% 52% Soil quality 26% 31% 43 Functional foods 22% 31% GM and nanotechnology 26 31% Organic production 34% Fair trade 17% 36% 47% Demographics 15% 49% 36% Health and safety 18% 38% 449 Family and community values 38% 14% 18% Digital communication systems 38% 18% **Cultural values** 10% 39% 51% Māori values 40% 13% 47% Air quality 75% 11% 149 Religion 2% 81% 0% 10% 30% 40% 50% 60% 70% 80% 90% 100% 20%

Figure 2.6: Impact of international drivers/issues on New Zealand land use change/practice

High Medium Low

Impact of domestic drivers/issues on New Zealand land use change/practice

Participants were then presented with a list of 38 domestic drivers (as identified by previous surveys and extensive literature review) and asked to identify whether these would have a high, medium or low impact on New Zealand land use change/practice.

As shown in Figure 2.7 below, 99 per cent of respondents indicated that *water quality* was either of high or medium importance in relation to New Zealand land use change/practice, followed by *nitrate limits* (83 per cent 'high', 15 per cent 'medium') and *condition of the environment* (85 per cent 'high', 12 per cent 'medium').

Figure 2.7: Impact of domestic drivers/issues on New Zealand land use change/practice

Nitrate limits					0 - / 0					0/01/0
				8	3%				1	<mark>5% 2%</mark>
Condition of the environment					85%				1	<mark>2% 2</mark> %
Greenhouse gas emissions				73%					24%	3%
Biosecurity				64%				32	2%	4%
Climate change				76%	6				20%	4%
Agricultural policy				77%	%				18%	4%
Emissions trading scheme (ETS)			52%					40%		8%
Extreme weather events			52%				3	8%		11%
Animal health and welfare		4	1%				47%			11%
Water footprinting/use			54%				3	3%		12%
Product quality			48%				39%			14%
Biodiversity		379	%			2	18%			15%
Trade effects			47%				35%		1	8%
Pasture-based production		379	6			45	5%		1	8%
Food safety		39	%			4	2%		1	8%
Sustainable supply		4	2%				38%		19	9%
Māori values		28%				52%			20)%
Soil quality		4	2%			3	8%		20)%
Trade agreements			49%				31%		21	%
Innovative products and services		28%				49%			229	%
Cultural values	2	2%			52%	6			26%	
Authentication/Traceability		26%			48	8%			26%	
Health and safety	2	2%			52%	,			26%	
Brand		28%			44	%			28%	
Waste/Recycling		24%			47%				29%	
Chemical residues	20	0%			48%				31%	
GM and nanotechnology	179	%		47	7%			3	6%	
Family and community values	18	%		4	5%			3	7%	
, Demographics	13%			47%				39	9%	
Digital communication systems	169	6		449	6			40)%	
Organic production	14%			46%				40	%	
Functional foods	9%			50%				41	%	
Country of origin	13%			43%				44%	0	
Local foods/Food miles	13%			39%				48%		
Fair trade	12%		349	6				55%		
Air quality	8%	19%					74%			
Religion	1 <mark>%8%</mark>					91%				
0 -	1				· · ·					

[■] High ■ Medium ■ Low

3. Future trends and challenges and their impact on New Zealand land use change/practice

The primary sector represents a large proportion of domestic land use and contributes heavily to the New Zealand economy. The value of primary exports reached \$42.7 billion in 2018, an 11.7 per cent increase over the previous year. By the end of 2019, it is expected that these exports will increase by 3.8% to \$44.3 billion (MPI, 2018). This was primarily driven by higher exports of dairy, meat and forestry products, especially into China (StatsNZ, 2019b). New Zealand's primary sector aims to continue to achieve high export returns while simultaneously addressing local and global trends and challenges. This chapter will examine key future trends and challenges that could have the potential to impact primary land use change/practice in New Zealand.

3.1 Climate change

1. Impacts on regional production

The impact of climate change is likely to vary between regions. A number of studies have shown that low-latitude countries' crop production is likely to be adversely affected by climate change, while countries at higher latitudes could experience either positive or negative impacts on production (FAO, 2017a; IPCC, 2019b). It is also shown that animal growth rates in primary production systems have declined in recent years, exacerbated by climatic conditions (IPCC, 2019a).

While the effects of climate change are most likely to be considerably negative, some studies have shown that climate change could marginally enhance primary production in colder climates, while higher concentrations of CO₂ could potentially boost aspects of primary production (Prentice, 2017; Wirehn, 2018). For example, satellite observations have observed increased rates of vegetation greening occurring in parts of Asia, Europe, South America, Central North America and Southeast Australia, caused in part by extended growing seasons and CO₂ fertilisation processes (IPCC, 2019a).

The effects of climate change, in combination with demographic shifts and population growth, are likely to shift arable production regions. This will include an expansion in land area for arid climate zones and contraction in land area for polar climate zones (IPCC, 2019a). In particular, the amount and availability of arable land in Africa, South America, India and Europe are likely to decrease, while arable land availability may increase in areas of Russia, China and the United States (Zhang and Cai, 2011; IPCC, 2019b). This may be brought about by shifts in the suitability of particular regions for growing particular crops, with crops that thrived in particular regions not able to adapt to climate change-induced environmental disruptions (IPCC, 2019a).

In a New Zealand context, primary producers are likely to need to develop climate change adaptation strategies to maintain the sustainability of their operations. In the New Zealand kiwifruit sector, for example, Cradock-Henry (2017) found multiple vulnerabilities to climate change adaptation, including the use of short-term reactive practices over long-term strategies. Similarly, Kalaugher et al. (2017) showed that New Zealand dairy productivity is likely to be negatively impacted by the effects of climate change without the implementation of adaptation strategies. New Zealand primary production is also likely to be affected by a

number of other impacts, including the likely increased demand for irrigation and fertiliser to maintain yields, as well as increased fire risk in the forestry sector (CCATWG, 2017).

Global shifts in agricultural yields could also affect market demand, supply and prices (Hsiang et al., 2017). In this sense, the onset of climate change could marginally improve returns for New Zealand's primary sector, with a decline in agricultural production overseas increasing demand for New Zealand products, as well as increasing commodity prices potentially benefitting New Zealand producers and exporters (NZAGRC, 2012; Saunders et al., 2009). In addition, for particular regions of New Zealand, climate change may produce higher productivity for particular crops through improvements in climatic conditions required for plant growth (Kenny, 2001). However, this may be offset by the increased incidence of extreme weather events, diseases and pests associated with climate change (expanded upon in the following sections).

2. Extreme weather events

Climate change is likely to significantly increase the frequency and severity of extreme weather events, potentially negatively affecting food security and significantly influencing land use change/practice. This includes increases in the frequency and intensity of events such as heatwaves, droughts, dust storms, precipitation, flooding and similar effects (IPCC, 2019a). As discussed above, extreme weather events are likely to cause reductions in total yield for many staple crops, thereby negatively affecting food security (IPCC, 2019b). Lesk et al. (2016) estimated that extreme weather events were responsible for approximately 9-10 per cent reductions in national cereal production losses internationally between 1964 and 2007, the frequency and intensity of which are likely to increase under climate change. Climate change-induced extreme weather events can also negatively affect food security and supply by disrupting food supply chains (IPCC, 2019a).

Climate change-induced extreme weather events are estimated to be already affecting New Zealand, at a conservative cost of approximately NZ\$840 million in total between 2007 and 2017 (Frame et al., 2018; Harrington et al., 2014). Changing weather patterns and increasing extreme weather events have also been identified as critical challenges in climate change adaptation for a number of New Zealand's primary industries (Cradock-Henry, 2017; HNZ, 2017). For example, the 2013 New Zealand drought, which impacted the primary sector, bringing about an estimated NZ\$1.3 billion in damages, has been suggested to be a direct result of anthropogenic climate change (Harrington et al., 2014; VUW, 2017). More frequent and intense extreme weather events are likely to increase in New Zealand as a result of climate change, which could impact on several primary industries, including meat, wool, arable, dairy, viticulture, horticulture and forestry (NZAGRC, 2012).

3. Higher biosecurity risks

Climate change-induced effects, such as changes in regional temperature and weather patterns, could affect the dispersal and spread of damaging pests and diseases, both internationally and domestically (IPCC, 2019a; NZAGRC, 2012). Historic losses to primary producers caused by pest or disease incidence have included direct losses of their products (e.g. crops and/or animals), or through the loss of income associated with declines in yield. On an international level, biosecurity risks affect subsistence farming and other activities that are relied upon for human survival. Climate change is likely to exacerbate biosecurity risks to primary production, thereby potentially greatly influencing land use change/practice internationally (FAO, 2008; IPCC, 2019a).

Climate change is likely to present challenges for maintaining biosecurity in New Zealand's primary sector. Specifically, changes in New Zealand's climate could provide ideal conditions for many pest species, including invasive and already-present "dormant" species, which could thrive and proliferate under these conditions (Aguilar et al., 2015; FAO, 2008; James, 2019; Kean et al., 2015; Roques, 2010). This also includes increasing prevalence of animal and plant diseases, which, in combination with the above pests, could significantly negatively affect ecosystems and primary production systems. New Zealand's ecosystems are particularly vulnerable to introduced pest or disease species, with outbreak discoveries such as Mycoplasma bovis in cattle-based industries and PSA-V in the kiwifruit industry exemplifying this (BNZ, 2019; Taunton, 2017). Biosecurity risks pose a clear threat to New Zealand's primary industry, and is highly likely to influence land use change/practice.

4. Ecosystems changes

Climate change is highly likely to negatively impact on the world's ecosystems. This is likely to be caused by changes in land, air and water temperature, shifting temperate zones, and other effects such as extreme weather events (IPCC, 2019a). As a result, critical ecosystems services such as pollination and natural predator control could be disrupted, having a dramatic effect on ecosystem health and stability (EPA, 2016; FAO, 2017a; IPCC, 2019a). In addition to services required for basic survival (e.g. oxygen generation), ecosystem services provide essential support for human primary production systems, such as pollination, nutrient cycling, water cycling, water purification, erosion regulation, pest and disease regulation, seed dispersal and climate regulation (CBD, 2008). Declines in ecosystem service provision could negatively affect crop yields, genetic variability, soil fertility, water quality and pasture production (Lorencová et al., 2013; Maes et al., 2016). Simultaneously, land use practices can impair ecosystem services, thereby undermining their sustainability (Bjorklund et al., 1999; Caride et al., 2012; Costanza et al., 1997; Lorencová et al., 2013).

In New Zealand, climate change is likely to have a significant impact on ecosystems and their services, altering temperate zones, annual/seasonal events and ecosystem functions (such as food webs). This includes New Zealand's unique native biodiversity as well as essential ecosystem services such as those listed above (CCATWG, 2017; Christie, 2014; DOC, 2011). The implementation of management systems that enhance native biodiversity, and therefore ecosystem services to primary production, are included in the strategic plans of New Zealand government entities, including the Department of Conservation and the Ministry for the Environment (Christie, 2014; Davis et al., 2016; MfE, 2017c, 2019b). Roberts et al. (2015) provide a review of estimates of the contribution of ecosystem services to New Zealand wellbeing, including specific services to agriculture such as pollination (approximately NZ\$4.5 billion annually). Similarly, New Zealand's land-based ecosystems and their services were preliminarily estimated to contribute NZ\$57 billion to human welfare in 2012 (Patterson and Cole, 2013). Adopting sustainable land use practices will be critical to mitigating the effects of climate change on ecosystems.

5. Social impacts

Climate change may also have impacts on population and social dynamics, migration and human health. It is likely that climate change will cause currently populous world regions to become unliveable, potentially leading to mass human migration. Estimates have shown that between 72 and 200 million people may be forced to migrate as a result of climate change-induced conditions rendering particular world regions uninhabitable for humans (IPCC, 2014; Rigaud et al., 2018). Population growth and increased migration leading to the requirement

for greater urbanisation can enhance temperature increases and extreme rainfall events in urban centres, contribute to GHG emissions from land use change, as well as act as carbon sinks by importing increasingly large intakes of carbon in the form of food, fibers and fuel (Churkina, 2016; IPCC, 2019a). Increased atmospheric CO₂ levels will also have a high likelihood of compromising the nutritional quality of crops, thereby influencing global human and animal nutrition (IPCC, 2019a). The social impacts of climate change may therefore impact on land use change/practice, as well as exacerbate existing climate change stressors. In New Zealand, the impacts are expected to lead to some movements of population, for example from vulnerable coastal zones, with possible increases in migration.

6. Paris Climate Agreement

In 2015, the Paris Climate Agreement was adopted by 195 countries, the first ever legally binding international climate agreement, entering into force on November 4th 2016. It was signed with countries agreeing on net zero emissions by 2050. A raft of measures were agreed upon, including attempting to limit global temperature increase to 1.5 degrees (European Commission, 2017). However, following a decision made by President Donald Trump upon taking office in 2017, the US recently notified the UN of their intention to withdraw from the Paris Climate Agreement, stating that the deal will place an "unfair economic burden" on the US (BBC, 2019a). At present, at least 55 countries responsible for approximately 55 per cent of global GHG emissions have implemented instruments to ratify the agreement (UN, 2019c).

New Zealand is a signatory to the Paris Agreement, and has committed to reducing its GHG emissions to 30 per cent below 2005 levels by 2030 (MfE, 2017a). As of December 2019, the Climate Action Tracker (2019), New Zealand's progress towards achieving GHG emissions in line with keeping global warming below the Paris Agreement's 1.5°C target are rated as "insufficient", with current rates of emissions likely to track towards 3°C of global warming (CAT, 2019). Legislative and policy instruments designed to limit New Zealand's total GHG emissions have been developed and implemented (discussed in the following Section 3.1.7).

7. New Zealand domestic policy/emissions trading scheme

New Zealand legislation addressing climate change includes the Climate Change Response Act (2002), which was developed in order to meet obligations under the United Nations Framework Convention on Climate Change and Kyoto Protocol. This has been recently amended by the Climate Change Response (Zero Carbon) Amendment Bill, which was passed into law in October 2019. This sets a target of reducing all greenhouse gas emissions (excluding biogenic methane emissions) to net zero by 2050, with biogenic methane emissions to be reduced to 10 per cent below 2017 levels by the year 2030 (MfE, 2019a). The legislation of emissions reduction targets will likely impact on land use in New Zealand (particularly primary production) by potentially requiring land-users to adapt practices to meet legal requirements.

Under the New Zealand Emissions Trading Scheme (ETS), all sectors *except agriculture* are obliged to surrender their New Zealand Units (NZUs) (equivalent to 1 tonne of CO₂ emissions) to the government, which can then be given by government to land users with carbon sequestration practices in place. As the New Zealand agriculture sector was responsible for approximately 48 per cent of New Zealand's GHG emissions in 2017, the New Zealand government is pursuing options for its inclusion within the ETS (MfE, 2017b, 2019c). In particular, the New Zealand Government introduced the Climate Change Response (Emissions Trading Reform) Amendment Bill to Parliament in October 2019, outlining legislation to include agriculture in New Zealand's ETS. Parliamentary consensus on the inclusion of

agriculture in the ETS was achieved in July 2019, with a select committee expected to begin in late 2019 (MfE, 2019d). The agricultural sector has until 2022 to show how it plans to achieve net zero emissions by 2050, otherwise it will go into the New Zealand ETS in 2025.

Similarly, in response to climate change and other growing environmental concerns, the New Zealand Government has implemented the One Billion Trees programme, which aims to plant one billion trees across New Zealand between 2018 and 2028. This is considered to be effective in carbon sequestration, as well as improving productivity from land-based production and water quality, enhancing biodiversity by providing natural habitats, mitigating erosion and providing alternative sources of income for landowners (CEDC, 2018; Forestry New Zealand, 2018). This is estimated to require between 230,000 and 430,000 hectares of trees to be planted to meet the one billion tree target, with foresters expected to plant 500 million trees based on current planting rates (CEDC, 2018). This programme will offer landowners direct grants, as well as partnership grants, for planting trees on their land, with higher per hectare rates paid for indigenous mixed plantings over exotic species (Forestry New Zealand, 2018). This has the potential to affect New Zealand land use practice/change by incentivising tree planting as an alternative land use.

The One Billion Trees programme intends to use afforestation as a means of sequestering carbon emissions, as well as enhance biodiversity and ecosystem services. However, afforestation using exotic tree species (i.e. non-native species) may also produce adverse effects for native biodiversity and ecosystem services by displacing native vegetation, altering water cycles and increasing fire risk (CEDC, 2018; Christie, 2014). The promotion of native afforestation is included in the One Billion Trees programme, with greater financial incentives paid to land owners/managers for mixed native planting over exotic species (CEDC, 2018; Forestry New Zealand, 2018).

Other policy instruments designed to address land use in relation to climate change include the proposed National Policy Statement for Highly Productive Land (NPSHPL) and the National Policy Statement on Freshwater Management (NPSFWM). The proposed NPSHPL was developed by Ministry for Primary Industries and Ministry for the Environment to address several pressing concerns regarding increasing urbanisation, particularly the encroachment of urban expansion into highly productive land areas. It is hoped that the implementation of the NPSHPL can allow for mechanisms under the RMA to protect highly productive land from inappropriate development and subdivision and ensure its availability and benefits are maintained. Public consultation and submissions closed on October 10th 2019, and its current status is unknown (MPI, 2019a).

Similarly, in 2014, the New Zealand government introduced the NPSFWM, which outlines requirements for effective freshwater use, including protections for water quality, public health, native biodiversity, sociocultural values and Tikanga Māori (MfE, 2019e). The NPSFWM was amended in 2017, with Government looking to make improvements to strengthen current requirements, particularly with regards to ecosystem and human health. An independent advisory panel is expected to produce a report providing consultation summaries, with additional impact analysis to be submitted to Government (MfE, 2019f).

8. Climate activism

Public concerns regarding the onset of climate change, particularly government response and the relationship between global societal institutions and increasing environmental degradation, have led to a growth in climate activism in recent years. This can be seen in the

rise of activist groups such as Fridays For Future (FFF) and Extinction Rebellion (XR). In particular, the FFF movement was initiated by Greta Thunberg, whose protest on government inaction on climate change became virally popular, inspiring a global movement of school students into climate activism (FFF, 2019). In addition, Extinction Rebellion was initiated in the United Kingdom in October 2018 in response to government and industry inaction on climate change (XR, 2019a). Both groups have New Zealand chapters on a local level that have participated in regular climate activism since their inception (FFF, 2019; Forrester, 2019; TVNZ, 2019; XR, 2019b). Increases in climate activism could lead to greater public awareness of the impact of land use practice on environmental and climatic conditions, thereby potentially affecting land users' social license to operate (see Section 3.2.4 below). These movements could also place pressure on governments to enforce stricter regulations on activities likely to exacerbate climate change. This could ultimately affect land use management practices, leading to land use change.

3.2 Global trends and challenges

1. Growing global population, rising incomes and dietary changes

The World Economic Forum (WEF) outlined that achieving inclusive, sustainable and efficient global food systems would require addressing the challenges and opportunities presented by major global trends such as demographic shifts, macroeconomic trends, the triple burden of malnutrition (undernourishment, micronutrient deficiencies and over nutrition), natural resource depletion and geopolitical dynamics (WEF, 2016).

The UNEDSA's 2019 report 'World Population Prospects' describes two demographic megatrends (*population growth* and *ageing*), as well key trends in human fertility, mortality, and net international migration. This shows that global population growth had continued to increase, albeit at a slowing rate. In the medium fertility projection, by 2100 the global population is projected to reach approximately 10.9 billion, with an annual growth rate of 0.1 per cent – a considerable decline from the current rate (see Figure 3.1 below). In addition, the global fertility rate is expected to be 1.9 births per woman by 2100, down from 2.5 today (UN, 2019a).



Figure 3.1. Global population size and annual growth rate: estimates, 1950-2020, and medium-variant projection with 95 per cent prediction intervals, 2020-2100.

Source: UN, 2019a.

It is anticipated that Africa will account for most of the growth of the world's population over the coming decades. Between 2020 and 2100, Africa's population is expected to increase from 1.3 to 4.3 billion. Projections have shown that the most increases will come predominately from sub-Saharan Africa, the population of which is expected to triple in size by 2100. European and Latin American populations are expected to decline by 2100. Populations within Asia are estimated to increase from 4.6 billion in 2020 to 5.3 billion in 2055, and then begin to decline. It is projected that India will surpass China as the most populous country by 2027 (UN, 2019a).

At the same time, economic growth has increased in major world regions. The United Nations (2019b) showed that economic growth had accelerated across 50 per cent of national economies in both 2017 and 2018. Developed economies expanded 2.2 per cent of GDP across both years, while unemployment rates in several developed countries e.g. United States increased. Both South and East Asia experienced growth, with GDP expanding by 5.6 and 5.8 per cent respectively in 2018. Global economic growth per annum was 3.1 per cent in 2018, and is expected to stay above 3 per cent through 2019. Similarly, PWC (2017) projected 130 per cent of cumulative global GDP growth between 2016 and 2050, with the size of the world economy expected to double by 2050 (PWC, 2017a). Global per capita incomes are expected to rise by 2.2 percent each year between 2005 and 2050, coupled with increasingly inequitable distribution (FAO, 2012). Similarly, emerging economies such as China and India are projected to continue to grow in terms of per-capita income. Economic growth coupled with rising consumer purchasing power means larger middle- and upper-class segments in these regions, with consumer preferences in these segments becoming more important.

Increasing incomes are associated with both increasing and decreasing food demand. International food demand is expected to increase up to 2050, likely driven by population growth over increasing incomes. The OECD-FAO's Agricultural Outlook 2019-2028 estimates that the demand for all agricultural commodities for feed, food, fuel and other uses will increase between 2019 and 2028. In particular, total global food use of pulses, roots and tubers is expected to grow at a rate of 1.9 per cent per annum, followed by sugar and vegetable oils (1.8 per cent per annum), animal products (1.7 per cent per annum) and cereals (1.2 per cent per annum). Increases in global food demand are expected to be driven by increases in population and income levels, as well as changing consumer preferences (OECD-FAO, 2019).

Global consumer demand for protein is currently experiencing high growth, which is likely to continue into the future. The OECD and FAO's Agricultural Outlook (2019) has estimated an increase in global meat consumption of 0.4 kg per capita over the next decade, driven by population and income growth, particularly in Asian and Latin American countries (OECD-FAO, 2019; Sogari et al., 2019). In addition, the FAO estimates that the average person will consume 45.3 kg of meat annually by 2030, up from 41.3 kg in 2015 (Bruinsma et al., 2015). The projected increase in meat consumption will likewise increase the environmental impacts of agriculture.

At the same time, intensive agriculture has contributed heavily to GHG emissions, natural resource degradation, biodiversity loss, and water scarcity. It is likely that increases in future demand will place even greater stress on agricultural land use, production processes and natural resources (FAO, 2017a; OECD-FAO, 2019).

2. Food waste

Meeting future global demand within environmental limits will likely require more than increasing existing food production levels. Future demand could also be supported by a reduction in food wastage and loss. The FAO has estimated that, on a global scale, approximately one-third of all food produced for human consumption is currently wasted, equating to approximately 1.3 billion tonnes of food annually (Corrado and Sala, 2018; FAO, 2015). Current per capita estimates of annual food waste range between 194 and 384 kg per person (Corrado and Sala, 2018). It was also estimated that a 25 per cent reduction in current rates of food waste could potentially provide food for approximately 870 million people (FAO, 2015). Internationally, food waste costs industrialised countries approximately US\$680 billion and developing countries approximately US\$310 billion annually (FAO, 2015). On a domestic level, it has been estimated that New Zealand consumers waste approximately NZ\$1.8 billion of food annually, with 94 per cent of New Zealanders stating that they have wasted food at some stage (Shaw, 2017).

Research has also confirmed that food wastage occurs at all stages of global supply chains, with the scope and nature of food waste differing considerably between regions and states (Roodhuyzen et al., 2017; Xue and Liu, 2019). For example, food waste per capita is far higher in Europe and North America than in Sub-Saharan Africa and South/Southeast Asia, with food waste often generated at the consumption stage of the supply chain in developed countries, whereas in developing countries it is at the harvest and post-harvest storage stage. Food waste has also been linked with a number of adverse environmental (GHG emissions), economic (cost of waste, supply chain inefficiencies) and social impacts (reduced labour productivity, reduced wages) (Roodhuyzen et al., 2017). Food waste poses considerable

challenges to food security, natural resource use, the natural environment and human health (Xue and Liu, 2019).

Meeting future global demand will not only require increasing existing levels of food production, but will also need to be supported by a reduction in food waste. In response, the United Nations have set a target within the Sustainable Development Goals (SDG 2 – Zero Hunger) of reducing food waste by 50 per cent globally by 2030, with the European Union and United States also indicating an intention to meet this target (UN, 2019b; Xue and Liu, 2019).

3. Commodity price fluctuations

The FAO has stated that food prices have been reasonably volatile over the past decade, with a number of peaks and troughs characterising price trends for core commodities tracked (Bellman and Hepburn, 2017). This has been attributed to strong demand for food and feed, extreme weather events, market speculation, declining stock-to-use ratios and expanding biofuel production, all combining to give rise to market shocks and price fluctuations (FAO, 2017b). Agricultural commodity prices are also linked to the energy market, with farm inputs, production, storage and transportation all influenced by energy prices, but also the link between biofuels and energy prices (Bellman and Hepburn, 2017).

The OECD and FAO's Agricultural Outlook (2019) anticipated that agricultural commodity prices are likely to remain relatively flat over the coming decade (OECD-FAO, 2019). However, these agencies have also predicted that there is a strong chance of at least one severe price fluctuation over the same period, as there is uncertainty regarding oil prices, yields, economic growth and the impacts of climate change (Bellman and Hepburn, 2017). Price spikes affect food security and negatively impact consumers, producers and countries (FAO, 2017b). Moreover, the IPCC (2019b) also argue that climate change-induced extreme weather events may also lead to increased food price spikes, which when coupled with reduced land availability may mean greater difficulty for market recovery from price shocks (IPCC, 2019b).

4. Social license to operate

Social license to operate (SLO) has been broadly defined as the ongoing acceptance or approval of an operation by local community stakeholders, consumers and the general population who are affected by it, and those who can affect its profitability. It could also be understood as a set of demands and expectations held by affected parties and broader civil society for how a business or sector should operate (Moffat et al., 2016).

There is now a growing consumer expectation and demand for transparency and information provisions in relation to primary production processes and practices. New Zealand's primary sector has been adversely affected by negative public opinion and trust, particularly with regards to public expectations around environmental and social issues. In this way, for example, the New Zealand dairy sector's SLO has been affected by poor public opinion and trust. In particular, perceptions of poor water quality, animal welfare and labour management have forced the industry to address and adopt new land use practices to rebuild public opinion and trust (DairyNZ, 2017). Another example is that the Ministry for Primary Industries (MPI) has recently outlined intentions to grow New Zealand's aquaculture sector into a NZ\$1 billion industry. However, it has been suggested that SLO may be a critical issue facing the development of this industry.

As primary production moves into areas proximal to large numbers of people, tensions between industry and communities could become more prominent, increasing the need for

building an effective and sustainable SLO (Baines and Edwards, 2018). SLO is therefore expected to play an important role into the future, potentially impacting domestic and international land use change/practice, especially as international consumers push industries to operate in a more environmentally and socially responsible manner.

A related issue is the increasing mistrust of information and affiliation with personal beliefs over facts has led to the development of a "post-truth" society. This term is commonly used in reference to political campaigns, but has been increasingly used to denote a mistrust of scientific information and its sources (Iyengar and Massey, 2018; Rose, 2017). This has been further exacerbated by the proliferation of "fake news" (i.e. intentionally incorrect information being presented and promoted as reputable to enhance ulterior motives/interests) via online sources including social media (Lazer et al., 2018). The promotion of misinformation can influence firms' SLO by actively encouraging poor public opinion and eroding trust, thereby potentially influencing land use change/practice in New Zealand.

5. Urban-Rural divide

Similarly, it has been suggested that there is an increasing social and knowledge gap between urban and rurally centred people, both internationally and domestically, referred to as the *urban-rural divide*. This concept includes the ratio of people living in urban and rural centres, urban dwellers' opinions of rural life and vice versa, gaps in knowledge of food production practices, media portrayal of primary production, relative access to information communication technology (ICT) infrastructure, and defined differences in personal and community wellbeing between urban and rural regions (Apatov et al., 2018; DIRG, 2017; Infometrics, 2019; UMR Research, 2017, 2019).

In New Zealand, the percentage of the total population based in urban centres was estimated to be approximately 87 per cent in 2018 (StatsNZ, 2019a). This considered, several studies have shown little difference between urban and rural New Zealanders in their attitudes towards each other. Research carried out by UMR Research showed that a higher percentage of urban people held positive views regarding rural people and primary production practices than negative views (55 per cent *positive*), with a higher percentage of urban people considering moving to a rural centre than vice versa. However (UMR Research, 2017, 2019). However, several gaps that define the urban-rural divide have been highlighted by researchers, including generally lower self-stated wellbeing and less access to ICT infrastructure in rural centres (DIRG, 2017; Infometrics, 2019). Relatively higher pressures placed on people in rural centres, including primary producers, could affect current land use practices, potentially leading to land use change. In addition, a lack of understanding of primary production by urban dwellers could influence primary producers' SLO, thereby influencing land use change.

6. Farmer wellbeing

It has been recognised that rural workers in New Zealand are subject to similar rates of mental health issues (such as anxiety and depression) as occur within the general urban population, with additional stressors unique to the farming environment and lifestyle (such as a lack of control of the environment, increased workload and financial burdens) (O'Hara, 2010; Scarlatti, 2017). Growing concerns over rural wellbeing in recent years have led to the development of multiple programmes aimed at improving mental health conditions for New Zealand farmers. This includes the Mental Health Foundation and FMG's FarmStrong initiative, the GoodYarn programme and DairyNZ's Wellbeing programmes (DairyNZ, 2019;

FarmStrong, 2019; GoodYarn, 2019). The FarmStrong initiative in particular has reported success in improving mental health outcomes for New Zealand farmers (Wyllie, 2019). In addition, Mortlock and Hunt (2008) found that farmers' wellbeing was enhanced by engaging in practices that improved the environmental sustainability of their operations, thereby influencing land use change/practice. The mental health and wellbeing of New Zealand primary producers could also potentially impact on land use change/practice by influencing decision-making processes such as stress-induced career changes by primary producers.

Similarly, one of the unique challenges faced by primary producers is the accumulation of debt. This is due, in part, to externalities faced by agricultural producers such as yield loss from extreme weather events, coupled with vulnerability to land price changes and increasing environmental policy enforcement (RBNZ, 2019). The Reserve Bank of New Zealand has recently valued New Zealand farmer debt at approximately NZ\$62.8 billion, with 35 per cent of farms in the dairy sector defined as "highly indebted farms" with debt equivalent to more than NZ\$35 per kilogram of milk solids produced annually (MoA, 2019; RBNZ, 2019). In response to increasing farmer debt, a law change known as the Farm Debt Mediation Bill has been proposed. If approved, this would effectively create a mandatory debt mediation scheme for farmers wherein creditors would be required to offer mediation to those who default on payments prior to taking enforcement action. It is expected that this will be entered into the legislature before the end of 2019 pending Select Committee approval, with farmers able to access debt mediation from October 2020 (MPI, 2019d).

7. Sustainable Development Goals (SDGs)

In 2015, the United Nations signed the 2030 Agenda for Sustainable Development, in which all UN Member States adopted a set of 17 Sustainable Development Goals (SDGs) (UNDP, 2019). These provide goals against which progress towards sustainable development can be measured, such as Zero Hunger (SDG 2), which aims to "end hunger, achieve food security and improved nutrition and promote sustainable agriculture". Progress towards achieving these goals is outlined in the UN's annual Sustainable Development Goals Report, the 2019 edition of which indicated that progress towards many SDGs is currently too slow (UN, 2019d). International governments are now increasingly integrating SDGs within policy instruments to achieve these goals. In the New Zealand context, policy focus for achieving SDGs has been placed on the provision of aid, particularly for small island developing nations in the Pacific (MFAT, N.D.).

8. Indigenous enterprise

On an international level, there has been some growth in the awareness of and participation in consumption that supports food sovereignty. This concept is linked with food decolonisation, or the process of recognising that indigenous peoples' cuisines have often been subordinated by external forces, thereby leading to a deficit of culturally appropriate food choices (Grey and Newman, 2018). This has led to the development of indigenous enterprises, particularly for food or other primary products, across a number of indigenous communities internationally (Brown, 2016; ILO, 2015; Qi, 2019).

In a New Zealand context, the growth and continued success of Māori enterprises has sometimes been referred to as *the Māori economy*, which is currently valued at approximately NZ\$50 billion (MFAT, 2018; RED, 2017). There is currently an estimated 1,200 Māori enterprises in New Zealand (in 2019), an approximate 30 per cent increase from 2011 numbers (918 Māori enterprises) (Figure.nz, 2019). Increasing Māori entrepreneurship and

business is being driven, in part, by a strong sense of cultural identity and competency, and promotion of Kaupapa Māori (philosophical principles underpinning Māori cultural identity) (Mika, 2018; Warren et al., 2018; Wood and Mika, 2018). This is also underpinned by an increasing interest in and use of Māori knowledge systems and frameworks, collectively known as Mātauranga Māori (Harmsworth and Awatere, 2016; Manaaki Whenua, 2019). The development and success of Māori agribusiness ventures is also being supported by two MPI programmes: the Māori Agribusiness Extension programme and the Māori Agribusiness Pathway to Increased Productivity programme (MPI, 2019b, 2019c).

3.3 Emerging technologies

Technological advancements are already impacting the global agricultural and food sector. Bio-innovation, gene editing, robotics, big data, artificial intelligence (AI) and machine learning will provide the agricultural sector with unparalleled access to data, which will help improve production processes, reduce waste and increase yields (FAO, 2019a; WEF, 2016). These emerging technologies, as well as advancements in urban and regenerative farming, could significantly affect land use change/practice in New Zealand (Trice, 2017).

1. Artificial intelligence (AI)

Al is a radical new technology that sees computer systems performing tasks and making decisions like humans. Al has been a core driver of industrial development and a crucial factor in promoting the adoption of technologies such as cloud computing, Big Data and blockchain (see Section 3.4.3) (Lu, 2019). The Al sector has experienced significant development over the past decade, and is expected to play an important role in the long-term sustainability of global agricultural systems, having a significant impact on production efficiency. Al will also allow processes to be automated and conducted remotely, which will aid in detection of risks and issues for agile and informed decision-making (Irimia, 2016). In relation to this, Smith (2018) identified a number of key areas where Al is believed to deliver value over time to the agricultural sector (see Figure 3.2 below).





Source: Smith, 2018.

Technology that uses AI condenses large data sources, comprising research, historical weather data, nutrient levels, crop health data, soil conditions and moisture levels to provide recommendations that will help enhance crop yields and improve land use practice (Irimia, 2016; Lu, 2019). For example, Artificial Neural Pathways (ANNs) are being used in agricultural intelligence applications, such as soil analysis and mapping, where non-invasive ground-detection radar imaging techniques are used to detect soil properties and collect signals from electromagnetic soil sensors. Analysis of the soil clay content data can determine which crop is suitable for growing in each plot (Lu, 2019). In addition, high speed variable rate planting equipment has already allowed farmers to gather technical data about harvest production and yield trends - it is expected that this data will be used as the foundation of new predictive algorithms in AI implementation (Trice, 2017). Furthermore, automated irrigation systems that incorporate AI to predict and assess soil conditions will likely increase yields and improve water usage.

The value of AI use in the agricultural sector was estimated to be US\$600 million in 2018, and expected to reach US\$2.6 billion by 2025. This is based on several identified growth factors, including growing demand for agricultural products, increased uptake of information management systems and technologies to enhance crop productivity, and increased government spending on agricultural technology initiatives. A number of commercial organisations have harnessed AI to deliver cutting-edge products and services to the agricultural sector, including precision herbicide deployment systems that claim to reduce chemical application by up to 80 per cent, and associated expenditure by up to 90 per cent (Claver, 2019). In addition, AI technology is currently being integrated into international primary production processes. The company uses cameras and software applications to learn and recognise dying plants and disease/pest incidence, which has already boosted productivity by between 2 and 4 per cent (McFarland, 2017).

In a New Zealand context, research has indicated that sectors with large labour forces and high technology uptake and use are most likely to benefit from AI technology. Meanwhile, sectors such as agriculture, with smaller comparative labour pools and relatively low technology penetration, can expect less direct benefit from AI-created labour efficiencies (AIFNZ, 2018).

2. Unmanned aerial vehicle technology (UAVs)

Over the past 10 years, there has been exponential growth in the development and use of Unmanned Aerial Vehicles (UAVs) in agriculture. The FAO has stated that the use of UAVs and connected analytics applications have real potential to address some of the most pressing concerns facing international agriculture. In addition, Goldman Sachs has predicted that the agricultural sector will become the second-largest user of UAVs in the world between 2018 and 2023 (Sylvester, 2018). The total value of UAV-based solutions in the agricultural sector was estimated at US\$32.4 billion in 2016 (Mazur et al., 2016).

UAVs are now widely accessible for crop supervision, soil and field analysis and health assessment purposes. Drones are able to be integrated at every stage of the crop life cycle – from soil analysis and seed planting to harvesting. Drones with hyperspectral, multispectral and/or thermal sensors are now able to identify physical deficiencies in-field (e.g. water deficiency), thereby enhancing the efficiency and sustainability of production processes. Drones are also currently being used to identify bacterial or fungal incidence on trees using

visible light (VIS) and near-infrared (NIR) light (Mazur et al., 2016). In addition to detecting physical deficiencies in-field, hyperspectral imaging technology is also capable of determining the characteristics of agricultural products, including possible defects, chemical composition and similar physical properties, thereby assisting in food safety and quality compliance (Ravikanth et al., 2017).

UAV technology is likely to influence land use change/practice in New Zealand as its capabilities expand into the future. Several commercial UAV companies have emerged in New Zealand, such as Agdrone, which provides agricultural spraying services in areas where helicopters cannot fly (Dudman, 2019). UAVs will also allow farming to become a highly datadriven industry, which will likely enhance productivity and yield for New Zealand farmers. In addition, the ease-of-use and low cost of UAVs allow for time-series analysis of crop development, which could help to improve crop management (Mazur et al., 2016).

3. Robotics and autonomous vehicles

Robotics and autonomous systems (RAS) are expected to have a significant impact on global industries, including the agricultural sector. It is envisaged that these technologies will improve agricultural land use practices, reduce time-consuming processes and increase crop yields. Autonomous vehicles use sensors and global positioning devices (GPS) to plot paths and generate accurate mapping data for precision seeding and yield monitoring (Ghobadpour et al., 2019).

In 2019, three such significant technologies emerged in North America. The first, DOT Power Platform, is an autonomous vehicle capable of multiple farming tasks including seed drilling and fertiliser spraying; the second, AutoCart, is a driverless tractor developed for grain harvesting; and the third, SmartCore, is a robot soil sampler developed for quick and efficient analysis of ground conditions (Bennett, 2019). Additionally, John Deere recently announced the development of a fully electric and autonomous tractor (Allen, 2019). As autonomous vehicle technology improves and prices drop, New Zealand could see a greater uptake in the technology, thereby influence land use change/practice.

4. Genetics

Gene-editing is a form of genetic engineering that has made significant progress in recent years. This technology is far more precise and reversible comparative to older techniques, with the ability to make single nucleotide changes that mimic natural base point mutations. Gene-editing technology includes systems such as TALEN and CRISPR, which are capable of making site-specific changes to DNA sequences through the use of proteins such as zinc-finger nucleases. Importantly, gene-editing can alter genes without requiring foreign DNA sequences, which could make it a more appealing form of genetic engineering from a consumer perspective (RSNZ, 2016).

Gene-editing is already being trialled in an agricultural context internationally. In the US, maize, soybean, sorghum and rice have been altered to achieve desirable traits. Researchers in China have also been using TALEN and CRISPR to modify the genes of crops and animals, with advancements allowing them to develop goats with longer coats and higher muscle content (RSNZ, 2016). Scientists have also used CRISPR to boost the immunity of the cacao plant to a virus that has significantly negatively affected West African crops, as well as developing a more resilient variety of banana to fend off funguses that were undermining the global commercial supply (Niiler, 2018).

Some researchers see a more radical future for plant genetics. Scientists from the United States Department of Agriculture (USDA) have proposed redesigning the process of photosynthesis using gene-editing techniques – for example, changing chlorophyll and CO₂ absorption rates could improve the overall growth rate and yield of crops (The Economist, 2017). In 2019, USDA researchers announced that they had achieved a modification of the genetic code for photosynthesis in tobacco plants that led to a 40 per cent increase in their growth rate compared with unmodified plants. However, approval for use in commercial settings is likely to be between 5 and 10 years away due to stringent vetting and regulatory processes for genetic modifications such as these (South et al., 2019; Temming, 2019).

In 2016, the Royal Society of New Zealand convened a multidisciplinary panel of experts to examine the implications of the use of gene-editing technologies for New Zealand. The report identified that gene-editing was already playing a significant role in global primary production and impacting on land use change/practice (RSNZ, 2016). In 2019, the Royal Society released a report outlining the need to overhaul gene-editing regulations in New Zealand. At present, no genetically modified plants can be grown outside of the laboratory, with any experiment requiring approval from the Environmental Protection Agency (EPA) (RSNZ, 2019).

Several research streams has assessed the use of gene-editing techniques on New Zealand primary production. Research conducted by AgResearch found that gene-edited ryegrass could grow up to 50 per cent faster, requiring less water, and ultimately causing cattle to emit 23 per cent less methane (Hogan, 2019). In addition, Plant and Food's Chief Scientist Richard Newcomb has outlined that gene-editing technology could help to create crop varieties that are more resilient to extreme weather events, enable native trees to be better protected, and help New Zealand to reach its 2050 carbon neutral and predator-free targets (Nicol-Williams, 2019). In this way, the use of gene-editing could impact on land use change/practice in New Zealand.

5. Urban farming

Urban farming is the practice of producing food and fibre products within and around cities. There has been increasing interest in urban farming as concerns regarding urban food security and consumer demand for local foods grows internationally (FAO, 2019b, 2019c; PwC, 2017b). There is also evidence to suggest an the existence of an emerging consumer segment interested in consuming urban agricultural products (Grebitus et al., 2017; Krikser et al., 2019; Specht et al., 2016).

An example of urban farming methods, vertical farming (VF), is an innovative production model that uses controlled indoor environments to improve the efficiency of farming. VF combines building design and farming within a high-rise building, often situation within cities. VF is not vulnerable to environmental disturbances and climate, reduces water use by approximately 95 per cent, and has lower emissions compared with traditional agriculture. This is important as supply chain disruptions, seasonality, production and price fluctuations can severely impact the availability of fresh produce in urban areas, meaning that millions of people may only access produce erratically or at a high cost (Kalantari et al., 2017; Pinstrup-Andersen, 2017).

Urban planners and agricultural leaders have argued that cities will need to produce food internally to meet increasing demand, with multilateral government talks convened to discuss VF development, identifying VF as integral to the long-term sustainability of urban populations. In addition, recent advances in greenhouse technologies such as hydroponics,

aeroponics, and aquaponics have enhanced the prospects of the VF concept (Al-Kodmany, 2018). These techniques tightly control water, sunlight and nutrient loads to generate high crop yields. VF still faces significant commercial and economic challenges but is likely to play a role in addressing urban production and consumption issues in the future (Pinstrup-Andersen, 2017).

In the New Zealand context, there has been an increase in interest in and development of urban food systems, but less vertical farming. For example, multiple urban agriculture systems are in operation in Christchurch, partially due to an increase availability of urban land following the 2010/11 earthquakes. Specific examples include Roimata Food Commons, Cultivate Christchurch and Ōtākaro Orchard (Frank Film, 2019; Ineson, 2017; McDonald, 2019).

6. Regenerative agriculture

The practice of "regenerative agriculture" has emerged in response to environmental and climate concerns associated with conventional farming practices. This term loosely includes practices that seek to enhance biodiversity, soil health and water quality, as well as sequestering carbon and boosting ecosystem services while also producing food and fibre products (Gosnell et al., 2019). In particular, increased uptake and use of regenerative organic agriculture techniques could sequester CO₂ to effectively limit global warming to 1.5°C, with several field trials showing significant reductions in CO₂ emissions from regenerative farms (Rodale Institute, 2013; Teague et al., 2016). Regenerative systems have also been shown to have higher yields, lower production costs, lower pest incidence and greater ecosystem service provision than their conventional counterparts (LaCanne and Lundgren, 2018).

Unlike systems such as organic farming, there is also no specific criteria or standards for regenerative agriculture, but rather a growing set of principles and practices (Chandra et al., 2017). There are also challenges associated with conversion from conventional to regenerative systems. In addition to adaptations in technical processes, management, innovation and policy, conversion to regenerative agriculture will also likely require a shift in values and cultural norms to accommodate changes in practice (Gosnell et al., 2019). Despite these challenges, regenerative agriculture is being taken up on an international level, with General Mills recently assessing the regenerate approximately 1 million acres of farmland in the US (Anzilotti, 2019; General Mills, 2019). In a New Zealand context, there has been increasing media coverage of agricultural production in transition to regenerative farming, with a growing awareness among farmers and growers (Eb, 2019; Merfield, 2019). Rapid adoption of or transition to regenerative agriculture would represent a clear shift in land use change/practice.

3.4 Innovative products – new food technology

1. Alternative protein sources

The heavy ecological impact of the traditional livestock industry is a major driver in the development and commercialisation of alternative protein products (Akhtar and Isman, 2018). These alternative sources can be placed into three main categories: edible insects, plant-based proteins and cellular or cultured proteins (Sexton et al., 2019).

Edible insects are becoming increasingly common. Entomophagy – the consumption of insects as food – is currently practised by around 2 billion people across Africa, Asia, Central and South

Africa, and Australia (Akhtar and Isman, 2018). While Western regions have typically consumed traditional livestock proteins, there is some evidence to suggest that insect protein consumption is increasing. Insect-based foods are also a healthy source of protein, fat, energy and fibre (Sogari et al., 2019). However, cultural attitudes regarding the palatability of insects may be a barrier to widespread consumption. In a study of New Zealand consumers' perceptions of and preferences for insect-based food products, approximately 67 per cent indicated that they would be most willing to consume insects as food if they were processed into a powder that could be added to existing foods. The researchers suggested that this indicates that an insect product with clearly detailed health and environmental benefits could be successful in New Zealand (Payne and Ryan, 2019).

A wide range of plant-based protein substitute products have entered the marketplace over the last decade. New plant-based protein products are available in supermarkets internationally, with growing interest in flexitarian, vegetarian and vegan diets likely to drive demand into the future, both domestically and internationally (RNZ, 2019b; Sakure and Manepalli, 2019). Examples of companies with an established presence for plant-based alternative protein products include Beyond Meat and Impossible Foods (Sexton et al., 2019). This market has exhibited significant growth in recent years, with Beyond Meat shares increasing from US\$25.00 in May to US\$234.90 per share in May 2019 (Reinicke, 2019).

The emergence of alternative proteins has been projected to decrease the global market share for traditional meat products into the future. One study has suggested that by 2040 the global market share for conventional meat products is likely to be 40 per cent (down from 90 per cent in 2025), followed by cultured meat products (35 per cent) and novel vegan meat replacement (25 per cent). Alternative protein production requires less land and water, as well as produces less GHG emissions compared with conventional meat production processes (ATKearney, 2019). These potential market shifts could therefore produce positive environmental outcomes relative to conventional practices.

In a New Zealand context, there has been an increase in demand for alternative protein products, with retailer Countdown stating that demand for plant-based vegan and vegetarian meals had increased by 36 per cent between 2018 and 2019 (RNZ, 2019a). There are now several New Zealand-based plant-based alternative protein companies, including Sunfed, Craft Meat Company and the Alternative Meat Company (AMC, 2019; CMC, 2019; Sunfed, 2019). Sunfed's 'Chicken-Free Chicken' can currently be found in 250 stores in the New Zealand market, and has recently expanded to the Australian market (Keall, 2019). Plant & Food Research has identified new New Zealand consumer trends regarding plant-based foods, plant proteins and flexitarian lifestyle. They also suggest ways in which ingredients for these products could be incorporated into existing New Zealand primary production systems. Thus it is estimated that New Zealand currently has approximately 1.74 million hectares of land suitable for growing plant protein crops (Sutton et al., 2019).

2. Cultured proteins

Similar to alternative proteins, the development and future commercialisation of cultured protein products (also known as synthetic proteins) has the potential to influence land use change/practice. This process involves the in vitro propagation of animal tissue cells intended for human consumption in the place of flesh from animal carcasses (Datar and Betti, 2010). This method of production may have environmental benefits over traditional meat

production, with meat produced in this manner believed to generate 90 per cent lower GHG emissions relatively (Memphis Meats, 2017).

In 2013, the first cell-cultured meat patty was produced at Maastricht University, reportedly costing €250,000 to produce (Burton, 2019). Following this, Memphis Meats developed the first cell-cultured meatball and chicken strips from animal cells in 2017 (The Economist, 2017). At the time of writing, there are 31 companies seeking to become the first to commercialise and market synthetic animal protein products (Burton, 2019). At present, no company has been successful in bringing these products to the market for a number of reasons, including ethical and consumer acceptance uncertainties, as well as a lack of institutional, political and regulatory systems to support synthetic meat commercialisation (Stephens et al., 2018). In addition, there is an ongoing debate regarding the naming of such products (i.e. *laboratory*, artificial, cell-based and/or cultivated as potential nomenclatures) (Shanker, 2019). Current estimates for the arrival of lab-grown meats on supermarket shelves range between 1-20 years, with preliminary studies showing that consumers are willing to try such products and, if satisfied with their experience, incorporate them into their regular diet (Heffernan, 2017). However, a synthesis of consumer research into lab-grown meat also suggests that consumers in the US and China are concerned about the safety of such products, which could influence their willingness to purchase (Antedote, 2018).

The New Zealand Treasury has stated that it will be monitoring the global synthetic protein market, concluding that while these products do not currently have the potential to significantly disrupt traditional meat markets, they could pose risks into the future (New Zealand Treasury, 2018). Similarly, New Zealand meat industry bodies have previously stated that they do not consider synthetic meat products to be a risk to the industry (RNZ, 2018). However, if a feasible international market for synthetic protein products were to emerge, this could influence land use change/practice in New Zealand.

3. Alternative dairy

Closely related to the above alternative protein products is the emergence of alternative dairy products. This includes alternatives to milk, butter, cheese, yoghurt and ice cream, based on plant ingredients, such as soy, nuts and other plant products. The global market for dairy alternatives was valued at US\$15.5 billion in 2017, and is expected to reach US\$38.9 billion by 2025 (FM, 2019). In 2018, alternative dairy products represented over 50 per cent of worldwide sales in the vegan alternative product category. Reflecting interest in alternative dairy products, global dairy company Danone has recently invested approximately US\$60 million in plant-based production to improve its overall product portfolio and compete in the alternative dairy category (ARC, 2019).

The production of alternative dairy products has also been shown to have lower environmental impacts compared with traditional dairy production. In particular, the production of ingredients and product processing for alternative dairy products has been shown to require significantly less water and land area, as well as producing significantly less GHG emissions compared with traditional dairy production (Guibourg and Briggs, 2019; Poore and Nemecek, 2018). In a New Zealand context, Fonterra has invested in Motif Ingredients, a US-based innovative foods developer involved in alternative milk production, to diversify their product portfolio (Fonterra, 2019). Growth in the alternative dairy market could potentially influence New Zealand land use change/practice by diversifying the dairy market and altering market shares for traditional dairy products.

3.5 Consumer Trends

1. Vegetarianism, veganism and consumer diet trends

Related to the above, vegetarianism and veganism have grown significantly over the last 20 years, overcoming scepticism and prejudice (Leitzmann, 2014). It was estimated that in 2018 approximately 73 per cent of international consumers identified as omnivorous, alongside flexitarian (occasional meat and fish consumption) (14 per cent), vegetarian (5 per cent), vegan (3 per cent) and pescetarian (3 per cent) (Ipsos, 2018). Vegetarian and vegan movements are influencing retailers, who are responding by increasingly stocking a wider range of alternative proteins and substitute products, with vegan foods growing ten times faster than all other food categories in the US in 2018 (The Economist, 2019). It is estimated that by 2026, the vegan food market will be worth approximately US\$24.3 billion, with a compound annual growth rate of 9.1 per cent between 2019 and 2026 (ARC, 2019).

In line with this trend is an increase in the number of consumers seeking to reduce their overall meat consumption, otherwise known as *flexitarians* or *meat reducers*. Adherents to this type of diet still eat meat and fish products, but make a conscious effort to reduce the amount of these products that they regularly consume. As mentioned above, flexitarians currently comprise 14 per cent of the global population (Ipsos, 2018). Nielsen (2019) found that, of those US consumer who had purchased meat alternatives, approximately 98 per cent also purchase meat products, with approximately 27 per cent purchasing meat alternatives five or more times per year.

Increasing interest in and adherence to vegetarian, vegan and flexitarian diets could have environmental implications. In particular, adherence to a diet that lowers overall consumption of animal-sourced foods has been shown to have environmental benefits, including significant reductions in food-related GHG emissions. In addition, a mass transition to these diets is likely to have global economic savings equivalent to between US\$1 and \$13 trillion (between 0.4 and 13 per cent of global gross GDP in 2050) in terms of both the cost of negative health impacts mitigated by dietary change and the value of output of those no longer affected by negative health (Springmann et al., 2016). This has been corroborated by a multitude of studies (Chai et al., 2019; Fresan and Sabate, 2019; Lynch et al., 2018; Rosi et al., 2017). Reductions in meat consumption (as in the flexitarian diet) has also been shown to be driven by consumer concerns regarding the environmental implications of traditional meat production (Nielsen, 2019). It is also likely that the promotion of alternative diets via social media platforms and meat reduction campaigns will drive growing consumer interest in these trends (Grassian, 2019; Puranen and Jansson, 2017).

The Keto (*ketogenic*) diet is a consumer diet trend that has increased in popularity in recent years. Keto is a low-carbohydrate, high-fat diet that is designed to stimulate *ketosis* – a metabolic state in which the human body burns fat for energy. The diet is purported to support several health benefits, including weight and fat loss, reduced blood sugar and insulin, increased brain health and acuity, and faster metabolism. As the diet suggests a restriction in foods high in carbohydrates, it encourages higher consumption of foods such as meat, fish, eggs, dairy, nuts and seeds, and low-carbohydrate vegetables (Healthline, 2018). This could influence land use change/practice by affecting shifts in consumer consumption of particular types of food products.

Intermittent fasting (IF) is another consumer diet trend that has grown in popularity in recent years. IF involves deliberating inhibiting caloric intake on between 1 and 3 days per week, or

only eating within a particular time window every day of the week (Barnosky et al., 2014; Healthline, 2017). One popular method of IF is known as the 16/8 method, whereby eating is restricted to an 8-hour window every day (e.g. between 12pm and 8pm). The diet specifies no restrictions on the type of foods that can be consumed, rather on the times at which food can be consumed. The diet is purported to have significant health benefits, including reductions in blood sugar and insulin levels, increases in human growth hormone, weight and fat loss, and metabolic benefits (Healthline, 2017). As adherence to this diet implies an overall restriction in the amount of calories consumed, more widespread consumer adoption could influence land use change/practice by reducing overall food demand in specific segments.

2. Products with credence attributes

An obvious contribution to consumer value are the physical qualities of the product itself, which in the case of food and beverage products include freshness, taste, texture and flavour. Another important contribution comes from qualities that cannot be seen or experienced at the point of purchase. These attributes are known as credence attributes.

Examples of credence attributes include food safety, environmental stewardship, animal welfare, social responsibility, cultural authenticity, fair trade, functional foods, organic production, GM-free, water footprint, biodiversity and local foods (Saunders et al, 2016b, p. 18). Sellers typically make claims about the credence attributes of their products on labels, perhaps reinforced by developing brands or trademarks that are trusted by consumers as assurance that claims are authentic.

A key factor that may contribute to consumer trust is the country-of-origin of the food or beverage being purchased. Indeed, country-of-origin labelling (COOL) is mandatory for at least some food products in the major countries importing from New Zealand such as the United States, China, the European Union and Australia (Miller et al, 2016a).

A number of studies have observed that COOL can support product differentiation (Carter et al, 2006) and so create a competitive advantage that is not easily copied (Baker and Ballington, 2002; FutureBrand, 2014 and 2015). In particular, country-of-origin may be used by consumers as a cue for judging attributes such as quality (Claret et al, 2012; Berry et al, 2015; Insch et al, 2015) and food safety (Cicia et al, 2011; Lim et al, 2014; Ortega et al, 2014; Lewis and Grebitus, 2016).

The New Zealand wine industry is an example of an entire land-based sector that promotes its sustainability credentials to create export value. In the year from 1 July 2016 to 30 June 2017, wine exports from New Zealand reached \$1.66 billion, making it the country's fifth largest export good (New Zealand Winegrowers, 2017a, pp. 2-3). The sector maintains a commitment to quality over quantity to protect New Zealand's reputation as a premium producer of wine, including a commitment to sustainability leadership (New Zealand Winegrowers, 2017b, p. 1):

In 2002, the industry launched the Sustainable Winegrowing New Zealand label, which has developed standards and audit programmes to provide consumers with information about the sustainability practices of New Zealand vineyards and wineries. By 2016, 98% of the country's vineyard producing area was certified by Sustainable Winegrowing New Zealand facilitating the publication of the first Sustainability Report by New Zealand Winegrowers (2017c).

Another exemplar of a New Zealand exporter creating a high-value product for international consumers is Zespri, "acknowledged as the global leader in the supply and marketing of branded premium kiwifruit" (New Zealand Government, 2012, p. 19; see also Dalziel et al, 2017, chapter 3). An important part of the Zespri brand is its commitment to sustainability (Zespri, 2016).

Zespri communicates its sustainability performance to stakeholders (Zespri, 2016, p. 2). Its sustainability brochure, for example, describes specific actions being actioned under five headings (Zespri, 2016):

- Soil and Water (including pest and disease management)
- Waste management and reduction
- Managing carbon and greenhouse gas emissions
- Supporting employment and backing worker welfare
- Investing in communities and building capability

Fonterra is New Zealand's largest businesses, and the world's largest processor and exporter of dairy products (Fonterra, 2017, p. 6). In 2017, it published its first Sustainability Report. It described Fonterra's approach to sustainability.

The above examples illustrate that credence attributes such as sustainability, nutrition and community responsibility are important elements of efforts by New Zealand businesses to create high value brands.

There is a wider movement in the New Zealand primary sector focused on creating value in this way. Te Hono involves "220 Chief Executives and leaders who have a deep-seated passion and desire to develop and innovate for transformational change in the New Zealand primary sector and agribusiness" (Te Hono, 2017a). Its vision is: "Transforming the primary sector to realise the opportunity for Aotearoa, New Zealand to be recognised for our natural environment and products, as world leaders in innovation" (Te Hono, 2017b).

Te Hono was launched by the CEO of the New Zealand Merino Company, John Brakenridge, in 2012 (Brakenridge, 2016). In its own words, 'Te Hono is a journey that is unlocking the potential that exists for New Zealand to be recognised as world leaders in innovation, reputation and trust' (Te Hono, 2015, p. 2). Success is defined as 'sustainable value delivered over the long term by increasing margin and capturing value across the entire value chain, not just volume or commodity price' (idem, 2015, p. 5).

An essential element of Te Hono is the Te Hono Stanford Bootcamp, which is a week-long, intensive programme held at Stanford University in California (Te Hono, 2018). The 2015 Bootcamp agreed that New Zealand agri-food exports should attract a premium of 20 percent for their sustainability and other attributes (Holborow, 2015). This was reinforced in a presentation by David Teece, who is one of New Zealand's most pre-eminent economists as a result of his seminal research on the capability theory of the firm (see Teece, 1982, 2017a and 2017b). Teece (2015, slide 5) proposed that to improve New Zealand's current competitive advantage, businesses need to develop dynamic capabilities, which he defines elsewhere as follows (Teece, 2017a, p. 698):

For applied purposes, dynamic capabilities can usefully be broken down into three primary clusters of activities: (1) identification, development, co-development and assessment of technological opportunities in relationship to customer needs (sensing); (2) mobilization of
resources to address needs and opportunities, and to capture value from doing so (seizing); and (3) continued renewal (transforming).

In that context, Teece (2015, slide 8) noted that there doesn't appear to be a single strong New Zealand brand, other than New Zealand itself. He observed that a brand is not simply a label, but "is a story, and a customer relationship/experience built on trust that is sufficiently valuable to support a 20-30% price premium."

The Ministry for Primary Industries has set a goal of increasing the value of New Zealand's primary exports from \$32 billion in 2012 to \$64 billion by 2025 (MPI, 2017). There is significant potential to increase the value of our agricultural products by marketing non-physical credence values. New Zealand producers have traditionally been successful at meeting international markets' requirements for physical attributes of products, but less successful at selling the credence attributes of our products (Guenther et al., 2015; Saunders et al., 2016a). New Zealand land use change/practice could be impacted as domestic producers and suppliers seek to leverage credence attributes in order to reach these future export targets.

3. Digital media and smart technology trends

On an international level, greater uptake and use of digital media and smart technology is changing the way consumer behave in-market. Access to and use of digital media and smart technology is increasing. As of July 2019, there was an approximate world average of 83 mobile broadband subscriptions per 100 people, as well as 14.9 fixed broadband subscriptions per 100 people. In addition, approximately 7.4 billion of the world population were covered by a mobile cellular network, with 7.1 billion having access to at a least 3G mobile network. This is particularly pronounced in the Asia-Pacific region (4.0 billion access to at least 3G) over all other world regions (ITU, 2019).

An increasing number of consumers are seeking information and purchasing primary products, particularly food products, online. The US food and beverage ecommerce market, for example, has been valued at approximately US\$19.9 billion in 2019, representing approximately 2 per cent of total food and beverage sales in this market. This is projected to increase to approximately US\$38.2 billion in 2023, representing approximately 3.5 per cent of total sales for food and beverages in the US (eMarketer, 2019). Other world regions are experiencing similar rates of food and beverage ecommerce growth, notably China. Ecommerce for food and beverages has experienced year-on-year growth in China, with food product ecommerce sales experiencing 40.9 per cent growth between 2017 and 2018, compared with 4.7 per cent total channel growth (i.e. physical and online retail) over the same period (Chemlinked, 2019). The total number of "digital buyers" in China (i.e. those consumers who have made at least one purchase using a digital channel) was estimated at approximately 485.4 million people (44.3 per cent of China's total population) in 2017, which is expected to grow to approximately 691 million people (58.8 per cent of China's total population) in 2022 (eMarketer, 2018).

In recent years, consumer patronage of alternative supply chain models for food purchasing services has been growing, enhanced by more ubiquitous digital media and smart technology use. This includes subscription models for food products, such as Amazon Prime. The current total value of the US subscription e-commerce market, for example, is estimated to be between 12 and 15 billion USD. This market has experienced rapid growth, signalling a compound annual growth rate of approximately 60 per cent since 2014 with this rate likely to

increase over time. Many of the leading services in this category are food subscription services, such as Hello Fresh and Blue Apron (Fenyo and Mitchell, 2019).

In addition to subscription models, food firms are increasingly engaging in direct-to-consumer (D2C) sales, aided by digital media and smart technology. This could include selling their products directly to consumers via their own website rather than outsourcing sales to traditional food product or online retailers. This model effectively bypasses conventional supply chain structure for food distribution and sales by removing entirely, or integrating at the producer/processor stage, the retail/sales arm of the supply chain (McKean, 2019). Direct-to-consumer models are expected to grow in market prevalence, with estimates suggesting that D2C could account for approximately 50 per cent of US consumer packaged goods sales by 2025 (Martino, 2019). While there is currently little evidence to suggest strong engagement from New Zealand food firms in D2C activities, there is an opportunity for New Zealand food exporters. Changes in the supply chain are likely to be captured in the value chain, providing higher returns for primary producers/processors.

There is also an increasing consumer demand for transparency in product purchasing. A 2017 study of US consumers by Response Media signalled the importance of product transparency in this market, with significant percentages of participants demanding transparency in ingredients listings (99 per cent), in-depth information on ingredients/materials (98 per cent), the source of ingredients (95 per cent), the production/manufacturing process (93 per cent), shipping and handling activities (90 per cent) and sustainability, charitable or labour efforts (91 per cent). In addition, participants indicated a willingness to pay for fresh food products (92 per cent of participants) and packaged foods (89 per cent of participants) with higher transparency. Food transparency is also shown to be of the highest importance when compared with all other product attributes (Response Media, 2017).

One tool that has shown promise in delivering product information to consumers is blockchain technology. This is essentially a public data tool that stores encrypted information at every stage of the supply chain, allowing users to trace items from their origin to their destination. This stored information cannot be altered retroactively, and is freely available to the public (Soon, 2019). There has been a marked increase in the use of blockchain technology to improve supply chain transparency in recent years. For example, in a bid to encourage consumer trust, retailer Carrefour and food firm Nestle have initiated the use of IBM's blockchain platform the trace the movements of the infant formula products stocked in Carrefour stores (Alexandre, 2019). Similarly, Walmart Canada will be initiating the use of a blockchain-based freight and payment network from February 2nd 2020, designed to manage stock delivered to its 400 outlets across Canada (SDCE, 2019). As the provision of transparency and traceability is associated with a market premium, the implementation of blockchain systems may provide higher returns for New Zealand primary producers. An increased need for both transparency and direct communication between producers and consumers is also emphasised in the strategic documents of New Zealand primary sector bodies (B+LNZ, 2018).

3.6 International trading environment

1. Bilateral free trade agreements

As small open economy, New Zealand relies on market access for the trade of agricultural products and the success of its primary industries. Food, beverage and animal products make up 60 per cent of total New Zealand exports by value (UN COMTRADE, 2018), with the New Zealand agricultural, forestry and fishing sectors accounting for 5 per cent of GDP, and 29 per

cent of GDP from goods producing industries (StatsNZ, 2019c). Accordingly, New Zealand has a policy of negotiating preferential trade agreements with other nations in order to lower tariff and non-tariff barriers, the most common of which are bilateral FTAs. In addition to multilateral FTAs (see Section 3.6.2), New Zealand currently has bilateral FTAs in force with China, Australia, Malaysia, Thailand, Singapore and South Korea, with FTAs under negotiation with the EU and India (MFAT, 2019a; 2019b). Free Trade Agreements (FTAs) are signed not only to reduce the tariff barriers for bi-lateral trade, but also to create market opportunities, streamline processes, reduce overhead costs, and generate more certainty and security for businesses conducting work overseas. In terms of market access FTAs can also help local businesses be more competitive in overseas markets (MFAT, 2017c). As new agreements come into force, the primary sector may need to adapt production processes and land use practices to comply with new standards, quotas, or policies.

The potential for future agreements could greatly affect the trading profile and agricultural production (and thus land-use) in New Zealand. One study by Saunders et al. (2016c) projected the EU-NZ FTA would results in increases in NZ horticultural production and dairy production, alongside increased consumption of European imported cereals in NZ. Verevis and Üngör (2019) estimated that without the 2008 China-NZ FTA NZ would export 22 per cent less commodity exports (185 per cent less for the food and animal sectors). FTAs will continue to play an important role in market connectivity and the international trading environment, especially for agriculture.

2. Multilateral free trade agreements

The failure of the World Trade Organisation to complete the Doha Development negotiations in 2015 was a set back to the development of global/multilateral FTAs (WTO, 2015). New Zealand's agricultural sector was well-placed to benefit and grow under the progressive Doha rounds (Cornish and Fernandez, 2005). However, with the failure of the negotiations, New Zealand could beat risk of being excluded from trade negotiations, which could result in lost market share in foreign markets.

Fortunately, New Zealand is also currently a signatory on a number of multilateral FTAs. These include:

- Comprehensive and Progressive Agreement for Trans-Pacific Partnership (CPTPP), signed by New Zealand, Australia, Singapore, Malaysia, Brunei Darussalam, Viet Nam, Japan, Canada, Mexico, Preu and Chile (MFAT, 2019c);
- ASEAN-Australia-New Zealand Free Trade Agreement (AANZFTA), signed by New Zealand, Australia, Indonesia, Singapore, Malaysia, Brunei Darussalam, The Philippines, Thailand, Cambodia, Viet Nam, Laos and Myanmar (MFAT, 2019d); and
- Trans-Pacific Strategic Economic Partnership (P4), signed by New Zealand, Singapore, Chile and Brunei Darussalam (MFAT, 2019e).

In addition to the above, a number of multilateral FTAs are currently concluded but not in force, or under negotiation, including:

- NZ-Gulf Cooperation Council FTA, including New Zealand, United Arab Emirates, Oman, Qatar, Saudi Arabia, Bahrain and Kuwait (MFAT, 2019f);
- PACER Plus, including Australia, Cook Island, Kiribati, Nauru, New Zealand, Niue, Samoa, Solomon Island, Tonga, Tuvalu and Vanuatu (MFAT, 2019g);

- Regional Comprehensive Economic Partership (RECP), including New Zealand, Australia, Indonesia, Singapore, Malaysia, Thailand, Cambodia, Brunei Darussalam, The Philippines, Viet Nam, Laos, Myanmar, India, China, Japan and South Korea (MFAT, 2019h);
- New Zealand-Pacific Alliance Free Trade Agreement, including New Zealand, Chile, Colombia, Mexico and Peru (MFAT, 2019i); and
- Russia-Belarus-Kazakhstan Customs Union FTA, including New Zealand, Russia, Belarus and Kazakhstan (MFAT, 2019j).

New Zealand has a relatively diverse profile of trading partners, with a Herfindahl export index of 0.11 (WITS, 2019). However trade with China, New Zealand's most significant bi-lateral trade partner, accounts for over 22 per cent of export trade, and 19 per cent of import trade (WITS, 2019). Thus any potential disruption of trade with China, or change in market access, political tensions, or the economy of China, would have significant ramifications for New Zealand's trading environment. These concerns are emphasised given the recent trade war between the US and China, two of New Zealand's most importance trading partners (the United States is the 3rd largest import & export market for New Zealand) (BBC, 2019b).

3. Comprehensive and Progressive Agreement for Trans-Pacific Partnership (CPTPP) Following the election of President Donald Trump, the United States introduced more protectionist policies in order to shield US industries, affecting trade deals such as the North America Free Trade Agreement (NAFTA) and the Trans-Pacific Partnership (TPP). The TPP was abandoned by the US in 2017, leading to the redevelopment of the TPP without the US as a signatory. Negotiations concluded in January 2018, with the CPTPP signed by Trade Ministers in Chile on March 8th 2018 (MFAT, 2019c).

CPTPP signatories represented approximately 30 per cent of New Zealand's goods exports (NZ\$16.7 billion) and 30 per cent of services exports (NZ\$7.3 billion) by value in June 2018. In addition, New Zealand has never had a bilateral FTA with four of the signatories (Japan, Canada, Mexico and Peru), to which New Zealand exports approximately NZ\$5.5 billion of goods and services annually. It has been estimated that reductions in tariff barriers for many of New Zealand's largest primary product exports resulting from the signing of the CPTPP could be up to NZ\$222 million each year (MFAT, 2019c). The signing of the CPTPP affects trade policy, tariffs and export/import quotas, which ultimately affect New Zealand's primary industries.

4. Brexit

On June 23rd, 2016, the United Kingdom (UK) voted narrowly (52:48) to leave the EU. The UK government then officially notified the EU on March 29th 2017, of its intention to leave, thus triggering Article 50 of the EU Treaty, which specifies that within two years the UK will cease to be a member. However, the date for Brexit is still unclear as EU leaders have granted the UK multiple extensions for Brexit. Consequently, the nature of the economic relationship between the UK and the remaining EU-27 is still to be defined.

Brexit is likely to change the domestic and trade policies affecting agriculture in the UK and have implications for agricultural commodity trade worldwide. Trade policy changes are key factors in determining the consequences of Brexit for agricultural markets in Europe and elsewhere. Therefore it is difficult to assess the impact of Brexit on New Zealand until the more detailed policies are known. The UK is still an important export market for New Zealand,

especially for sheep meat. However, modelling suggests that the impact of Brexit on New Zealand will be minimal (Saunders et al., 2019). Since 1973, New Zealand's agricultural exports to the UK were subject to EU trade policy, hence, the UK exiting the EU will affect trade between the UK, the EU and third party countries like New Zealand.

5. Agricultural subsidies/policy

The subsidisation of agriculture undermines market competitiveness for all countries exporting and importing food. The removal of subsidies encourages land use practices that are more focused on sustainability, efficiency and yield (Strubenhoff, 2016). The EU is expected to introduce new reforms addressing its Common Agricultural Policy (CAP), which currently spends €60 billion subsidising farmers (EURACTIV, 2017). The World Trade Organisation abolished agricultural export subsidies in 2015 (Strubenhoff, 2016). However, the agricultural sectors in some countries such as China still remain heavily protected by government subsidies (Arsenault, 2014). Over the last 20 years China has become the world's largest producer, consumer and importer of agricultural products (Lopez et al., 2017). Its transformation from a rural to urban manufacturing and service economy has affected its agricultural sector have been introduced and are expected to rise in the future (Lopez et al., 2017). As has been seen in developed regions such as the European Union, Japan and the United States, this may mean that China reduces imports which could affect New Zealand's export market.

6. Non-tariff barriers

Over the last twenty years, Governments internationally have introduced an increasing number of non-tariff trade measures (USC, 2016). Non-tariff trade barriers restrict imports and exports of goods and service, and range from import quotas, technical and licencing requirements, custom delays, and subsidies. The relaxing of NTBs is a significant component of trade agreements, 70 per cent of the projected benefit from the proposed TPP RTA were expected to come from reductions in NTBs (NZIER 2017). Non-tariff barriers when applied correctly can increase competition and product quality, and improve social and environmental wellbeing. They can also become barriers to trade, distorting markets, raising costs, reducing competiveness, and impacting food security (USC, 2016). Non-tariff barriers could affect market access and quotas for New Zealand primary products in the future.

4. Conclusion

Enhancing primary sector production and productivity while maintaining and improving our land and water quality for future generations is a key outcome of the Our Land and Water National Science Challenge. It is therefore important to identify the hierarchy of international and national issues, is needed to provide an evidence base to guide investment and inform the Challenge Research Strategy. This report presents an overview of the international and domestic drivers that have the potential to influence land use change/practice In New Zealand. This report also looks to inform the strategic direction of the OLW Challenge by identifying the likely impact of these drivers in the future.

The current report has been informed by two previous iterations, in which workshops, stakeholder surveys and extensive literature review produced a series of 32 key drivers of land use change/practice. This was later expanded to 34 key drivers. Links to updated summaries of the key drivers are provided, along with an evidence base comprising 1,097 unique sources

(850 international and 247 domestic sources) across the three iterations of this project, are included in this report.

The current report modified and extended a survey of New Zealand primary sector stakeholders, designed to identify which drivers they believed to be the most important for land use change/practice domestically and internationally. The survey was distributed to 335 participants in total, receiving 226 completed surveys.

Using an unprompted, open text entry, the survey results showed that a highly significant percentage of stakeholders thought that climate change was the most important driver of land use change/practice on an international level, and the second-most important driver of land use change/practice on a domestic level (below water quality). Participants also identified environmental condition, GHG emissions, water quality and trade agreements as international drivers that would have a high impact on land use. Furthermore, participants identified water quality, nitrate limits, environmental condition, GHG emissions and biosecurity as domestic drivers that would have a high impact on land use.

This report also examined future trends and challenges and their likely impact on New Zealand land use change/practice. These were grouped under six broad headings: climate change, global trends and challenges, emerging technologies, innovative products/new food technology, consumer trends, and international trading environment.

Climate change is most likely to be highly impactful on land use change/practice into the future, producing significant disruptions to regional production trends, biosecurity, ecosystem integrity and social conditions, as well as producing higher frequency and intensity extreme weather events and heavily influencing domestic policy regarding land use. A growing global population, coupled with the challenges of maintaining a social license to operate, combating food waste and maintaining stable markets, are seen as challenges on a global scale. Greater development and use of new technologies designed to provide data and improve practices, both on-farm and in-market, are also highly likely to influence land use trends. Consumer preferences are also changing, with an increased market presence for alternative protein products, as well as increasing consumer interest in vegetarian, vegan and flexitarian diets. The international trading environment will also continue to contribute to the success of New Zealand's primary product exports, particularly with the development of bilateral and multilateral free trade agreements.

The mission statement of the Our Land and Water Challenge is to "enhance primary sector production and productivity while maintaining and improving our land and water quality for future generations" (OLW, 2018). This report has examined the impact of domestic and international drivers on New Zealand land use change/practice, and has utilised the knowledge and expertise of those involved in the primary sector to help inform these. The likely impact of future trends and challenges on land use change/practice in New Zealand was also examined. The future of sustainable and productive primary land use will require identifying and adapting to the issues, trends and drivers outlined in this report.

References

Acumen Research and Consulting (ARC) (2019). Vegan Food Market Size Worth Around US\$24.3 Billion by 2026 [website]. Accessed 21st November 2019 from https://www.globenewswire.com/news-release/2019/06/11/1867025/0/en/Vegan-Food-Market-Size-Worth-Around-US-24-3-Billion-by-2026.html.

Aguilar, G., Blanchon, D., Foote, H., Pollonais, C. and Mosee, A. (2015). Queensland fruit fly invasion of New Zealand: predicting area suitability under future climate change scenarios. Unitec ePress Perspectives in Biosecurity Research Series (2015/2). ISSN: 2422-8494. Retrieved November 26th 2019 from https://unitec.researchbank.ac.nz/handle/10652/3055.

AI Forum New Zealand (AIFNZ) (2018). Artificial Intelligence: Shaping a Future New Zealand. Retrieved 20th November 2019 from https://www.mbie.govt.nz/dmsdocument/5754artificial-intelligence-shaping-a-future-new-zealand-pdf.

Akhtar, Y. and Ishman, M.B. (2018). Insects as an alternative protein source. In Proteins in Food Processing (pp. 263-288). Woodhead Publishing.

Alexandre, A. (2019). Carrefour, Nestle Use IBM's Blockchain Platform to Track Infant Formula [website]. Accessed 18th November 2019 from https://cointelegraph.com/news/carrefour-nestle-use-ibms-blockchain-platform-to-trackinfant-formula.

Al-Kodmany, K. (2018). The Vertical Farm: A Review of Developments and Implications for the Vertical City. Buildings, 8(2); 24.

Allen, J. (2019). John Deere develops fully electric, autonomous tractor [website]. Accessed 20th November 2019 from https://www.ivtinternational.com/news/agriculture/john-deere-develops-fully-electric-autonomous-tractor.html.

Alternative Meat Company (AMC) (2019). Alternative Meat Company [website]. Accessed 21st November 2019 from https://altmeatco.com/.

Animal Products Act 1999.

Antedote (2018). Future of Meat: How should New Zealand's red meat sector respond to alternative protein advancements? Summary Report, February 2018. Retrieved 2nd December 2019 from

https://beeflambnz.com/sites/default/files/levies/files/Alternative%20Proteins%20summary %20report.pdf

Anzilotti, E. (2019). General Mills has a plan to regenerative 1 million acres of farmland [website]. Accessed 2nd December 2019 from

https://www.fastcompany.com/90313818/general-mills-has-a-plan-to-regenerate-1-million-acres-of-farmland.

Apatov, E., Chappell, N. and Grimes, A. (2018). Is internet on the right track? The digital divide, path dependence, and the rollout of New Zealand's ultra-fast broadband. Motu Working Paper 18-04, March 2018: Motu Economic and Public Policy Research. Retrieved 19th November 2019 from http://motu-www.motu.org.nz/wpapers/18_04.pdf.

Arsenault, C. (2014). Developing countries blast rich-world farm subsidises at Rome talks. Retrieved 8th November 2017 from https://www.reuters.com/article/us-foundation-foodsubsidies/developing-countries-blast-rich-world-farm-subsidies-at-rome-talks-idUSKCN0HV1NK20141007

ATKearney (2019). How Will Cultured Meat and Meat Alternatives Disrupt the Agricultural Food Industry? Retrieved 21st November 2019 from

https://www.atkearney.com/documents/20152/2795757/How+Will+Cultured+Meat+and+ Meat+Alternatives+Disrupt+the+Agricultural+and+Food+Industry.pdf/06ec385b-63a1-71d2c081-51c07ab88ad1?t=1559860712714.

Baines, J. and Edwards, P. (2018). The role of relationships in achieving and maintaining and social license in the New Zealand aquaculture sector. Aquaculture, 485; 140-146.

Baker, M. J. and Ballington, L. (2002). Country of origin as a source of competitive advantage. Journal of Strategic Marketing, 10(2), pp. 157-168.

Barnosky, A.R., Hoddy, K.K., Unterman, T.G. and Varady, K.A. (2014). Intermittent fasting vs daily calorie restriction for type 2 diabetes prevention: a review of human findings. Translational Research, 164(4); 302-311.

Barrios, E.X. and Costell, E. (2004). Review: Use of Methods of Research into Consumers' Opinions and Attitudes in Food Research. Food Science and Technology International December, 10(6), 359-371.

Beattie, A. B., Jim. (2017). UK and EU strike initial deal on WTO quotas in Brexit breakthrough. Retrieved 8th November 2017 from https://www.ft.com/content/e30185c6-a83d-11e7-ab55-27219df83c97.

Beef+Lamb New Zealand (B+LNZ, 2018). Environment Strategy and Implementation Plan, 2018-22. Retrieved December 3rd 2019 from https://beeflambnz.com/sites/default/files/levies/files/Env-strategy-Imp-plan.pdf.

Bellman, C. and Hepburn, J. (2017). The decline of commodity prices and global agricultural trade negotiations: A game changer? International Development Policy, 8 (8.1).

Bennett, C. (2019). Driverless Goes Big in Farming 2019 [website]. Accessed 20th November 2019 from https://www.agprofessional.com/article/driverless-goes-big-farming-2019.

Berry, C., Mukherjee, A., Burton, S. and Howlett, E. (2015). A COOL Effect: The direct and indirect impact of country-of-origin disclosures on purchase intentions for retail food products. Journal of Retailing, 91(3), pp. 533-542.

Biosecurity New Zealand (BNZ) (2019). Media Releases [website]. Accessed 29th November 2019 from https://www.biosecurity.govt.nz/protection-and-response/mycoplasma-bovis/resources-for-mycoplasma-bovis/media-releases/.

Bjorklund, J., Limburg, K.E. and Rydberg, T. (1999). Impact of production intensity on the ability of the agricultural landscape to generate ecosystem services: an example from Sweden. Ecological Economics, 29(2); 269-291.

Brackenridge, J. 2016. Why are we wasting a good crisis? The value shift our primary sector needs. In C. Massey (Ed), New Zealand Land and Food Annual, Volume 1. Auckland: Massey University Press, pp. 27-36.

British Broadcasting Corporation (BBC) (2019a). Paris climate accords: US notifies UN of intention to withdraw [website]. Accessed December 2nd 2019 from https://www.bbc.com/news/world-us-canada-50297029.

British Broadcasting Corporation (BBC) (2019b). A quick guide to the US-China trade war [website]. Accessed December 11th 2019 from https://www.bbc.com/news/business-45899310.

Brown, C. (2016). Country, culture and community: Understanding Indigenous agribusiness in southern Australia [website]. Accessed December 3rd 2019 from https://www.abc.net.au/news/rural/2016-02-16/indigenous-agribusiness-in-southernaustralia/7079354.

Bruinsma, J., de Haen, H., Alexandratos, N., Schmidhuber, J., Bödeker, G., Ottaviani, M., & Harrison, P. (2015). World agriculture: Towards 2015/2030: An FAO perspective. FAO, Rome. World agriculture: towards, 2030. Retrieved 15th November 2017 from http://www.fao.org/3/a-y4252e.pdf.

Burton, R.J. (2019). Potential impact of synthetic animal protein on livestock production: The new "war against agriculture"? Journal of Rural Studies, 68; 33-45.

Cabinet Economic Development Committee (CEDC) (2018). The One Billion Trees programme – actions and decisions for implementation. Retrieved 18th November 2019 from https://www.teururakau.govt.nz/dmsdocument/30942-the-one-billion-trees-programme-cabinet-paper.

Caride, C., Pineiro, G. and Paruelo, J.M. (2012). How does agricultural management modify ecosystem services in the argentine Pampas? The effects on soil C dynamics. Agriculture, Ecosystems and Environment, 154(1); 23-33.

Carter, C., Krissoffr, B. and Zwane, A. P. (2006). Can country of origin labelling succeed as a marketing tool for produce? Lessons from three case studies. Canadian Journal of Agricultural Economics, 54 (2006), pp. 513-530.

Cicia, G., Cembalo, L., Del Giudice, T. and Scarpa, R. (2011). The impact of country-of-origin information on consumer perception of environment-friendly characteristics. International Journal on Food System Dynamics, 2(1), pp. 106-111.

Convention on Biological Diversity (CBD) (2008). New Zealand's Fifth National Report to the United Nations Convention on Biological Diversity. Retrieved December 11th 2019 from https://www.cbd.int/doc/world/nz/nz-nr-05-en.pdf.

Chai, B.C., van der Voort, J.R., Grofelnik, K., Eliasdottir, H.G., Kloss, I. and Perez-Cueto, F.J.A. (2019). Which diet has the least environmental impact on our planet? A systematic review of vegan, vegetarian and omnivorous diet. Sustainability, 11(15); 4110.

Chandra, A., McNamara, K.E. and Dargusch, P. (2017). Climate-smart agriculture: perspectives and framings. Climate Policy, 18(4); 526-541.

Chemlinked (2019). Food Sector Growth in China's New Retail Era [website]. Accessed December 3rd 2019 from https://food.chemlinked.com/market-insights/food-sector-growth-chinas-new-retail-era.

Christie, J.E. (2014). Adapting to a changing climate: A proposed framework for the conservation of terrestrial native biodiversity in New Zealand. Department of Conservation. Retrieved 27th November 2019 from

https://www.doc.govt.nz/globalassets/documents/science-and-technical/sap257.pdf.

Churkina, G. (2016). The Role of Urbanization in the Global Carbon Cycle. Frontiers in Ecology and Evolution, https://doi.org/10.3389/fevo.2015.00144.

Claret, A., Guerrero, L., Aguirre, E., Rincon, L., Hernandez, M.D., Martinez, I., Peleteiro, J.B., Grau, A. and Rodriguez-Rodriguez, C. (2012). Consumer preferences for sea fish using conjoint analysis: Exploratory study of the importance of country of origin, obtaining method, storage conditions and purchasing price. Food Quality and Preference, 26(2), pp. 259-266.

Claver, H. (2019). Rapid adoption of artificial intelligence in agriculture [website]. Accessed 20th November 2019 from https://www.futurefarming.com/Smartfarmers/Articles/2019/8/Rapid-adoption-of-artificial-intelligence-in-agriculture-461266E/#targetText=According%20to%20research%20company%20MarketsandMarkets,22 .5%25%20during%20the%20forecast%20period.

Climate Action Tracker (CAT) (2019). New Zealand [website]. Accessed December 11th 2019 from https://climateactiontracker.org/countries/new-zealand/.

Climate Change Adaptation Technical Working Group (CCATWG) (2017). Adapting to Climate Change in New Zealand. Retrieved 26th November 2019 from https://www.mfe.govt.nz/sites/default/files/media/Climate%20Change/adapting-toclimate-change-stocktake-tag-report.pdf.

Climate Change Response (Zero Carbon) Amendment Bill.

Climate Change Response Act (2002).

Cornish, G.A. and Fernandez, S. (2005). Agricultural trade liberalization: Implications for irrigated agriculture. IPTRID Secretariat, Food and Agriculture Organization of the United Nations. Rome.

Corrado, S. and Sala, S. (2018). Food waste accounting along global and European food supply chains: State of the art and outlook. Waste Management, 79; 120-131.

Costanza, R., d'Arge, R., de Groot, R., Farber, S., Grasso, M., Hannon, B., Limburg, K. ... and van den Belt, M. (1997). The value of the world's ecosystem services and natural capital. Nature, 387 (1997); 253-260.

Cradock-Henry, N.A. (2017). New Zealand kiwifruit growers' vulnerability to climate and other stressors. Regional Environmental Change, 17(1); 245-259.

Craft Meat Company (CMC) (2019). Craft Meat Company [website]. Accessed 21st November 2019 from https://www.craftmeat.co.nz/.

DairyNZ (2017). Dairy Tomorrow: The Future of New Zealand Dairying. New Zealand Dairy Industry Strategy 2017-2025. Retrieved 18th November 2019 from https://www.dairytomorrow.co.nz/wp-content/uploads/2017/12/dairy-strategy-2017-A4booklet-Part3.pdf. DairyNZ (2019). Wellbeing [website]. Accessed December 9th 2019 from https://www.dairynz.co.nz/people/wellbeing/.

Dalziel, P., Saunders, C., Tait, P. and Saunders, J. (2019). Credence Attributes and New Zealand Country of Origin: A Review. AERU Research Report No. 351, January 2019. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Datar, I. and Betti, M. (2010). Possibilites for an in vitro meat production system. Innovative Food Science & Emerging Technologies, 11(1); 13-22.

Davis, M., Head, N.J., Myers, S.C. and Moore, S.H. (2019). Department of Conservation guidelines for assessing significant ecological values. Department of Conservation. Retrieved December 11th 2019 from https://www.doc.govt.nz/documents/science-and-technical/sfc327entire.pdf.

Department of Conservation (DOC) (2011). Potential effects of climate change on New Zealand's terrestrial biodiversity and policy recommendations for mitigation, adaptation and research. Science for Conservation 312. Retrieved 27th November 2019 from https://www.doc.govt.nz/globalassets/documents/science-and-technical/sfc312entire.pdf.

Digital Inclusion Research Group (DIRG) (2017). Digital New Zealanders: The Pulse of Our Nation. A report to MBIE and DIA, May 2017. Retrieved 19th November 2019 from https://www.mbie.govt.nz/dmsdocument/3228-digital-new-zealanders-the-pulse-of-our-nation-pdf.

Dudman, S. (2019). New agricultural spraying business Agdrone takes off in Whanganui [website]. Accessed 20th November 2019 from https://www.nzherald.co.nz/the-country/news/article.cfm?c_id=16&objectid=12205638.

Eb, D. (2019). With the walls closing in, regenerative farming is a way forward for agriculture [website]. Accessed December 11th 2019 from https://thespinoff.co.nz/business/04-11-2019/with-the-walls-closing-in-regenerative-farming-is-a-way-forward-for-agriculture/.

The Economist (2015). Long-term macroeconomic forecasts: Key trends to 2050. A special report from The Economist Intelligence Unit. Retrieved 12th November 2019 from https://espas.secure.europarl.europa.eu/orbis/sites/default/files/generated/document/en/Long-termMacroeconomicForecasts_KeyTrends.pdf.

The Economist. (2017). Technology Quarterly - The Future of Agriculture. Retrieved 30th October 2017 from http://www.economist.com/technology-quarterly/2016-06-09/factory-fresh.

The Economist (2019). The World in 2019: The year of the vegan [website]. Accessed 21st November 2019 from https://worldin2019.economist.com/theyearofthevegan.

eMarketer (2018). China Retail and Ecommerce 2018: The Convergence of Online, Offline and Technology [website]. Accessed December 3rd 2019 from https://www.emarketer.com/content/china-retail-and-ecommerce-2018.

eMarketer (2019). Grocery Ecommerce 2019: Online Food and Beverage Sales Reach Inflection Point [website]. Accessed December 3rd 2019 from https://www.emarketer.com/content/grocery-ecommerce-2019. Environmental Protection Agency (EPA). (2016). Climate Change Impacts on Ecosystems. Retrieved 8th November 2017 from https://19january2017snapshot.epa.gov/climate-impacts/climate-impacts-ecosystems_.html.

EURACTIV (2017). EU, Brazil propose farm subsidies overhaul to help developing countries. Retrieved 3rd November 2017 from https://www.euractiv.com/section/agriculture-food/news/eu-brazil-propose-farm-subsidies-overhaul-to-help-developing-countries/

European Commission (2017). Paris Agreement. Retrieved 2nd November 2017 from https://ec.europa.eu/clima/policies/international/negotiations/paris_en.

Extinction Rebellion (XR) (2019a). About Us [website]. Accessed December 2nd 2019 from https://rebellion.earth/the-truth/about-us/.

Extinction Rebellion (XR) (2019b). Local Groups [website]. Accessed December 2nd 2019 from https://rebellion.earth/act-now/local-groups/.

FarmStrong (2019). About FarmStrong [website]. Accessed December 9th 2019 from https://farmstrong.co.nz/about/.

Fenyo, K. and Mitchell, A. (2019). Sizing Up The Subscription E-Commerce Market: 2018 Update. Accessed 18th November 2019 from https://get.fuelbymckinsey.com/article/sizingup-the-subscription-e-commerce-market/.

Figure.nz (2019). Māori enterprises in New Zealand (as at February 2011-2019). Accessed 19th November 2019 from https://figure.nz/chart/TaxNr9SAiwrZbFfp-Nz0JzbkYc3ILk4Js.

Fior Markets (FM) (2019). Global Dairy Alternatives Market is Expected to Reach USD 38.9 Billion by 2025 [website]. Accessed 21st November 2019 from https://www.globenewswire.com/news-release/2019/09/11/1913982/0/en/Global-Dairy-Alternatives-Market-is-Expected-to-Reach-USD-38-9-Billion-by-2025-Fior-Markets.html.

Fonterra (2017). Sustainability report for the year Ending 31 July 2017. Auckland: Fonterra Cooperative Group Ltd.

Fonterra (2019). Fonterra to explore opportunities in complementary nutrition [website]. Accessed 21st November 2019 from https://www.fonterra.com/nz/en/our-stories/media/fonterra-to-explore-opportunities-in-complementary-nutrition.html.

Food and Agriculture Organization of the United Nations (FAO) (2008). Climate change and transboundary pests and diseases. Retrieved 26th November 2019 from http://www.fao.org/tempref/docrep/fao/010/i0142e/i0142e06.pdf.

Food and Agriculture Organization of the United Nations (FAO) (2012). Global Trends and Future Challenges for the Work of the Organisation Retrieved 30th October 2017 from http://www.fao.org/docrep/meeting/025/md883E.pdf.

Food and Agriculture Organization of the United Nations (FAO) (2015). 70 Years of FAO: 1945-2015: United Nations. Retrieved 30th October 2017 from http://www.fao.org/18196504-6474-4FFE-9F46EB65C8597346/FinalDownload/DownloadId-DFEEC01D8BC6A640BFB4A67055FA0FBB/18196504-6474-4FFE-9F46-EB65C8597346/3/a-i5142e.pdf.

Food and Agriculture Organization of the United Nations (FAO) (2017a). The Future of Food and Agriculture: Trends and Challenges. Rome: United Nations FAO. Retrieved 30th October 2017 from http://www.fao.org/FBB6CAA4-D3AD-4902-AB99-

1913B3FF32C0/FinalDownload/DownloadId-

7A6682F20A0075F6B1D6208C12704A10/FBB6CAA4-D3AD-4902-AB99-1913B3FF32C0/3/a-i6583e.pdf.

Food and Agriculture Organization of the United Nations (FAO) (2017b). Trade and Markets. Retrieved 1st December 2017 from

http://www.fao.org/economic/est/issues/volatility/en/#.WiiwKbkUnxN.

Food and Agriculture Organization of the United Nations (FAO) (2019a). Digital Technologies in Agriculture and Rural Areas: Briefing Paper. Retrieved December 3rd 2019 from http://www.fao.org/3/ca4887en/ca4887en.pdf.

Food and Agriculture Organization of the United Nations (FAO) (2019b). Urban agriculture [website]. Accessed December 3rd 2019 from http://www.fao.org/urban-agriculture/en/.

Food and Agriculture Organization of the United Nations (FAO) (2019c). Food for the Cities. Retrieved December 3rd 2019 from http://www.fao.org/tempref/docrep/fao/012/ak824e/ak824e00.pdf.

Forestry New Zealand (2018). The One Billion Trees Programme: Our future, our billion trees. Retrieved 18th November 2019 from https://www.mpi.govt.nz/dmsdocument/31860/direct.

Forrester, G. (2019). School climate strike: Adults join with kids in huge day of national protest [website]. Accessed December 2nd 2019 from https://www.stuff.co.nz/environment/climate-news/116156233/school-climate-strike-adults-join-with-kids-in-huge-day-of-national-protest.

Frame, D., Rosier, S., Carey-Smith, T., Harrington, L., Dean, S. and Noy, I. (2018). Estimating financial costs of climate change in New Zealand: An estimate of climate change-related weather costs. New Zealand Climate Change Research Institute and NIWA. Retrieved 26th November 2019 from https://treasury.govt.nz/sites/default/files/2018-08/LSF-estimating-financial-cost-of-climate-change-in-nz.pdf.

Frank Film (2019). Cultivate Christchurch growing both veges and opportunities [website]. Accessed December 3rd 2019 from https://www.stuff.co.nz/the-press/christchurch-life/112637267/cultivate-christchurch-growing-both-veges-and-opportunities.

Fridays For Future (FFF) (2019). About #FridaysForFuture [website]. Accessed 25th November 2019 from https://www.fridaysforfuture.org/about.

Fresan, U. and Sabate, J. (2019). Vegetarian Diets: Planetary Health and Its Alignment with Human Health. Advances in Nutrition, 10(4); S380-S388.

FutureBrand (2014). Made In: The Value of Country of Origin for Future Brands. London: FutureBrand, accessed 23 February 2016 at www.futurebrand.com/.

FutureBrand (2015). Country Brand Index 2014/2015. London: FutureBrand. Accessed 23 June 2016 at www.mumbrella.asia/content/uploads/2014/11/CountryBrandIndex2014.pdf.

General Mills (2019). Regenerative Agriculture [website]. Accessed 2nd December 2019 from https://www.generalmills.com/Responsibility/Sustainability/Regenerative-agriculture.

Ghobadpour, A., Boulon, L., Mousazadeh, H., Malvajerdri, A.S. and Rafiee, S. (2019). State of the art of autonomous agricultural off-road vehicles driven by renewable energy systems. Energy Procedia, 162; 4-13.

GoodYarn (2019). About [website]. Accessed December 9th 2019 from http://www.goodyarn.org/about/.

Gosnell, H., Gill, N. and Voyer, M. (2019). Transformational adaptation on the farm: Processes of change and persistence in transitions to 'climate-smart' regenerative agriculture. Global Environmental Change, 59 (2019); 101965.

Grassian, T. (2019). Meat Reduction and Vegan Promotion: Motivators, barriers, and dietary changes from the largest study of participants in meat reduction and vegan campaigns. Retrieved 21st November 2019 from

https://forusallsite.files.wordpress.com/2019/03/grassian-reductionandveganpromotion-v2.pdf.

Grebitus, C., Printezis, I. and Printezis, A. (2017). Relationship between Consumer Behavior and Success of Urban Agriculture. Ecological Economics, 136 (2017); 189-200.

Grey, S. and Newman, L. (2018). Beyond culinary colonialism: indigenous food sovereignty, liberal multiculturalism, and the control of gastronomic capital. Agriculture and Human Values, 35(3); 717-730.

Guenther, M., Saunders, C., & Tait, P. (2014). Impacts of changes in China and India on New Zealand trade and greenhouse gas emissions. In 2014 Conference, August 28-29, 2014, Nelson, New Zealand (No. 189051). New Zealand Agricultural and Resource Economics Society. Retrieved 7th December 2017 from http://ageconsearch.umn.edu/handle/189051

Guenther, M., Saunders, C., Dalziel, P., Rutherford, P. and Driver, T. (2015). Maximising Export Returns (MER): Consumer attitudes towards attributes of food and beverages in export markets relevant to New Zealand. AERU Research Report No. 336, November 2015. Agribusiness & Economics Research Unit (AERU), Lincoln University: Lincoln, New Zealand.

Guibourg, C. and Briggs, H. (2019). Climate change: Which vegan milk is best? BBC News. Accessed 21st November 2019 from https://www.bbc.com/news/science-environment-46654042.

Harmsworth, G.R. and Awatere, S. (2016). Indigenous Māori knowledge and perspectives of ecosystems. In Dymond, J.R. (ed.), Ecosystem services in New Zealand – conditions and trends. Manaaki Whenua Press, Lincoln, New Zealand.

Harrington, L.J., Rosier, S., Dean, S.M., Stuart, S. and Scahill, A. (2014). The Role of Anthropogenic Climate Change in the 2013 Drought over North Island, New Zealand. American Meteorological Society, December 2014. Retrieved 26th November 2019 from https://ui.adsabs.harvard.edu/abs/2014AGUFMGC43B0703H/abstract.

Healthline (2017). What is Intermittent Fasting? Explained in Human Terms [website]. Accessed 21st November 2019 from https://www.healthline.com/nutrition/what-is-intermittent-fasting.

Healthline (2018). The Ketogenic Diet: A Detailed Beginner's Guide to Keto [website]. Accessed 21st November 2019 from https://www.healthline.com/nutrition/ketogenic-diet-101.

Heffernan, O. (2017). Sustainability: A meaty issue. Nature, 544(7651), 518-520.

Hemmerling, S., Hamm, U. and Spiller, A. (2015). Consumption behaviour regarding organic food from a marketing perspective: A literature review. Organic Agriculture, 5(4), 277-313.

Hogan, F. (2019). NZ embracing gene-editing is a 'no-brainer' – Geoff Simmons [website]. Accessed 20th November 2019 from

https://www.newshub.co.nz/home/shows/2019/05/nz-embracing-gene-editing-is-a-no-brainer-geoff-simmons.html.

Holborow, R. (2015) E-mail headed "RE: Helping secure that 20% price premium we talked about at Stanford..." Personal communication from Te Hono Stanford Bootcamp participant, dated 24 July.

Horticulture New Zealand (HNZ) (2017). New Zealand domestic vegetable production: The growing story. Retrieved 26th November 2019 from http://www.hortnz.co.nz/assets/Media-Release-Photos/HortNZ-Report-Final-A4-Single-Pages.pdf.

Hsiang, S., Kopp, R., Jina, A., Rising, J., Delgado, M., Mohan, S., Rasmuss, D.J. ... and Houser, T. (2017). Estimating economic damage from climate change in the United States. Science, 356(6345); 1362-1369.

Ineson, J. (2017). Roimata Food Commons: 'The very first step in creating a food system' for Christchurch's Radley Park [website]. Accessed December 3rd 2019 from https://www.stuff.co.nz/the-press/christchurch-life/95662680/roimata-food-commons-the-very-first-step-in-creating-a-food-system-for-christchurchs-radley-park.

Infometrics (2019). Regional Wellbeing: A broader view of community outcomes around New Zealand. Retrieved 19th November 2019 from https://static.infometrics.co.nz/Content/Infometrics_Regional_Wellbeing_2019.pdf.

Insch, A., Williams, S. and Knight, J. G. (2015). Managerial perceptions of country-of-origin: An empirical study of New Zealand food manufacturers. Journal of Food Products, 22(3), pp. 304-319.

Intergovernmental Panel on Climate Change (IPCC) (2014). AR5 Climate Change 2014: Impacts, Adaptation, and Vulnerability – Part A: Global and Sectoral Aspects. Retrieved 27th November 2019 from https://www.ipcc.ch/site/assets/uploads/2018/02/WGIIAR5-PartA_FINAL.pdf.

Intergovernmental Panel on Climate Change (IPCC) (2019a). Climate Change and Land: Summary for Policymakers. Retrieved 25th November 2019 from https://www.ipcc.ch/site/assets/uploads/2019/08/Edited-SPM_Approved_Microsite_FINAL.pdf.

Intergovernmental Panel on Climate Change (IPCC) (2019b). Special Report on Climate Change and Land. Accessed 25th November 2019 from https://www.ipcc.ch/srccl/.

International Labor Organization (ILO) (2015). Indigenous Peoples in the World of Work in Asia and the Pacific: A Status Report. Retrieved December 3rd 2019 from

https://www.ilo.org/wcmsp5/groups/public/---dgreports/--gender/documents/publication/wcms_438853.pdf.

International Telecommunication Union (ITU) (2019). Key ICT indicators for developed and developing countries and the world (totals and penetration rates). Retrieved December 3rd 2019 from https://www.itu.int/en/ITU-

D/Statistics/Documents/statistics/2019/ITU_Key_2005-2019_ICT_data_with%20LDCs_28Oct2019_Final.xls.

Ipsos (2018). An exploration into diets around the world (August 2018). Retrieved 21st November 2019 from https://www.ipsos.com/sites/default/files/ct/news/documents/2018-09/an_exploration_into_diets_around_the_world.pdf.

Irimia, M. (2016). Five ways agriculture could benefit from artificial intelligence. Retrieved 7th November 2017 from https://www.ibm.com/blogs/watson/2016/12/five-ways-agriculture-benefit-artificial-intelligence/

Iyengar, S. and Massey, D.S. (2018). Scientific communication in a post-truth society.Proceedings of the National Academy of Sciences of the United States of America, 116(16); 7656-7661.

James, L. (2019). Biosecurity NZ: More fruit flies will find their way to NZ as global warming continues [website]. Accessed 26th November 2019 from https://www.tvnz.co.nz/one-news/new-zealand/biosecurity-nz-more-fruit-flies-find-their-way-global-warming-continues.

Kalantari, F., Mohd Tahir, O., Mahmoudi Lahijani, A. and Kalantari, S. (2017). A review of vertical farming technology: A guide for implementation of building integrated agriculture in cities. In Advanced Engineering Forum (Vol. 24, pp. 76-91). Trans Tech Publications.

Kalaugher, E., Beukes, P., Bornman, J.F., Clark, A. and Campbell, D.I. (2017). Modelling farmlevel adaptation of temperature, pasture-based dairy farms to climate change. Agricultural Systems, 153 (May 2017); 53-68.

Keall, C. (2019). Chicken-free Chicken marker Sunfed invades Australia [website]. Accessed 21st November 2019 from https://www.nzherald.co.nz/business/news/article.cfm?c_id=3&objectid=12242103.

Kean, J. M., Brockerhoff, E. G., Fowler, S., Gerard, P. J., Logan, D. P., Mullan, A. B., Ward, D. F. (2015). Effects of climate change on current and potential biosecurity pests and diseases in New Zealand: Ministry for Primary Industries.

Kelly, S. (2017). Brexit: the past, present and future impact on New Zealand. Retrieved 8th November 2017 from http://www.stuff.co.nz/national/politics/opinion/91474550/brexit-the-past-present-and-future-impact-on-new-zealand.

Kenny, G. (2001). Climate Change: Likely impacts on New Zealand agriculture. A report prepared for the Ministry for the Environment as part of the New Zealand Climate Change Programme. Retrieved November 26th 2019 from https://www.mfe.govt.nz/sites/default/files/impacts-agriculture-sep01.pdf.

Krikser, T., Zasada, I. and Piorr, A. (2019). Socio-Economic Viability of Urban Agriculture – A Comparative Analysis of Success Factors in Germany. Sustainability, 11; 1999. doi:10.3390/su11071999 LaCanne, C.E. and Lundgren, J.G. (2018). Regenerative agriculture: merging farming and natural resource conservation profitably. PeerJ, 6:e4428; DOI 10.7717/peerj.4428.

Lazer, D.M.J., Baum, M.A., Benkler, Y., Berinsky, A.J., Greenhill, K.M., Menczer, F., Metzger, M.J. ... and Zittrain, J.L. (2018). The science of fake news. Science, 359(6380); 1094-1096.

Leitzmann, C. (2014). Vegetarian nutrition: past, present, future. The American journal of clinical nutrition, 100(Supplement 1), 496S-502S. Retrieved 18th November 2017 from http://ajcn.nutrition.org/content/100/Supplement_1/496S.short

Lesk, C., Rowhani, P. and Ramankutty, N. (2016). Influence of extreme weather disasters on global crop production. Nature, 529 (2016); 84-87.

Lewis, K. E. and Grebitus, C. (2016). Why U.S. consumers support country of origin labeling: Examining the impact of ethnocentrism and food safety. Journal of International Food & Agribusiness Marketing, 28(3), pp. 1-17.

Lim, K. H., Hu, W., Maynard, L. J. and Goddard, E. (2014). A taste for safer beef? How much does consumers' perceived risk influence willingness to pay for country-of-origin labeled beef. Agribusiness, 30(1), pp. 17-30.

Lopez, R. A., He, X., & De Falcis, E. (2017). What Drives China's New Agricultural Subsidies?. World Development, 93, 279-292. Retrieved 24th November 2017 from http://www.sciencedirect.com/science/article/pii/S0305750X1630211X.

Lorencová, E., Frélichová, J., Nelson, E., & Vačkář, D. (2013). Past and future impacts of land use and climate change on agricultural ecosystem services in the Czech Republic. Land Use Policy, 33, 183-194. Retrieved 29th November 2017 from http://www.sciencedirect.com/science/article/pii/S0264837712002578.

Lu, Y. (2019). Artificial Intelligence: A survey on evolution, models, applications and future trends. Journal of Management Analytics, 6(1); 1-29.

Lusk, J. L. and Briggeman, B. C. (2009). Food values. American Journal of Agricultural Economics, 91(1), pp. 184-196.

Lynch, H., Johnston, C. and Wharton, C. (2018). Plant-Based Diets: Considerations for environmental impact, protein quality, and exercise performance. Nutrients, 10(12); 1841.

Maes, J., Liquete, C., Teller, A., Erhard, M., Paracchini, M.L., Barredo, J.I., Grizzetti, B. ... and Lavalle, C. (2016). An indicator framework for assessing ecosystem services in support of the EU Biodiversity Strategy to 2020. Ecosystem Services, 17 (2016); 14-23.

Manaaki Whenua (Landcare Research) (2019). What is Matauranga Maori? [website]. Accessed December 11th 2019 from https://www.landcareresearch.co.nz/about/sustainability/voices/matauranga-maori/what-

is-matauranga-maori. Martino, V. (2019). Why direct-to-consumer will become imperative for food companies [website]. Accessed December 11th 2019 from https://www.just-food.com/analysis/why-

direct-to-consumer-will-become-imperative-for-food-companies_id140731.aspx.

Mazur, M., Wisniewski, A. and McMillan, J. (2016). Clarity from above: PwC Global report on the commercial applications of drone technology. Warsaw: Drone Powered Solution, PriceWaterhouseCoopers.

McDonald, L. (2019). Ōtākaro Orchard edible garden building to start after \$1m-plus fundraising campaign [website]. Accessed December 3rd 2019 from https://www.stuff.co.nz/national/112858761/tkaro-orchard-edible-garden-building-to-start-after-1mplus-fundraising-campaign.

McFarland, M. (2017). Farmers turn to artificial intelligence to grow better crops. Retrieved 1st November 2017 from http://money.cnn.com/2017/07/26/technology/future/farming-ai-tomatoes/index.html

McKean, B. (2019). Why Food Brands Don't Sell Direct-to-Consumer, and What That Means for You [website]. Accessed December 11th 2019 from https://medium.com/@benjaminmckean/why-food-brands-dont-sell-direct-to-consumer-and-what-that-means-for-you-36bf73a26084.

Memphis Meats. (2017). Our Story. Retrieved 7th November 2017 from http://www.memphismeats.com/about-us/.

Merfield, C.N. (2019). An Analysis and Overview of Regenerative Agriculture. The BHU Future Farming Centre, Permanent Agriculture and Horticulture Science and Extension. Retrieved 2nd December 2019 from https://www.bhu.org.nz/future-farmingcentre/ffc/information/misc/an-analysis-and-overview-of-regenerative-agriculture-2019-ffcmerfield.pdf.

Mika, J.P. (2018). Strong sense of cultural identity drives boom in Māori business [website]. Accessed 19th November 2019 from https://theconversation.com/strong-sense-of-cultural-identity-drives-boom-in-maori-business-87500.

Miller, S., Driver, T., Saunders, C. and Dalziel, P. (2016a). High Value Nutrition: Country of Origin Literature Review. AERU Client Report prepared for the High Value Nutrition National Science Challenge. Lincoln University: Agribusiness and Economics Research Unit.

Ministry for the Environment (MfE) (2017a). New Zealand's 2030 Climate Change Target: Ministry for the Environment. Retrieved 14th November 2017 from http://www.mfe.govt.nz/climate-change/reducing-greenhouse-gas-emissions/New-Zealand%E2%80%99s-post-2020-climate-change-target.

Ministry for the Environment (MfE) (2017b). New Zealand Emissions Trading Scheme. Retrieved 23rd November 2017 from http://www.mfe.govt.nz/climate-change/reducinggreenhouse-gas-emissions/new-zealand-emissions-trading-scheme.

Ministry for the Environment (MfE) (2017c). Four Year Plan, 2017-2021. Retrieved 27th November 2019 from https://www.mfe.govt.nz/sites/default/files/media/About/Four%20year%20plan-2017.pdf.

Ministry for the Environment (MfE) (2019a). Climate Change Response (Zero Carbon) Amendment Bill: Summary. Retrieved 18th November 2019 from https://www.mfe.govt.nz/sites/default/files/media/Climate%20Change/climate-changeresponse-zero-carbon-amendment-bill-summary.pdf. Ministry for the Environment (MfE) (2019b). Environment Aotearoa 2019 Summary [website]. Accessed 27th November 2019 from https://www.mfe.govt.nz/environment-aotearoa-2019-summary.

Ministry for the Environment (MfE) (2019c). About the New Zealand Emissions Trading Scheme [website]. Accessed December 2nd 2019 from https://www.mfe.govt.nz/climate-change/new-zealand-emissions-trading-scheme/about-nz-ets.

Ministry for the Environment (MfE) (2019d). Action on agricultural emissions [website]. Accessed December 2nd 2019 from https://www.mfe.govt.nz/consultation/action-agricultural-emissions.

Ministry for the Environment (MfE) (2019e). About the National Policy Statement for Freshwater Management [website]. Accessed December 2nd 2019 from https://www.mfe.govt.nz/fresh-water/national-policy-statement/about-nps.

Ministry for the Environment (MfE) (2019f). Action for healthy waterways: Our proposals, your views [website]. Accessed December 2nd 2019 from https://www.mfe.govt.nz/consultation/action-for-healthy-waterways.

Ministry of Agriculture (MoA) (2019). Coversheet: Farm Debt Mediation. Retrieved 10th December 2019 from https://www.mpi.govt.nz/dmsdocument/35433/direct.

Ministry of Foreign Affairs and Trade (MFAT) (N.D.). Sustainable Development Goals [website]. Accessed December 2nd 2019 from https://www.mfat.govt.nz/en/peace-rights-and-security/work-with-the-un-and-other-partners/new-zealand-and-the-sustainable-development-goals-sdgs/.

Ministry of Foreign Affairs & Trade (MFAT) (2017a). Annual Report 2016-17. Retrieved 11th November 2017 from https://www.mfat.govt.nz/assets/MFAT-Corporate-publications/MFAT-Annual-Report-2016-17.pdf.

Ministry of Foreign Affairs & Trade (MFAT) (2017b). Trade Agenda 2030: Securing our place in the world. Retrieved 11th November 2017 from https://www.mfat.govt.nz/assets/Trade2030/Trade-Agenda-2030-Strategy-document.pdf.

Ministry of Foreign Affairs and Trade (MFAT) (2017c). Free Trade Agreements. Retrieved 3rd November 2017 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/.

Ministry of Foreign Affairs and Trade (MFAT) (2018). Te Ōhanga Māori: The Māori Economy. Retrieved 19th November 2019 from https://www.mfat.govt.nz/assets/FTAs-innegotiations/UK-FTA/The-Maori-Economy_2.pdf.

Ministry of Foreign Affairs and Trade (MFAT) (2019a). Free trade agreements in force [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/free-trade-agreements-in-force/.

Ministry of Foreign Affairs and Trade (MFAT) (2019b). Free trade agreements under negotiation [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/agreements-under-negotiation/.

Ministry of Foreign Affairs and Trade (MFAT) (2019c). Comprehensive and Progressive Agreement for Trans-Pacific Partnership [website]. Accessed 25th November 2019 from

https://www.mfat.govt.nz/en/trade/free-trade-agreements/free-trade-agreements-in-force/cptpp/cptpp-overview/.

Ministry of Foreign Affairs and Trade (MFAT) (2019d). ASEAN-Australia-New Zealand Free Trade Agreement Overview [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/free-trade-agreements-inforce/asean-australia-new-zealand-free-trade-agreement-aanzfta/aanzfta-overview/.

Ministry of Foreign Affairs and Trade (MFAT) (2019e). Trans-Pacific Strategic Economic Partnership (P4) [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/free-trade-agreements-inforce/p4/.

Ministry of Foreign Affairs and Trade (MFAT) (2019f). NZ-Gulf Cooperation Council FTA [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/free-trade-agreements-concluded-but-not-in-force/gcc/.

Ministry of Foreign Affairs and Trade (MFAT) (2019g). PACER Plus [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/free-trade-agreements-concluded-but-not-in-force/pacer/.

Ministry of Foreign Affairs and Trade (MFAT) (2019h). RCEP Overview [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/agreements-under-negotiation/regional-comprehensive-economic-partnership-rcep/rcep-overview/.

Ministry of Foreign Affairs and Trade (MFAT) (2019i). New Zealand-Pacific Alliance Free Trade Agreement Overview [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/agreements-undernegotiation/new-zealand-pacific-alliance-free-trade-agreement/new-zealand-pacificalliance-free-trade-agreement-overview/.

Ministry of Foreign Affairs and Trade (MFAT) (2019j). Russia-Belarus, Kazakhstan Customs Union FTA [website]. Accessed 25th November 2019 from https://www.mfat.govt.nz/en/trade/free-trade-agreements/agreements-undernegotiation/rbk/.

Ministry for Primary Industries (MPI) (2017). Growing Exports. Retrieved 1st November 2017 from https://www.mpi.govt.nz/exporting/overview/growing-exports/

Ministry for Primary Industries (MPI) (2018). Situation and Outlook for Primary Industries: December 2018. Retrieved 7th October 2019 from https://www.mpi.govt.nz/dmsdocument/32260/send.

Ministry for Primary Industries (MPI) (2019a). Valuing highly productive land: a summary. A summary of the proposed national policy statement for highly productive land. Ministry for Primary Industries and Ministry for the Environment. Retrieved December 2nd 2019 from https://www.mpi.govt.nz/dmsdocument/36621-valuing-highly-productive-land-a-summary.

Ministry for Primary Industries (MPI) (2019b). Māori Agribusiness Extension (MABx) Programme [website]. Accessed December 3rd 2019 from https://www.mpi.govt.nz/funding-and-programmes/maori-in-the-primary-industries/maoriagribusiness/maori-agribusiness-extension-mabx-programme/. Ministry for Primary Industries (MPI) (2019c). Māori Agribusiness Pathway to Increased Productivity (MAPIP) Programme [website]. Accessed December 3rd 2019 from https://www.mpi.govt.nz/funding-and-programmes/maori-in-the-primary-industries/maoriagribusiness/maori-agribusiness-pathway-to-increased-productivity-mapip-programme/.

Ministry for Primary Industries (MPI) (2019d). Farm Debt Mediation Bill [website]. Accessed December 10th 2019 from https://www.mpi.govt.nz/law-and-policy/legislation/farm-debt-mediation-bill/.

Moffat, K., Lacy, J., Zhang, A. and Leipold, S. (2016). The social license to operate: a critical review. Forestry: An International Journal of Forest Research, 89(5); 477-488.

Mortlock, B. and Hunt, L. (2008). Linking farmer wellbeing and environmentally sustainable land use: A comparison between converting organic and conventional dairy farmers. ARGOS Research Report No. 08/03. Agriculture Research Group on Sustainability.

New Zealand Agricultural Greenhouse Gas Research Centre (NZAGRC) (2012). Impacts of Global Climate Change on New Zealand Agriculture. Palmerston North: New Zealand Agricultural Greenhouse Gas Research Centre. Retrieved 21st November 2017 from https://www.nzagrc.org.nz/user/file/96/2_Impacts%20of%20Global%20Climate%20Change %20on%20New%20Zealand%20Agriculture%20-%20Fact%20Sheet.pdf.

New Zealand Emissions Trading Scheme (2010).

New Zealand Institute of Economic Research (NZIER) (2017). The benefits of trade. A report for Export New Zealand. Retrieved 11th December from https://www.exportnz.org.nz/__data/assets/pdf_file/0018/128025/Benefits-of-Trade.pdf

New Zealand Treasury (2018). Monthly Economic Indicators, August 2018 [website]. Accessed 26th November 2019 from https://treasury.govt.nz/publications/mei/monthly-economic-indicators-august-2018-html#section-3.

New Zealand Winegrowers (2017a). Annual Report 2017. Accessed 4 September 2017 at https://www.nzwine.com/en/news-media/statistics-reports/new-zealand-winegrowers-annual-report/.

New Zealand Winegrowers (2017b). "A Land Like No Other." Pamphlet accessed 4 September 2017 at https://www.nzwine.com/media/6390/a-land-like-no-other.pdf.

New Zealand Winegrowers (2017c). Sustainability Report 2016. Accessed 4 September 2017 at https://www.nzwine.com/en/sustainability/sustainability-report/.

Nicol-Williams, K. (2019). NZ will fall behind unless archaic gene editing law is updated, scientists say [website]. Accessed 20th November 2019 from https://www.tvnz.co.nz/one-news/new-zealand/nz-fall-behind-unless-archaic-gene-editing-law-updated-scientists-say.

Nielsen (2019). The F Word: Flexitarian is not a curse to the meat industry [website]. Accessed 21st November 2019 from

https://www.nielsen.com/us/en/insights/article/2019/the-f-word-flexitarian-is-not-a-curse-to-the-meat-industry/.

Niiler, E. (2018). Why Gene Editing Is The Next Food Revolution [website]. National Geographic. Accessed November 26th 2019 from

https://www.nationalgeographic.com/environment/future-of-food/food-technology-gene-editing/.

OECD-FAO (2019). OECD-FAO Agricultural Outlook 2019-2028, OECD Publishing, Paris/Food and Agriculture Organization of the United Nations, Rome.

O'Hara, Y. (2010). Down on the Farm: Depression and mental health in the rural South. A Southern Rural Life/Courier Country Publication, September/October 2010. Retrieved December 9th 2019 from https://www.mentalhealth.org.nz/assets/Uploads/down-on-the-farm-in-the-rural-south.pdf.

Ortega, D. L., Wang, H. H. and Olynk Widmar, N. J. (2014). Aquaculture imports from Asia: An analysis of U.S. consumer demand for select food quality attributes. Agricultural Economics, 45(5), pp. 625-634.

Our Land and Water National Science Challenge (OLW) (2018). Wai Ora, Whenua Ora, Tangata Ora: Strategy for the Our Land and Water National Science Challenge. Retrieved December 3rd 2019 from https://ourlandandwater.nz/wpcontent/uploads/2019/02/OLWStrategy2019-2024.pdf.

Patterson, M.G. and Cole, A.O. (2013). 'Total Economic Value' of New Zealand's land-based ecosystems and their services. In Dymond J.R. (ed.), Ecosystem services in New Zealand – conditions and trends. Manaaki Whenua Press, Lincoln, New Zealand.

Payne, P. and Ryan, A. (2019). Insects as mini-livestock? A study of New Zealand attitudes towards insect consumption. Report for AgResearch: Client Report No. RE450/2019/034. Retrieved 21st November 2019 from https://www.agresearch.co.nz/assets/agresearch-report-insects-as-mini-livestock-june-2019.pdf.

Pinstrup-Andersen, P. (2017). Is it time to take vertical indoor farming seriously? Global Food Security.

Poore, J. and Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. Science, 360(6392); 987-992.

Prentice, C. W., Han. (2017). Primary Production and Global Environmental Change. Retrieved 2nd December 2017 from https://www.imperial.ac.uk/grantham/ourwork/impacts-and-adaptation/ipcc-working-group-ii/food-security/primary-production-andglobal-environmental-change/.

PriceWaterhouseCooper (PwC) (2017a). The World in 2050: How will the global economic order change? Retrieved 12th November 2019 from https://www.pwc.com/gx/en/world-2050/assets/pwc-world-in-2050-slide-pack-feb-2017.pdf.

PriceWaterhouseCooper (PwC) (2017b). Futures of food provision: Four possible scenarios for the AgriFood industry. Retrieved December 3rd 2019 from https://www.pwc.co.nz/pdfs/2017pdfs/futures-of-food-provision.pdf.

Puranen, N., & Jansson, M. (2017). Alternative Food Networks and Social Media in Marketing: A multiple case study exploring how Alternative Food Networks use social media in order to help small local food producers reach the market. Accessed 6th December 2017 from http://www.diva-portal.org/smash/record.jsf?pid=diva2:1077115. Qi, L. (2019). Fighting Food Colonialism in the Hopi Nation [website]. Civil Eats. Accessed December 3rd 2019 from https://civileats.com/2019/11/20/fighting-food-colonialism-in-the-hopi-nation/.

Radio New Zealand (RNZ) (2018). NZ meat industry confident despite protein threat [website]. Accessed December 10th 2019 from

https://www.rnz.co.nz/news/country/351643/nz-meat-industry-confident-despite-protein-threat.

Radio New Zealand (RNZ) (2019a). Supermarket shoppers hunt down plant-based proteins [website]. Accessed 21st November 2019 from

https://www.stuff.co.nz/business/farming/116565410/supermarket-shoppers-hunt-down-plantbased-proteins.

Radio New Zealand (RNZ) (2019b). Countdown says customers moving to plant-based proteins [website]. Accessed 2nd December 2019 from

https://www.rnz.co.nz/news/country/400942/countdown-says-customers-moving-to-plant-based-proteins.

Ravikanth, L., Jayas, D.S., White, N.D.G., Field, P.G. and Sun, D-W. (2017). Extraction of Spectral Information from Hyperspectral Data and Application of Hyperspectral Imaging for Food and Agricultural Products. Food and Bioprocess Technology, 10(1); 1-33.

Reinicke, C. (2019). Beyond Meat insiders are rushing to sell after the company's post-IPO lockup expired - here's how much each sold for [website]. Accessed 26th November 2019 from https://www.businessinsider.com.au/beyond-meat-insiders-sell-profit-since-post-ipo-lockup-expiry-2019-11?r=US&IR=T.

Reserve Bank of New Zealand (RBNZ) (2019). Financial Stability Report, May 2019. Retrieved December 9th 2019 from https://www.rbnz.govt.nz/-/media/ReserveBank/Files/Publications/Financial%20stability%20reports/2019/fsr-may19.pdf?revision=47e0d60a-bdca-4fbb-bddc-2ad9f20a4b2d.

Resource Management Act 1991.

Response Media (2017). Transparency Content and the Consumer Journey. Retrieved December 11th 2019 from http://www.responsemedia.com/wp-content/uploads/2017/07/RM_Transparency_Survey_Final.pdf.

Rigaud, K.K., de Sherbinin, A., Jones, B., Bergmann, J., Clement, V., Ober, K., Schewe, J. ... and Midgley, A. (2018). Groundswell: Preparing for internal climate migration. Retrieved 27th November 2019 from

https://openknowledge.worldbank.org/bitstream/handle/10986/29461/WBG_ClimateChan ge_Final.pdf.

Roberts, L., Brower, A., Kerr, G., Lambert, S., McWilliam, W., Moore, K., Quinn, J. ... and Wratten, S. (2015). The nature of wellbeing: How nature's ecosystem services contribute to the wellbeing of New Zealand and New Zealanders. Department of Conservation. Retrieved 27th November 2019 from https://www.doc.govt.nz/Documents/science-andtechnical/sap258entire.pdf. Rodale Institute (2013). Regenerative Organic Agriculture and Climate Change: A Down-to-Earth Solution to Global Warming. Retrieved 2nd December 2019 from https://rodaleinstitute.org/wp-content/uploads/rodale-white-paper.pdf.

Roodhuyzen, D.M.A., Luning, P.A., Fogliano, V. and Steenbekkers, L.P.A. (2017). Putting together the puzzle of consumer food waste: Towards an integral perspective. Trends in Food Science & Technology, 68; 37-50.

Roques, A. (2010). Alien forest insects in a warmer world and a globalised economy: impacts of changes in trade, tourism and climate on forest biosecurity. New Zealand Journal of Forestry Science, 40 (2010); S77-S94.

Rose, J. (2017). Brexit, Trump and Post-Truth Politics. Public Integrity, 19(6); 555-558.

Rosi, A., Mena, P., Pellegrini, N., Turroni, S., Neviani, E., Ferrocino, I., Di Cagno, R. ... and Scazzina, F. (2017). Environmental impact of omnivorous, ovo-lacto-vegetarian, and vegan diet. Scientific Reports, 7; 6105.

Rotorua Economic Development (RED) (2017). Indigenous Food 2018 – Rotorua Market Opportunity. Retrieved December 3rd 2019 from https://www.rotoruanz.com/RNZ/media/Media-Library/Business/Do%20Business/PDFs/Rotorua-Indigenous-Food-Analysis.pdf.

Royal Society of New Zealand (RSNZ) (2016). Revolution occurring in gene editing has significant implications for New Zealand. Royal Society of New Zealand. Retrieved 10th November 2017 from https://royalsociety.org.nz/news/revolution-occurring-in-gene-editing-has-significant-implications-for-new-zealand/.

Royal Society of New Zealand (RSNZ) (2019). Calls for overhaul of gene-technology regulations and wide public discussion [website]. Accessed 20th November 2019 from https://www.royalsociety.org.nz/news/calls-for-overhaul-of-gene-technology-regulations-and-wide-public-discussion.

Sakure, K. and Manepalli, N. (2019). Vegan and flexitarian population drive global plantbased meat demand [website]. Accessed 2nd December 2019 from https://www.newfoodmagazine.com/article/86234/vegan-flexitarian-plant-based-meatdemand/.

Saunders, C. and Barber, A. (2008). Carbon footprints, life cycle analysis, food miles: Global trade trends and market issues. Political Science, 60(1), pp. 73-88.

Saunders, C., Barber, A. and Taylor, G. (2006) Food Miles – Comparative Energy/Emissions Performance of New Zealand's Agriculture Industry. AERU Research Report No. 285. Lincoln University: Agribusiness and Economics Research Unit.

Saunders, C.M., Kaye-Blake, W. and Turner, J. (2009). Modelling climate change impacts on agriculture and forestry with the extended LTEM (Lincoln Trade and Environment Model). AERU Research Report No. 316, December 2009. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Saunders, C., Guenther, M., Driver, T., Tait, P., Dalziel, P. and Rutherford, P. (2015a). Consumer Attitudes to New Zealand Food Attributes and Technology Use in Key International Markets. AERU Research Report No. 333. Lincoln University: Agribusiness and Economics Research Unit.

Saunders, C., Guenther, M., Tait, P., Saunders, J., Dalziel, P. and Rutherford, P. (2015b). Consumer preferences in developing and developed country markets and the impact on EU and New Zealand. Asia Pacific Journal of EU Studies, 13(2): pp. 1-24.

Saunders, C. M., Dalziel, P. C., Guenther, M., Saunders, J. and Rutherford, P. (2016a). The land and the brand. Lincoln University. AERU. Retrieved 26th November 2017 from http://researcharchive.lincoln.ac.nz/bitstream/handle/10182/6920/AERU%20RR%20339%20 Land%20and%20the%20Brand%20Report.pdf?sequence=3&isAllowed=y

Saunders, C., Driver, T., Mowat, A., Kaye-Blake, B., Payn, T., Bayne, K., Saunders, J. and Tait, P. (2016b). Driving better programme investment and accelerating challenge impact through a prioritisation matrix of international and national perspectives. Retrieved 2nd November 2017 from http://ourlandandwater.nz/the-challenge/greater-value-in-global-markets.

Saunders, C., Duff, S. and Driver, T. (2018). Driving Better Programme Investment and Accelerating Challenge Impact through a Priorisation Matrix of International and National Perspectives: The Matrix. Retrieved December 11th 2019 from https://ourlandandwater.nz/wp-

 $content/uploads/2019/03/Matrix_Saundersetal 2018 Drivers Project Update.pdf.$

Saunders, J. (2016). Trade Implications for New Zealand Agriculture with Price Premiums for Credence Attributes in Food and Beverages. AERU Research Report No. 338. Lincoln University: Agribusiness and Economics Research Unit.

Saunders, J., Saunders, C., McLellan, B., Obadovic, I. and Driver, T. (2016c). Modelling Agricultural Impacts of EU-NZ Trade Liberalisation. AERU Research Report No. 341, Lincoln University.

Saunders, J., Guenther, M. and Saunders, C. (2019). The impacts of changes in agricultural policies in the United Kingdom on trade and agriculture, especially in New Zealand. AERU Discussion Research Report Series [forthcoming], September 2019. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Sausman, C., Garcia, M., Fearne, A., Felgate, M., Mekki, A. A. E., Cagatay, S., Soliman, I., Thabet, B., Thabet, M. B. S., Laajimi, A., Ashkar, H. A., El Hadad-Gauthier, F., Mili, S. and Martínez, C. (2015). From Value Chain Analysis to Global Value Chain Analysis: Fresh Orange Export Sector in Mediterranean Partner Countries. In M. Petit, E. Montaigne, F. El Hadad-Gauthier, M. J. García Álvarez-Coque, K. Mattas and S. Mili (Eds.) Sustainable Agricultural Development: Challenges and Approaches in Southern and Eastern Mediterranean Countries. Cham: Springer International Publishing, pp. 197-225.

Sautier, M., Legun, K. and Campbell, H. (2015). Adoption and outcomes of sustainable winegrowing practices in New Zealand. Poster prepared for a NZSD Stakeholder Workshop, 6 August, accessed 4 September 2017 at http://www.nzdashboard.org.nz/ adoption.html.

Scarlatti (2017). Health and Wellness of New Zealand Farmers: Initial findings from research into the wellness of, and wellness behaviour patterns of, New Zealand farmers. Retrieved December 9th 2019 from https://www.growingnz.org.nz/health-and-wellness-of-new-

zealand-farmers_Initial-findings-from-research_Into-the-wellness-of-and-wellness-behaviour-patterns-of-new-zealand-farmers-research_paper-27.html.

Sexton, A.E., Garnett, T. and Lorimer, J. (2019). Framing the future of food: The contested promises of alternative proteins. Environment and Planning E: Nature and Space, 2(1); 47-72.

Shanker, D. (2019). These \$50 chicken nuggests were grown in a lab. Bloomberg. Accessed 21st November 2019 from https://www.bloomberg.com/news/articles/2019-10-22/clean-meat-just-chicken-nuggets-grown-in-a-lab-coming-soon.

Shaw, A. (2017). Kiwi's waste \$1.8b of food a year. Retrieved 1st November 2017 from http://www2.nzherald.co.nz/the-country/news/article.cfm?c_id=16&objectid=11938608

Smith, M.J. (2018). Getting value from artificial intelligence in agriculture. Animal Production Science, https://doi.org/10.1071/AN18522.

Sogari, G., Liu, A. and Li, J. (2019). Understanding Edible Insects as Food in Western and Eastern Societies. In Environmental, Health, and Business Opportunities in the New Meat Alternatives Market (pp. 166-181). IGI Global.

Soon, S. (2019). Blockchain could tell you everything about your groceries. Here's what it may cost [website]. Accessed 18th November 2019 from https://www.cnbc.com/2019/11/15/blockchain-may-be-coming-to-food-aisles-heres-what-it-may-cost.html.

South, P.F., Cavanaugh, A.P., Liu, H., and Ort, D.R. (2019). Synthetic glycolate metabolism pathways stimulate crop growth and productivity in the field. Science, 363 (6422), eaat9077.

Specht, L., Weith, T., Swoboda, K. and Siebert, R. (2016). Socially acceptable urban agriculture businesses. Agronomy for Sustainable Development, 36(17); https://doi.org/10.1007/s13593-016-0355-0.

Springmann, M., Godfray, H.C.J., Rayner, M and Scarborough, P. (2016). Analysis and valuation of the health and climate change cobenefits of dietary change. PNAS Early Edition, DOI:10.1073/pnas.1523119113.

Statistics New Zealand (2012). Using National Accounts Input-Output Tables. Wellington: Statistics New Zealand.

Statistics New Zealand (StatsNZ) (2019a). 2018 Census population and dwelling counts [website]. Accessed December 12th 2019 from https://www.stats.govt.nz/information-releases/2018-census-population-and-dwelling-counts.

Statistics New Zealand (StatsNZ) (2019b). March exports hit a record \$5.7 billion [website]. Accessed 12th November 2019 from https://www.stats.govt.nz/news/march-exports-hit-a-record-5-7-billion.

Statistics New Zealand (StatsNZ) (2019c). Gross domestic product: June 2019 quarter: Gross domestic product by industry – annual values.

Stephens, N., Di Silvio, L., Dunsford, I., Ellis, M., Glencross, A. and Sexton, A. (2018). Bringing cultured meat to market: Technical, socio-political, and regulatory challenges in cellular agriculture. Trends in Food Science and Technology, 78 (2018); 155-166.

Sunfed (2019). Sunfed Foods [website]. Accessed 21st November 2019 from https://sunfedfoods.com/.

Supply and Demand Chain Executive (SDCE) (2019). Walmart Canada Unveils Largest Blockchain for Supply Chain Management [website]. Accessed 18th November 2019 from https://www.sdcexec.com/software-technology/news/21097675/walmart-canada-unveilslargest-blockchain-for-supply-chain-management.

Sutton, K., Larsen, N., Moggre, G-J., Huffman, L., Clothier, B., Eason, J. and Bourne, R. (2018). Opportunities in Plant-Based Foods: Protein. Plant & Food Research report prepared for Ministry of Primary Industries and Plant & Food Research. SPTS No. 15748.

Sylvester, G. (2018). E-agriculture in action: Drones for agriculture. Published by the Food and Agriculture Organization of the United Nations (FAO) and International Telecommunication Union (ITU), Bangkok.

Tait, P.R., Rutherford, P., Driver, T., Li, X., Saunders, C., Dalziel, P. and Guenther, M. (2018a). Consumer insights and willingness to pay for attributes: Kiwifruit in Shanghai, China. AERU Research Report No. 346. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Tait, P.R., Rutherford, P., Driver, T., Li, X., Saunders, C., Dalziel, P. and Guenther, M. (2018b). Consumer insights and willingness to pay for attributes: New Zealand wine in California, USA. AERU Research Report No. 349. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Tait, P.R., Rutherford, P., Driver, T., Li, X., Saunders, C., Dalziel, P. and Guenther, M. (2018c). Consumer insights and willingness to pay for attributes: New Zealand beef products in California, USA. AERU Research Report No. 348. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Tait, P.R., Rutherford, P., Driver, T., Li, X., Saunders, C., Dalziel, P. and Guenther, M. (2018d). Consumer insights and willingness to pay for attributes: New Zealand yogurt products in Shanghai, China. AERU Research Report No. 347. Agribusiness and Economics Research Unit, Lincoln University: Lincoln, New Zealand.

Taunton, E. (2017). Kiwifruit industry bounces back from 'heartbreak' of Psa [website]. Accessed 26th November 2019 from

https://www.stuff.co.nz/business/farming/93931853/kiwifruit-industry-bounces-back-from-heartbreak-of-psa.

Te Hono (2017a). Te Hono: Unlocking the Power of New Zealand's Primary Sector [website]. Accessed September 17th 2017 from https://www.tehono.co.nz/.

Te Hono (2017b). Our Story [website]. Accessed September 17th 2017 from https://www.tehono.co.nz/our-story.

Te Hono (2018). Stanford Bootcamp [website]. Accessed September 17th 2017 from https://www.tehono.co.nz/stanford-bootcamp.

Teague, W.R., Apfelbaum, S., Lal, R., Kreuter, U.P., Rowntree, J., Davies, C.A., Conser, R. ... and Byck, P. (2016). The role of ruminants in reducing agriculture's carbon footprint in North America. Journal of Soil and Water Conservation, 71(2); 156-164. Teece, D. J. (1982). Towards an economic theory of the multiproduct firm. Journal of Economic Behavior & Organization, 3(1), pp. 39-63.

Teece, D. J. (2015) Injecting Dynamic Capabilities Thinking into the New Zealand Primary Sector. PowerPoint presentation to the 2015 Stanford University New Zealand Primary Sector Bootcamp, 2 July.

Teece, D. J. (2017a). Towards a capability theory of (innovating) firms: Implications for management and policy. Cambridge Journal of Economics, 41(3), pp. 693-720.

Teece, D. J. (2017b). A capability theory of the firm: An economics and (strategic) management perspective. New Zealand Economic Papers.

Television New Zealand (TVNZ) (2019). Extinction Rebellion climate change protest underway in Wellington [website]. Accessed December 2nd 2019 from https://www.tvnz.co.nz/one-news/new-zealand/extinction-rebellion-climate-changeprotest-underway-in-wellington.

Temming, M. (2019). A new way to genetically tweak photosynthesis boosts plant growth [website]. Accessed 20th November 2019 from https://www.sciencenews.org/article/new-way-genetically-tweak-photosynthesis-boosts-plant-growth.

Trice, R. (2017). Can Artificial Intelligence Help Feed The World? Retrieved 31st November 2017 from https://www.forbes.com/sites/themixingbowl/2017/09/05/can-artificial-intelligence-help-feed-the-world/#69aefe846dbb

UMR Research (2017). New Zealanders' views of the primary sector. A report prepared for Ministry for Primary Industries, October 2017. Retrieved 19th November 2019 from https://www.mpi.govt.nz/dmsdocument/27582/direct.

UMR Research (2019). Opinion of Primary Industries. September 2019. Retrieved 19th November 2019 from https://umr.co.nz/wp-content/uploads/2019/09/report-opinion-of-primary-industries-sept-2019.pdf.

United Nations (UN) (2018). 2018 International trade Statistics Yearbook: Volume 1 Trade by Country. Department of Economic and Social Affairs, Statistics Division. (ST/ERA/STAT/SER.G/67 Vol.1).

United Nations (UN) (2019a). World Population Prospects 2019: Highlights. A report prepared by the Department of Economic and Social Affairs, Population Division (ST/ESA/SER.A/423).

United Nations (UN) (2019b). Sustainable Development Goals [website]. Accessed 19th November 2019 from https://www.un.org/sustainabledevelopment/sustainable-development-goals/.

United Nations (UN) (2019c). The Paris Agreement [website]. Accessed December 2nd 2019 from https://unfccc.int/process-and-meetings/the-paris-agreement/the-paris-agreement.

United Nations (UN) (2019d). The Sustainable Development Goals Report 2019. Retrieved December 2nd 2019 from https://unstats.un.org/sdgs/report/2019/The-Sustainable-Development-Goals-Report-2019.pdf.

United Nations Development Programme (UNDP) (2019). What are the Sustainable Development Goals? [website]. Accessed December 2nd 2019 from https://www.undp.org/content/undp/en/home/sustainable-development-goals.html.

USC. (2016). Tackling Non-Tariff Barriers in Agriculture and Food Trade within APEC: A Business Perspective. South Carolina: University of Southern Carolina. Retrieved 2nd December 2017 from

http://info.marshall.usc.edu/academic/gpp/abac/Documents/Exec%20Summary%20Final%2 0Layout%20Version.pdf.

Verevis, S. and Üngör, M. (2019). What Has New Zealand Gained From The FTA With China?: Two Counterfactual Analyses, University of Otago, Economics Discussion Papers No. 1906, April.

Victoria University of Wellington (VUW) (2017). Explaining the Extremes. Retrieved 13th November 2017 from https://www.victoria.ac.nz/news/2017/04/explaining-the-extremes.

Warren, L., Mika, J. and Palmer, F. (2017). How does enterprise assistance support Māori entrepreneurs? An identity approach. Journal of Management & Organisation, 23(6); 873-885.

World Economic Forum (WEF) (2016). Shaping the Future of Global Food Systems: A Scenarios Analysis: World Economic Forum. Retrieved 29th October 2017 from http://www3.weforum.org/docs/IP/2016/NVA/WEF_FSA_FutureofGlobalFoodSystems.pdf

Wilcock, A., Pun, M., Khanona, J., & Aung, M. (2004). Consumer attitudes, knowledge and behaviour: a review of food safety issues. Trends in Food Science & Technology, 15(2), 56-66.

Wirehn, L. (2018). Nordic agriculture under climate change: A systematic review of challenges, opportunities and adaptation strategies for crop production. Land Use Policy, 77 (September 2018); 63-74.

Wirth, F.F., Stanton, J.L. and Wiley, J.B. (2011). The relative importance of search versus credence product attributes: organic and locally grown. Agricultural and Resource Economics Review, 40; 48-62.

Wood, P. and Mika, J.P. (2018). Te Manu Ka Rere: Fostering Māori enterprise financial capability. Retrieved 19th November 2019 from

https://www.massey.ac.nz/massey/fms/Colleges/College%20of%20Business/School%20of% 20Economics%20&%20Finance/FinEd/documents/210844%20Fostering%20Maori%20Enterp rise%20Financial%20Capability%20Report%20v9%20HR.pdf?D0FA4FA5C6B8D77B63875EA1 8E84DE29.

World Integrated Trade Solution (WITS) (2019). New Zealand Trade Summary 2017 Data. World Bank. Retrieved 11th December from https://wits.worldbank.org/CountryProfile/en/Country/NZL/Year/LTST/Summary.

World Trade Organization (WTO) (2015). Ministerial Conference, Tenth Session: Nairobi, 15-19 December 2015. Retrieved 25th November 2019 from https://www.wto.org/english/thewto_e/minist_e/mc10_e/nairobipackage_e.pdf. Wyllie, A. (2019). Farmstrong Fourth Year Monitoring Report. Report prepared for Farmstrong, July 2019. Retrieved December 9th 2019 from https://farmstrong.co.nz/wp-content/uploads/2019/09/Monitor-report-2019-Final.pdf.

Xue, L. and Liu, G. (2019). Introduction to global food losses and food waste. In Saving Food (pp. 1-31). Academic Press.

Zespri (2016). Growing a Better, Healthier Future: Zespri and Sustainability. Retrieved September 6th 2017 from www.zespri.com/Documents/Zespri-Sustainability-Brochure.pdf.

Zhang, X., & Cai, X. (2011). Climate change impacts on global agricultural land availability. Environmental Research Letters, 6(1), 1-8. Retrieved 28th November 2017 from http://iopscience.iop.org/article/10.1088/1748-9326/6/1/014014/meta.

Appendix A: Review of international consumer preferences studies – choice experience (CE) and willingness-to-pay (WTP) case studies

It is important to value the range of premiums that international consumers are willing to pay for the inclusion of attributes in products. One method to assess this is the use of choice experiments. A choice experiment (CE) is an economic valuation method used to assess willingness-to-pay (WTP) for different attributes of goods or services that can (but does not have to be) traded in markets. This belongs to the category of stated preference non-market valuation methods (Hanley et al., 2013; Hensher et al., 2015). CE can be used to explore consumer preferences for attributes that do not currently exist in-market (Teratanavat and Hooker, 2006) for application in product development or market access, and to simulate real markets and the product choices involving trade-offs (Carlsson et al., 2005; Mueller Loose and Remaud, 2013; Poelmans and Rousseau, 2016).

This chapter updates a literature review of consumer WTP for a series of basic and credence attributes relating to the international and domestic drivers included in this report. This review complements previous large-scale literature reviews produced as part of the Maximising Export Returns (MER) research programme by Agribusiness and Economics Research Unit (Miller et al., 2014), as well as Stage 1 and 2 of the Drivers Project for the Our Land and Water component of the National Science Challenge (Saunders et al., 2016b; 2018), and covers mainly academic CE literature published between 2003 and 2019. Previous reviews identified food safety as a key credence attribute across all markets, including positive WTP with high associated premiums in some cases (e.g. food safety credentials on food products in China). This is understandable due to widespread public concerns regarding previous food safety incidents around the world. Moreover, some developing countries are experiencing rapid change such as growing populations and increased urbanisation - it is possible that these can also impact on consumer preferences.

Previous reviews also identified product quality (and associated indicators) as another popular credence attribute. Examples of this include the freshness of milk products or tenderness of steak products. Product quality can also extent to aspects of a product's origin, whereby a common finding is that people prefer domestically-produced over imported food products. There is also a range of case studies considering production methods, typically comparing organic, genetically modified (GM) and conventional production practices. Regarding GM production, evidence is mixed, while WTP for organic production (for dairy, fruit and vegetable, wine, oil and flour products) was found to be consistently positive. It has also been shown that consumers can associate organic foods with a range of benefits, such as increased healthiness and limited use of pesticides.

Similarly, functional foods (i.e. food products that offer health benefits beyond basic nutrition) have also shown some positive WTP. In China and Singapore, for example, there is growing interest in these types of products, such as those intended to enhance the immune system, supplement basic nutrition or assist with aspects of beauty, among other effects. Miller et al. (2014) includes limited empirical examples examining oil, bread, eggs and wine products.

Finally, the previous review found some evidence that consumers are concerned with environmental or animal welfare issues, particularly in relation to the ethical dimensions of production. For example, studies indicate that consumers in the UK, China and India are willing to pay for reduced water pollution, reduced greenhouse gas (GHG) emissions and improved biodiversity in agricultural production (Saunders et al., 2013), and for certified paper towels

associated with several environmental attributes in the USA (O'Brien and Teisl, 2004). Likewise, research has indicated that many consumers are concerned about the health and welfare of animals, potentially influencing their purchase decisions. The CE studies have included general animal welfare or free range attributes alongside other types of attributes related to animal health and welfare.

A1.1. Meat and seafood products

The current review includes 35 CE and other WTP studies examining the attributes of meat and seafood products in Europe, North America, Asia and other regions. The most commonly examined markets across these studies include Germany, the United Kingdom (UK), the United States (US) and China. Attributes examined in these studies include animal health and/or welfare, organic, different production methods, traceability, local food, country-oforigin, nutritional content, functional foods, social responsibility, environmental condition, certification, carbon/GHG emissions associated with production, water use and genetic modification (GM), as well as generic attributes such product quality, appearance and taste.

General studies

Clark et al. (2017) conducted a review of international WTP literature regarding farm animal welfare for pigs, chickens, cattle and fish. The authors estimated a weighted mean WTP (in Euros) for the provision of higher standards of farm animal welfare across a range of studies, measures and differences in WTP by type of production animal. As shown in Table A1, the authors found higher mean WTP for beef cows and fish compared to pigs and broiler chickens. This indicates that consumers prefer the provision of farm animal welfare depending on the type of animal involved in production.

Animal Type	No. of Measures	No. of Studies	Weighted Mean WTP (€)
Pig	90	13	0.54
Layer Hen 47		10	0.09
Broiler Chicken	26	8	1.24
Dairy Cow	27	7	0.50
Beef Cow	24	7	5.00
More than one type	6	2	11.20
Fish	6	3	3.53

Table A1: Willingness-to-pay for farm animal welfare, international literature review

Source: Clark et al., 2017.

European studies

The current review includes 16 CE and other WTP studies examining the attributes of meat and seafood products in Europe, including studies conducted in Germany, Denmark, Portugal, Spain, France, UK, Sweden, Italy, Netherlands and Belgium. Attributes examined in these studies include animal health and/or welfare, organic, different production methods, traceability, local food, country-of-origin, nutritional content, functional foods, social responsibility, environmental condition, certification, carbon/GHG emissions associated with production, water use and genetic modification (GM), as well as generic attributes such product quality, appearance and taste.

Denver et al. (2017) conducted a WTP study to value Danish consumers' WTP for the provision of relative levels of animal welfare for pigs in pork production. The study was designed to assess consumers' WTP for trade-offs between standard, medium and high levels of animal welfare in production. Table A2 shows that there is a small difference between WTP for

medium and high levels, with many consumers not willing to pay additional premiums to move beyond the medium level of animal welfare.

		Stated WTP for welfare pork				
Attribute Level	Market price premiums	Respondents usually buying standard or medium level welfare pork	Respondents usually buying high level welfare pork			
Standard	0%	Base (WTP not estimated)				
Medium (relative to standard)	17-75% higher	80% higher	170% higher			
High (relative to medium)	14% higher	0% higher	15% higher			

Table A2: Willingness-to-pay for animal welfare in relation to pork, Denmark (N=396)

Source: Denver et al., 2017.

Risius and Hamm (2017) examined the effects of exposure to communication materials on German consumers' WTP for organic and animal husbandry attributes in relation to beef products. The authors tested consumer preferences and WTP for beef products before and after being shown communication materials regarding different animal husbandry and production methods. Prior to being shown material, participants indicated a preference for enhanced husbandry practices and organic production. Participants were then shown either an image film, a documentary film or a leaflet giving further information regarding each type of production method or husbandry practice (including organic production, extensive suckler cow husbandry and pasture-based husbandry). As shown in Table A3, following the presentation of this information, consumer preferences and WTP for each system changed based on the type of information presented.

Table A3: Willingness-to-pay (€) for organic and animal husbandry attributes following presentation of communication materials (image film, documentary film and leaflet), Germany (N=676)

Communication	Attributes					
Communication	Oragnia	Extensive suckler cow	Pasture-based			
material	Organic	husbandry	husbandry			
Image film	2.98	3.79	0.98			
Documentary film 2.67		5.93	0.27			
Leaflet	4.22	4.68	-0.31			

Source: Risius and Hamm, 2017.

Kallas et al. (2019) used a discrete choice experiment to determine Spanish consumers' WTP for health-enhancing properties in pork patty products before and after a hedonic taste test of product types. Specifically, this involved innovative pork patty products with enhanced health claims through the addition of Porcini (added dietary fibre) and blueberries (added antioxidants). Initially, the researchers determined the "food neophobia" (degree of aversion to innovative food products) of the participants, subsequently segmenting participants into three groups – low, average and high food neophobic (LN, AN and HN respectively). WTP values were calculated prior to and following taste testings of each of the products, deriving a range of premiums associated with each product – these are shown in Table A4 below. This showed a generally higher WTP for both traditional and innovative pork products by

consumers with lower food neophobia, as well as a perceived higher WTP prior to tasting for those innovative products including blueberries over Porcini (Kallas et al., 2019).

		WTP (€)	WTP (€)
Segment	Product Type	Expected	Experienced
		Before Tasting	After Tasting
Low Food Noonhohio	Traditional Pork Product	3.87	4.31
Low Food Neophobia (I,N) (n = 24)	Innovative Pork Product 1 – Porcini	3.60	2.70
(LIN) (II – 24)	Innovative Pork Product 2 – Blueberries	4.60	2.34
Average Food	Traditional Pork Product	3.71	3.38
Neophobia (AN)	Innovative Pork Product 1 – Porcini	3.50	2.79
(n = 41)	Innovative Pork Product 2 – Blueberries	3.71	1.86
High Food Neophobia	Traditional Pork Product	2.88	3.43
	Innovative Pork Product 1 – Porcini	2.88	2.41
(11) (11 – 55)	Innovative Pork Product 2 – Blueberries	3.34	1.89

Table A4. Willingness-to-pay (€) for traditional and innovative pork products before and after tasting, Spanish consumers (2018) (N = 121)

Source: Kallas et al., 2019.

Calvo Dopico et al. (2016) examined European fish consumers' (Portugal, Spain, France, UK and Germany) preferences and WTP for the provision of traceability information with fish products. Table A5 shows that while around half of participants stated that they would not be willing to pay a premium for this (particularly Portuguese and Spanish participants).

Country	Samala			WTP for traceability programme		
Country	Sample	WTP: NO	WIP: tes	Premium	% participants	
				€0-0.25	10.2	
				€0.26-0.50	8.8	
Spain	410	262 (63.9%)	148 (36.1%)	€0.51-0.75	6.3	
				€0.76-1	5.9	
				€>1	4.9	
				€0-0.25	9.93	
				€0.26-0.50	18.87	
UK	302	147 (48.68%)	155 (51.32%)	€0.51-0.75	9.27	
				€0.76-1	7.28	
				€>1	5.96	
	728	553 (75.96%)	175 (24.04%)	€0-0.25	7.69	
				€0.26-0.50	7.42	
Portugal				€0.51-0.75	4.67	
				€0.76-1	3.02	
				€>1	1.24	
				€0-0.25	14.93	
				€0.26-0.50	17.31	
France	335	160 (47.8%)	175 (52.2%)	€0.51-0.75	9.25	
				€0.76-1	7.46	
				€>1	3.28	
				€0–0.25	6.00	
				€0.26-0.50	21.33	
Germany	300	126 (42%)	174 (58%)	€0.51-0.75	16.00	
				€0.76-1	11.00	
				€>1	3.67	

Table A5: Willingness-to-pay for traceability programme, European countries

Source: Calvo Dopico et al., 2016.

Hempel and Hamm (2015) examined German consumers' preferences and WTP for organic and local attributes across a range of food products, including beef steak, butter, apples and flour products. Based on a series of questions regarding preferences for organic and local products, the authors segmented participants into two groups – organic-minded consumers (OMC) and non-organic-minded consumers (NOMC). Table A6 shows differences in WTP for local and organic attributes between OMC and NOMC, with both groups indicating the highest WTP for local beef steak products (as opposed to 'from a neighbouring country'.

	Organic-m	ninded consume	ers (N=221)	Non-organic-minded consumers (N=427)			
	Organic	Local (as opposed to "from Germany")	Local (as opposed to "from a neighbouring country")	Organic	Local (as opposed to "from Germany")	Local (as opposed to "from a neighbouring country")	
Apples (/kg)	1.22	0.63	4.25	-0.13	0.17	2.07	
Butter (/250g)	0.31	0.37	1.26	-0.01	0.12	0.56	
Flour (/kg)	0.97	0.36	3.44	-0.03	0.23	1.28	
Steak (/200g)	2.46	1.26	5.56	0.46	1.94	4.80	

Table A6: Willingness-to-pay	/ (€) foi	r organic and loo	cal attributes,	Germany	(N=638)
------------------------------	-----------	-------------------	-----------------	---------	---------

Source: Hempel and Hamm, 2015

Lagerkvist et al. (2017) examined Swedish consumers' WTP for a range of credence attributes in relation to beef products using a discrete choice experiment. Attributes included countryof-origin labelling, traceability to various parts of the supply chain, animal health and welfare, human health, social responsibility, and production methods. As shown by Table A7 below, participants indicated a range of positive WTP values for all attributes, particularly to move from basic to slightly improved levels (e.g. Price 1 to Price 2).

Table A7: Willingness-to-pay (SEK) for a range of attributes in beef products (discrete price
level), Sweden (N=440) (base price=200 SEK/kg)

Attribute	Price 2: 225 SEK/kg	Price 3: 250 SEK/kg	Price 4: 275 SEK/kg	Price 5: 300 SEK/kg	Price 6: 325 SEK/kg
Reference code	2.09	0.79	0.42	0.28	0.23
Traceability to specific slaughterhouse	1.46	0.55	0.30	0.20	0.16
Traceability to group or specific animal	2.00	0.75	0.41	0.27	0.22
Traceability to specific breeder	1.49	0.56	0.30	0.20	0.17
Animal welfare	2.89	1.09	0.59	0.39	0.32
Animal medication used for preventative purposes	2.52	0.95	0.51	0.34	0.28
Organic production	2.03	0.76	0.41	0.28	0.22
Environmental impact	1.68	0.63	0.34	0.23	0.19
Health impact	1.71	0.64	0.35	0.23	0.19
Social responsibility	1.96	0.74	0.40	0.27	0.22
Type of animal feed used	1.44	0.54	0.29	0.20	0.16

Source: Lagerkvist et al., 2017.

Balcombe et al. (2016) examined UK consumers' WTP for country-of-origin, production methods, product quality and certification attributes in 12 types of poultry, beef, pork and sheep meat products. Table A8 presents mean estimates of WTP for the range of products

and attributes mentioned above. Results show that participants were willing to pay a premium for each of the attributes across most products, with negative WTP uniformly shown for products of non-UK origin.

	Attributes							
Product Type	Choice*	Premium*	Organic	UK Origin	EU Origin	Origin Outside EU	Freedom Food Label	Intl. Quality Label
Pork sausages (/450g)	0.17	1.08	0.91	0.84	-0.27	-0.73	0.33	0.87
Pork joint (/1.5kg)	0.46	2.40	2.62	3.15	-1.09	-2.28	1.68	2.42
Beef lasagne (/600g)	0.87	2.55	1.92	1.68	-1.0	-0.71	0.96	1.68
Bacon (/300g)	0.35	0.88	0.93	0.67	-0.62	-1.04	0.6	0.85
Beef burger (/450g)	0.49	1.02	0.67	0.65	-0.77	-0.86	0.48	0.85
Chicken curry (/400g)	0.4	1.45	1.29	1.16	-0.41	-0.87	0.52	1.19
Leg lamb (/1.5kg)	0.5	1.69	2.03	2.85	-2.62	0.03	1.68	1.43
Chicken breasts (/500g)	0.63	1.4	2.06	2.23	-0.38	-1.99	1.41	1.7
Pepperoni pizza (/14" pizza)	0.51	1.59	1.48	0.91	-0.95	-0.5	1.35	1.31
Chicken pie (/550g)	0.43	1.37	1.02	0.72	-0.86	-0.76	0.55	1.18
Gammon steaks (/225g)	0.52	1.44	1.06	1.59	-0.64	-1.31	0.8	0.75
Turkey mince (/400g)	0.32	1.05	1.21	1.12	-0.14	-1.01	0.69	1.03

Table A8: Mean willingness-to-pay (£) for a range of attributes in meat products, UK
(N=2,951 – approx. N=490 per choice experiment)

**Choice* refers to improved product quality from the base product; *premium* refers to the top level of product quality.

Source: Balcombe et al., 2016.

Kallas et al. (2015) designed a study using a simulated market setting to assess the impact of a possible ban on surgical castration of pigs in the EU. This study also included a sensory parameter by including a scent and taste test between two CEs. As Table A9 shows, participants were willing to pay a small amount for the welfare attribute while the sensory impact resulted in some differences in WTP estimates, such as the WTP for flavour attribute changing from a negative to a positive WTP of 0.66 euros/package (55% premium) after exposure to product tasting. The results also show that participants' WTP was lower for the manufacturer's own brand compared to the private brand.
		Pre Sensory CE		Post sensory CE	
		WTP	Premium	WTP	Premium
		€/package	(%)*	€/package	(%)*
Flavour (vs. Original/ non- flavoured)	With spices and naturally smoked	-0.558	(-47%)	0.660	(55%)
Castration (vs. none)	Meat from castrated pigs or boars	0.340	(29%)	-	-
Brand (vs. manufacturer)	Private	-0.252	(-21%)	-0.342	(-29%)

Table A9: Willingness-to-pay for pork sausage attributes, Spain (N= 150*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant. *Compared to the average of the applied price vector: €1.19/package Source: Kallas et al. (2015)

Animal welfare was also included in the Zanoli et al. (2013) investigation of consumers' beef product preferences in Italy. In particular, the study contrasted animal welfare with production methods, origin and quality indicators (e.g. fat content and colour). Table A10 shows that organic and domestic attributes had the highest relative WTP of between 24 and 26 euros/kg (109% and 206% of base price) respectively.

Table A10: Willingness-to-pay for beef attributes, Italy (N = 145*)

		WTP €/kg	Premium (%)**
Production method (vs. not organic)	Organic	26.25	(109%)
Production method (vs. not conventional)	Conventional	12.76	(106%)
Animal welfare (vs. Box)	Free-range	17.29	(144%)
Place of production (vs. abroad)	Italy	24.69	(206%)
Breed origin (vs. not local)	Local	6.40	(53%)

* Data were gathered from three different locations (medium-sized towns) in northern, central and southern Italy, in 2008.

** Compared to the basic prices reported in study: €24/kg for the organic beef attribute, and €12/kg for other attributes

Source: Zanoli et al. (2013)

Van Loo et al. (2014) combined different environmental and ethical attributes in a CE of chicken products, segmenting participants into income brackets. The attributes were presented in different logos, labels and claims associated with production, with CE results showing a consumer preference for product labels or claims over not having them at all. As Table A11 shows, average WTP is higher for free-range claims (43-93%), with respondents also favouring the introduction of domestic or EU-organic logos, carbon footprint and animal welfare labels.

		WTP	Premium	WTP	Premium
Attributes		euros/kg	(%)**	euros/kg	(%)**
		Low ii	псоте	High In	соте
Organic logo	Biogarantie logo (Belgium)	2.16	(23%)	3.18	(34%)
(vs. none)	EU Organic logo	1.16	(12%)	1.70	(18%)
Animal welfare label (vs. none)	European animal welfare label	2.50	(26%)	3.67	(39%)
Free range	Free range	4.12	(43%)	6.06	(64%)
claims (vs.	Traditional free range	4.77	(50%)	7.02	(74%)
none)	Free range-total freedom	5.99	(63%)	8.81	(93%)
Carbon	20% CO2-reduction: 5.6 kg CO2e compared to 7 kg CO2	1.73	(18%)	2.54	(27%)
(vs. none)	30% CO2-reduction: 4.9 kg CO2e compared to 7 kg CO2	2.31	(24%)	3.40	(36%)

Table A11: Willingness-to-pay for chicken breast attributes, Belgium (N = 359*)

* Online survey conducted in the northern Belgium, 2012.

** Compared to the average price for conventional chicken breast in Belgium in 2012 (€9.49/kg) Source: Van Loo et al. (2014)

Viegas et al. (2014) estimated Portuguese consumers' WTP for animal welfare in the context of testing whether premiums paid for credence attributes can justify higher associated production costs. Specifically, the authors hypothesised that WTP for a particular attribute (e.g. animal welfare) is conditional on the presence of other attributes (e.g. environmental quality and/or food safety). The reference alternative included legal minimums and a status quo price. As shown in Table A12 below, the estimated WTP suggests that the highest value was placed on food safety, ranging from 7-16 euros/kg, followed by animal welfare and environmental protection. An important implication was that the WTP for different combinations of attributes should not be obtained from independent valuation and summation due to significant interaction effects. The authors then applied a conditional approach on estimating attribute WTP (Table A12, last column) whereby, for example, the WTP for food safety in the presence of both animal welfare and environmental certification decreases the average WTP (from up to 16 euros to negative or close to zero). This suggests that animal welfare and environmental attributes may be proxies for food safety.

Attribute	Levels	Average WTP €/kg (premium %*)		Conditional WTP** €/kg (premium %*)		
	_	main effects	main + interaction effects			
Beef safety (vs. legal standards)	Certified additional level: Reduction/control of the quantity of antibiotic residues in beef	7.31 (42%)	16.23 (93%)	AW =0 ENV = 0 AW = 1 ENV = 0 AW = 0 ENV = 1 AW = 1 ENV = 1	16.23 7.47 7.32 -1.43	(93%) (43%) (42%) (-8%)
Animal welfare (vs. legal standards)	Certified additional level	7.30 (42%)	12.07 (69%)	FS = 0 FS = 1	12.08 3.32	(69%) (19%)
Environmental Protection (vs. legal standards)	Certified additional level: Air, water, soil pollution and reduction/ prevention	4.81 (28%)	7.35 (42%)	FS = 0 FS = 1	7.35 -1.55	(42%) (-9%)

Table A12: Willingness-to-pay for beef attributes, Portugal (N = 613)

*Compared to average of the applied price vector (€17.98/kg)

** 1 indicates the condition, zero otherwise: AW = Animal Welfare; ENV = Environmental Protection; FS = Food Safety

Source: Viegas et al. (2014)

Gracia (2014) investigated Spanish consumers' WTP for local lamb products using a simulated market environment with an additional objective of reducing the risk of hypothetical bias in the results. The results shown in Table A13 indicate that consumers are willing to pay a premium of between 9 and 13 per cent for local and "Ternasco" lamb, respectively, over unlabelled or "suckling" lamb, respectively.

Table A13: Willingness-to-pav	for fresh local	lamb attributes.	Spain (N = 133)

Attribute		WTP €/package	(Premium %)
Locally grown label (vs. unlabelled)	Labelled as "Ojinegra from Teruel"	0.29	(9%)
Type of commercial lamb (vs. <i>"Suckling" lamb</i>)	"Ternasco" lamb	0.43	(13%)

Source: Gracia, 2014.

Van Wezemael et al. (2014) conducted a European cross-country study exploring consumer preferences and WTP for nutrition and health claims in relation to beef steak. The study tested an information/framing effect in a split-sample approach wherein one sample was shown attributes with nutritional claims only (N sample) and other sample were shown both nutritional and health claims together (NH sample). The results from Table A14 suggest that the valuation of nutritional and health claims varies across countries. Across samples, the NH sample had consistently higher WTP, with the exception of a "rich in protein" claim in the UK. This indicated the existence of country-specific marketing opportunities when considering nutrition and health claims on beef products, such as information regarding product protein levels in the UK.

	N sample		WTP €/kg	Premium (%)**
1		Netherlands	5.44	(33%)
Iron	Nutritional claim: "Source of iron"	Belgium	4.26	(26%)
(VS. NO		France	4.11	(25%)
Clainti)		UK	5.04	(31%)
Dratain		Netherlands	2.71	(16%)
Protein	Nutritional Claims "Dish in protain"	Belgium	3.42	(21%)
(VS. 110 claim)	Nutritional Claim: Rich in protein	France	4.96	(30%)
Clainty		UK	5.81	(35%)
Coturated		Netherlands	5.78	(35%)
Saturated	Nutritional Claim: "noor in saturated fat"	Belgium	5.60	(34%)
Idt (VS. 110	Nutritional Claim: poor in saturated fat	France	6.73	(41%)
Clainty		UK	1.20	(7%)
NH sample				
Iron (va no	Nutritional claims "Course of iron"	Netherlands	5.62	(34%)
rion (vs. no	Health Claim: "Iron contributes to the normal	Belgium	5.89	(36%)
Clairit)		France	5.49	(33%)
	cognitive function	UK	4.27	(26%)
Dratain	Nutritional Claims "Dish in protain"	Netherlands	4.22	(26%)
Protein (vic no	Health Claim: "Protoin contributor to the growth	Belgium	6.20	(38%)
(VS. 110 claim)	or maintenance of muscle mass "	France	9.70	(59%)
Clainty	of maintenance of muscle mass.	UK	4.39	(27%)
	Nutritional Claim: "poor in saturated fat"	Netherlands	8.45	(51%)
Saturated	Health Claim: "Consumption of saturated fat	Belgium	11.66	(71%)
fat (vs. no	increases blood cholesterol concentration.	France	11.71	(71%)
claim)	Consumption of foods with reduced amounts of saturated fat may help to maintain normal blood cholesterol concentrations."	UK	4.60	(28%)

Table A14: Willingness-to-pay for beef steak attributes, Belgium, France, The Netherlands and UK (N = 600/country*)

* Online survey in 2011 with people consuming beef at least once a month.

**Compared to average of the applied price vector (€16.5/kg)

Source: Van Wezemael et al. (2014)

In Sweden, Lagerkvist et al. (2014) focused on COO and ethical cues in the presence or absence of price attribute, the differences of which should not (in theory) impact on the preferences and structural validity of CE. A large of range attributes with quality and ethical cues were included in the study (see Table A15) where the absence of labelling information was used as a reference point. In addition, a non-parametric test was used to confirm attribute ranking by consumers. A sample of over 1,000 participants completed the survey. The WTP results in Table A15 are only reported for that part of the sample who saw the CE with the price vector (required for WTP calculation). These results show that consumers were willing to pay an average 10% premium for a verified SR labelling in beef products – approximately four times lower than COO information. COO was also found to be the top ranked attribute in both samples. In regards to the comparison between the inclusion and exclusion of price attributes, one of the results indicated that there was consistently less heterogeneity in the CE without the price attribute.

		"Price sample"		"Price sample"	"No-price sample"
		WTP SEK/kg	Premium (%)**	Attribute ranking	
Origin Information (vs. zone of origin inside or outside EU)	COO (inside or outside EU)	113.7	43%	1	1
Animal specific Reference code (vs. not present)	Information on package	15.0	6%	12	12
Traceability to specific slaughterhouse (vs. not present)	Information on package	32.0	12%	6	6
Traceability to group or specific animal (vs. not present)	Information on package	29.5	11%	7	9
Traceability to specific breeder (vs. not present)	Information on package	32.6	12%	5	4
Verified animal welfare for livestock production (vs. not present)	Information on package	42.1	16%	1	1
Organic production (vs. not present)	Information on package	37.0	14%	4	5
Verified Environmental impact of livestock production (vs. not present)	Information on package	25.6	10%	9	8
Verified health impact from consumption of beef production (vs. not present)	Information on package	21.5	8%	10	10
Verified social responsibility for livestock production (vs. not present)	Information on package	27.4	10%	8	7
Information about medication use (vs. not present)	Information on package	41.2	16%	3	3
Type of animal feed (vs. not present)	Information on package	18.4	7%	11	11

Table A15: Willingness to pay for beef attributes, Sweden (N = 1,070*; n = 630 "no-price sample" and n = 440 "price sample")

* Online survey in 2012 amongst beef consumers.

**compared to the average of the applied price vector: 262.5 SEK per kg Source: Lagerkvist et al. (2014)

Paci et al. (2018) examined Italian consumers' WTP for the inclusion of environmental and health attributes in fresh fish burger products, finding a WTP of up to an additional 0.57 Euro for the "environment" attribute and 0.37 Euro for the "health" attribute.

North American studies

The current review includes 8 CE and other WTP studies examining the attributes of meat and seafood products in the US. Attributes examined in these studies include animal health and/or welfare, organic, different production methods, traceability, country-of-origin, food safety, environmental condition and certification, as well as generic attributes including product quality and appearance.

Li et al. (2016) examined US consumers' household WTP for a programme aimed at reducing Greenhouse Gas (GHG) emissions associated with beef production. The authors created four consumer segments based on their willingness to support a programme certifying "carbon-

friendly" beef products – 'does not support', 'supports but will not pay more', 'supports and will pay more', and 'willing to pay specific premium for certified beef'. For the latter two segments combined, results indicated that participants in these segments would be willing to pay an average US\$306 per year to support this programme (equating to 51.6 per cent of their average annual total beef product spend). Across all segments, including those that would not support this programme, average annual WTP was valued at US\$64 (just over 10 per cent of all participants' average annual total beef product spend). Average WTP was also shown to be higher for participants that donated to environmental organisations (Li et al., 2016).

Merritt et al. (2018) undertook a choice experiment to examine US consumers' WTP for a range of beef product attributes, including quality assurance, region of origin and various production practices, as well as a combination of these attributes. Specifically, these attributes were *Tennessee Certified Beef, Certified Angus Beef, grass-fed, Master Quality Raised Beef* and *no hormones administered*. In addition, WTP estimates were carried out for two types of beef products – USDA Choice boneless ribeye beef steak, and USB Choice ground beef (85% lean/15% fat). Furthermore, participants undertaking a choice experiment for either product were evenly distributed into either a control treatment (who were shown no additional information about the attributes of each product) and an information treatment (who were shown additional information about the attributes of each product) and an information treatment (who were shown additional information about the attributes of each product) and an information treatment (who were shown additional information about the attributes of each product). Estimates of WTP for each attribute within and between each of the above groups is shown in Table A17 and A18 below. Both tables show a generally higher WTP for all attributes by those in the Information Treatment segment, with the highest overall WTP for both product types across both segments to be for a combination of *Tennessee Certified Beef (TCB)* and *grass-fed* attributes (Merritt et al., 2018).

Attribute	Control Treatment (n = 204)	Information Treatment (n = 204)	WTP Treatment Difference
Tennessee Certified Beef (TCB)	2.42	2.89	0.47
Certified Angus Beef (CAB)	1.19	1.43	0.24
Grass-fed	0.95	1.43	-0.48
Master Quality Raised Beef (MQRB)	1.39	1.67	0.28
No hormones administered	2.35	2.71	0.37
TCB and CAB	2.51	3.36	0.85
TCB and grass-fed	3.93	3.56	-0.37
TCB and MQRB	2.62	3.67	1.05
TCB and No hormones administered	4.37	3.28	-1.10

Table A17. Willingness-to-pay for USDA Choice boneless ribeye beef steak product attributes (USD (\$) per pound (Ib)) (2018) (N = 408 total)

Source: Merritt et al., 2018.

Table A18. Willingness-to-pay for USDA Choice ground beef (85% lean/15% fat) product attributes (USD (\$) per pound (Ib)) (2018) (N = 408 total)

Attribute	Control Treatment	Information Treatment	WTP Treatment
	(n = 204)	(n = 204)	Difference
Tennessee Certified Beef (TCB)	1.15	1.53	0.38
Certified Angus Beef (CAB)	0.41	0.73	0.33
Grass-fed	0.81	0.59	-0.22
Master Quality Raised Beef (MQRB)	0.65	0.91	0.26
No hormones administered	1.27	1.59	0.33
TCB and CAB	1.29	1.61	0.31
TCB and grass-fed	1.76	1.98	0.21
TCB and MQRB	1.45	1.72	0.27
TCB and No hormones administered	1.63	2.41	0.78

Source: Merritt et al., 2018.

Byrd et al. (2017) examined US consumers' WTP for a range of attributes associated with chicken and pork products, including local production, animal welfare and food safety. These attributes were also assessed against a range of certifying bodies, including the USDA, retailers and industry bodies. Table A19 shows a range of premiums that participants were willing to pay in relation to the above, with results indicating the highest positive WTP for pasture access for chicken, particularly when certified by the USDA.

Attaihuta	Verifier	Chicker	n breast	Pork chop	
Allribule	vermer	WTP	% positive WTP	WTP	% positive WTP
	USDA	1.78	91.7		
Pasture access	Retailer	1.47	92.7		
	Industry	1.43	82.3		
Individual crate	USDA			1.98	84.0
	Retailer			0.27	45.5
	Industry			2.34	72.6
	USDA	1.87	75.0	4.55	85.7
Antibiotic use	Retailer	1.33	74.3	1.32	61.7
	Industry	1.11	61.7	1.17	70.0
Local	USDA	2.06	89.6	1.44	9.4
	Retailer	0.49	68.9	1.31	9.9
	Industry	0.49	59.7	3.37	3.9

Table A19: Willingness-to-pay for chicken and pork products with associated local, animal welfare and food safety attributes, US (N=825) (US\$/Ib)

Source: Byrd et al., 2017.

In another pork CE, Ubilava et al. (2011) compared US consumers' WTP for the *certification* of credence attributes for branded and non-branded products. Selected credence attributes included antibiotic use, animal welfare and environmental friendliness in the production process where, in a split-sample, some CEs also included a product brand (*Hormel, Tyson, Store brand* or *no brand*). Table A20 reports the WTP results which range from 4 to 28 per cent (0.2 to 1 \$/lb) for certified antibiotic-free, environmentally-friendly and animal welfare attributes. The study also reported a greater variation in WTP for the non-branded case, which could be related to an increased uncertainty when no brand information is provided; while it also appears that the attributes as *bundles* (i.e. attribute interactions) influenced consumer preferences.

		Choices with brands		Choices without brands	
	By brand	WTP \$/lb	Premium (%)**	WTP \$/lb	Premium (%)**
3 rd party certified antibiotic-free production	Hormel	0.78	22%		
	Tyson	0.35	10%	0.62	1.00/
	Store Brand	0.61	18%	0.05	10%
(vs. no certification)	No brand	0.98	28%		
3 rd party certified	Hormel	0.76	22%		
environment-friendly	Tyson	0.26	7%		
production: water and air	Store Brand	0.15	4%	0.24	7%
quality (vs. no certification)	No brand	0.32	9%		
3 rd party certified animal	Hormel	0.58	17%	0.42	12%
welfare in the production	Tyson	0.41	12%		
process (vs. no	Store Brand	0.18	5%	0.42	
certification)	No brand	0.67	19%		
	Tyson	0.45	13%	0.07	440/
ANTI*ENV	Store Brand	0.25	7%	0.37	11%
	Hormel	0.37	11%		
ANTI*WEL	Tyson	0.40	12%	0.31	9%
	Store Brand	0.29	8%		
	Tyson	0.35	10%		
ENV*WEL	Store brand	0.54	16%	0.48	14%
	No brand	0.37	11%		

Table A20: Willingness-to-pay for pork chop attributes, USA (N = 839*: brand CEs n = 642, non-brand CEs n = 197)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

ANTI = antibiotic-free production; ENV = environment-friendly production; WEL = animal welfare

* A mail survey in 2004 with a sample of 9,600 randomly selected households.

** Compared to the average of the applied price vector: US\$ 3.475/lb

Source: Ubilava et al. (2011)

In the United States, Lim et al. (2014) focused on the valuation of COO information alongside trade-offs such as quality (e.g. tenderness), production practices (use of hormones and antibiotics), food safety (identified by testing and/or traceability), and price of beef. A nationwide survey was conducted with a sample size of 1000. WTP was only estimated for the COO attribute, either independently or taking into account the respondent specific attitudes toward food safety¹. The results in Table A21 show that, on average, consumers preferred domestic beef, with negative WTP shown for imported products indicating a compensation of around \$5-\$7/lb to achieve these levels. A further analysis show that, ceteris paribus, COO preferences were related to the perceived food-safety level of the country. For example, consumers who had a high risk perception or distrust about the safety of Australian products were willing to pay less for imported beef from Australia, or that people who were risk-averse in regards to food safety had an overall lower WTP for imported products.

¹ General food safety attitudes and perceptions were explored in a Likert scale question.

Table A21: Willingness-to-pay for beef attributes, USA (N = 1,000*)

Attribute	Levels	WTP US\$/lb	Premium (%)**
Country of Origin (vs. USA)	Canada	-5.75	(-53%)
	Australia	-7.33	(-68%)

* A nationwide online survey in 2010.

** Compared to average (USD 10.75) from a vector of low-to-high-end actual market prices Source: Lim et al. (2014)

Van Loo et al. (2011) assessed US consumers' WTP for different organic label types on chicken products. Their analysis focused not just on average WTP but also WTP by different consumer segments based on the purchase-frequency of organic meat (*'non-buyers', 'occasional buyers'*, and *'habitual buyers'*) and on demographics (gender, age, education, household income and number of children). Table A22 shows positive premiums for both types of organic labelling, with higher premiums associated with the USDA organic label (\$3.6/lb or 104% premium) over the generic label (\$1.2/lb or 35%). Further analysis showed that WTP differs between demographic groups as well as between different organic buyers. Most respondents (59%) were occasional buyers; around one fourth of the respondents had never bought organic chicken; and only a small group of respondents (15%) bought organic chicken always or often. As expected, the premiums that consumers were willing to pay for organic chicken increased by the frequency of purchase. Consumer WTP estimated for each demographic group showed, for example, that females had a higher WTP than males, and that having more children reduced WTP, while higher income increased WTP for products with organic labels.

Table A22: Willingness-to-pay for chicken meat attributes, USA (N = 256 non-buyer, N = 571 occasional buyers, N = 149 habitual buyers)

		WTP full sample \$/lb	Premium (%)**	By the type of buyer	WTP \$/lb	Premium (%)**
Label (vs. no label) Gen orga	LISDA organic		(104%)	Non-buyer	0.90	(26%)
	label	3.55		Occasional	3.33	(97%)
				Habitual	8.37	(244%)
	Generic organic label	1.19	(35%)	Non-buyer	-1.01	(-30%)
				Occasional	1.22	(36%)
				Habitual	5.02	(147%)

*Online survey amongst the members of a consumer database in Arkansas.

** Compared to the average price for boneless chicken breast (\$3.424/lb)

Source: Van Loo et al. (2011)

Compared to meat products, consumer preferences towards the credence attributes of seafood products is relatively unexplored. In United States, Ortega et al. (2014) explored consumer WTP for imported seafood products for which past food contamination and adulteration incidents may have impacted on consumer preferences for Chinese tilapia. Two surveys were conducted (for shrimp and Chinese tilapia products) with 335 respondents each. The corresponding CEs included a variety of credence attributes: COO (US, China and Thailand) information was considered only for shrimps and the verification entity (US government, Chinese Government, US Third Party) was considered only for Chinese tilapia. The estimation process included attribute interactions between the credence attributes and COO for shrimps, and between credence attributes and verification entity for Chinese tilapia. The results in Table A23 show that consumers were willing to pay more for enhanced food safety: \$10.65/lb for domestic shrimp, \$3.71/lb shrimp from China, and \$4.12/lb shrimp from Thailand. The respective premiums were 118 per cent, 41 per cent and 46 per cent. A similar relationship

was found for no-antibiotic use and environmentally friendly production, which were both associated with a higher WTP for the US product by US consumers.

WTP assessments for Chinese Tilapia, as presented in Table A24, show that consumers were, on average, willing to pay between \$4 and \$6 per pound (or 89-120 per cent of the base price) for enhanced food safety when verified by a US entity. Likewise for no-antibiotic use and environmental friendly production claims, the only statistically significant evidence was associated with US verification bodies. Overall, the government verification system was valued slightly higher relative to third-party verification. These results are consistent with the shrimp CE results wherein US consumers had a higher WTP for domestic over overseas seafood products and verification systems.

Table A23: Willingness-to-pay for seafood	l (shrimps) attributes,	USA (N = 335*)
---	-------------------------	----------------

			WTP \$/lb	Premium (%)**
Food safety (vs. no claim)		US product	10.65	(118%)
	Enhanced	Chinese product	3.71	(41%)
		Thai product	4.12	(46%)
Antibiotic use (vs.	Not permitted	US product	9.83	(109%)
permitted)	Not permitted	Thai product	2.84	(32%)
Production practice (vs. conventional)	Eco-friendly	US product	5.40	(60%)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* An online survey in 2011.

** Compared to average of the applied price vector (US\$9/lb)

Source: Ortega et al. (2014)

Table A24: Willingness-to-pay for seafood (imported tilapia) attributes, USA (N = 335*)

			WTP \$/nound	Premium
			<i>γγρουπα</i>	(10)
Food safety (vs.	Enhanced	US government verified	6.02	(120%)
no ciaim)		US third party verified	4.43	(89%)
Antibiotic use (vs.	Not permitted US government verified		5.39	(108%)
permitted)		US third party verified		(55%)
Production practice (vs. conventional)	Eco-friendly	US government verified	2.67	(53%)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* An online survey administered by a market research company in 2011.

** Compared to the lowest given price option (\$5.00/pound) in the price vector

Source: Ortega et al. (2014)

Asian studies

The current review includes 7 CE and other WTP studies examining the attributes of meat and seafood products in Asia, including the markets of China, Japan, Korea and India. Attributes examined in these studies include animal health and/or welfare, organic, different production methods, traceability, country-of-origin, food safety, environmental condition, certification, water use and GM production, as well as the generic attributes of product quality and appearance.

In Asia, Wu et al. (2015) explored consumer preferences and WTP for a traceability and certification information for pork meat. The sample consisted of consumers in seven Chinese cities that had been designated by the China Ministry of Commerce as pilot cities for a meat and vegetable traceability system. Each respondent was classified by their level of income and education, which was used in the WTP analysis. As shown in Table A25, estimated WTP across the full sample ranged from 2.31 Yuan/kg to 15.80 Yuan/kg (or 19% to 32% premiums) for the different product attributes. The provision of product traceability information had the highest WTP (ranging from 42% to 91% premiums of base price) for the full traceability over no information. Only those consumers with low income/education level were willing to pay for the minimum level of traceability information. Likewise, regarding quality certification, most consumers were willing to pay more (ranging from 104% to 149% premiums of base price) for government certification over no certification. The high profile consumers were the only group that valued third-party certification (over no certification), which is consistent with findings that higher education and income are related to the WTP for traceability certification (Zhang et al. 2012). It was also found that product freshness had a significant impact on respondents' meat choice preferences.

A separate consumer class-based analysis generated four distinct consumer classes based on the respondents' choices, thus further supporting the preference heterogeneity in the sample. These were labelled as 'certification-preferred', 'price-sensitive', 'appearance-preferred' and 'scared' consumers, whereby the first class included over half of the respondents. Overall, the findings presented in Table A26 complement those presented above, including that WTP for quality certification appears slightly higher than for others, apart from the 'appearance preference' class; and that there are obvious class-specific preferences. The 'scared' class was different to the others in that they preferred the possibility to opt-out in the given alternatives. Furthermore, for this class, no WTP values are reported here (as the price attribute was not statistically significant).

Table A25: Willingness-to-pay for pork attributes, China (N = 1,489)

		WTP full	WTP by age and income/education level				
A ++++ih++++ =		sample		yuan	/500g (prem	nium %**)	
Attribute		yuan/500g		High	Madium	Low	High income
		(premium %**)		підп	wealum	LOW	Low education
			Age = 35	10.95	7.94	6.70	9.44
				(91%)	(66%)	(56%)	(79%)
	E	8.32	Age = 45	9.78	6.76	5.53	8.26
	Full	(69%)		(82%)	(56%)	(46%)	(69%)
			Age = 60	8.01	5.00	-	6.49
				(67%)	(42%)	-	(54)%
Traceability			Age = 35	8.13	5.72	5.00	7.96
Information				(68%)	(48%)	(42%)	(66%)
*** (vs.	Dartial	5.72	Age = 45	7.96	5.55	4.83	7.78
none)	Partial	(48%)		(66%)	(46%)	(40%)	(65%)
			Age = 60	7.71	5.29	4.57	7.43
				(64%)	(44%)	(38%)	(62%)
			Age = 45	-	-	2.29	-
	Minimauma	2.31				(19%)	-
	wiinimum	(19%)	Age = 60	-	-	2.84	-
						(24%)	
	Government	13.83 (115%)	Age = 35	11.35	14.01	15.16	12.84
				(95%)	(117%)	(126%)	(107%)
			Age = 45	12.42	15.09	16.23	13.92
				(104%)	(126%)	(135%)	(116%)
			Age = 60	14.04	16.70	17.85	15.53
				(117%)	(139%)	(149%)	(129%)
			Age = 35	11.22	10.12	10.33	13.17
Quality				(94%)	(84%)	(86%)	(110%)
Certification	Domestic	15.80	Age = 45	10.19	9.09	9.30	12.15
(vs. no	third-party	(132%)		(85%)	(76%)	(78%)	(101%)
certification)			Age = 60	8.64	7.54	7.75	10.60
				(72%)	(63%)	(65%)	(88%)
			Age = 35	12.03	-	-	-
				(100%)			
	International		Age = 45	10.86	-	-	-
	third-party	-		(91%)			
			Age = 60	9.11	-	-	-
				(76%)			
	Very fresh-	13.74					
Appearance	looking	(115%)					
(vs. Bad-	Fresh-	11.34					
looking but	looking	(95%)					
edible)	Passable-						
	looking	-					

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* In-store intercept interviews, in 2013, in seven cities across different regions of China.

**Compared to the average price of pork hindquarters (12 yuan/500g) as reported in the study

*** Full traceability information covering farming, slaughter and processing, circulation and marketing; Partial traceability information covering farming, slaughter and processing; Minimum traceability information covering only farming.

Source: Wu et al. (2015)

		Clas	ss 1*	Clas	s 2*	Clas	is 3*	Class 4*
		certifi	cation-	nrica s	price-sensitive		irance-	scared
Attribute		pref	erred	price-se			erred	consumers
	Class probability	52	.7%	12.	6%	20	.8%	13.9%
				WTP Yua	n/500g (pr	emium %	**)	
Traceability	Full	5.24	(44%)	-		3.40	(28%)	-
Information	Partial	2.68	(22%)	0.50	(4%)	2.37	(20%)	-
*** (vs. none)	Minimum	-1.30	(-11%)	-		-		-
Quality	Government	8.82	(74%)	0.78	(7%)	3.05	(25%)	-
Quality Certification (vs. no certification)	Domestic third- party	6.28	(52%)	-		2.71	(23%)	-
	International third- party	4.06	(34%)	0.54	(5%)	3.64	(30%)	-
Appearance	Very fresh-looking	5.16	(42%)	0.69	(6%)	10.95	(91%)	-
(vs. Bad-	Fresh-looking	4.76	(40%)	-		9.49	(79%)	-
looking but edible)	Passable-looking	-4.18	(-35%)	-		-6.21	(-52%)	

Table A26: Willingness-to-pay for pork attributes, China (N = 1,489)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* In-store intercept interviews, in 2013, in seven cities across different regions of China.

**Compared to the average price of pork hindquarters (12 yuan/500g) as reported in the study

*** Full traceability information covering farming, slaughter and processing, circulation and marketing; Partial traceability information covering farming, slaughter and processing; Minimum traceability information covering only farming.

Source: Wu et al. (2015)

Wu et al. (2016) examined Chinese consumers' WTP for the provision of traceability information in relation to pork products using real choice experiments (RCE) and experimental auctions (EA). In particular, the authors examined WTP for different types of traceability information, including farming, slaughter and processing, distribution and marketing, and government certification information against a base of a pork product without traceability information. Consistent with previous studies, Table A27 shows that mean WTP was positive but varied between the two methods used (RCE and EA) and the types of information provided, with consumers showing higher WTP across both experiments for government certification and farming information (Wu et al., 2016).

Table A27: Willingness-to-pay for traceability information in relation to pork, China (N=108)

Information Type	Mean WTP (Yuan/500g) (95% confidence interval)			
	RCE	EA		
Farming information	4.375	2.405		
Slaughter and processing information	1.565	1.215		
Distribution and marketing information	1.071	0.735		
Government certification information	4.934	2.785		

Source: Wu et al., 2016.

Lai et al. (2018) used a series of choice experiments to determine Chinese consumers' (Beijing and Shanghai) WTP for a range of attributes of pork products, including environmental, food safety and animal welfare standards, as well as country of origin. Results showed a range of premiums associated with different attributes, as shown in Table A28 below. This shows

generally higher WTP for all attributes from Shanghai participants, with *food safety, Chinese origin* and *environmental standards* having the highest associated WTP values (Lai et al., 2018).

Table A28. Willingness-to-pay for pork product certification attributes by Chinese consumers (Beijing and Shanghai) (2018) (N = 480 total)

Attribute	Mean WTP (RMB) – Beijing (N = 259)	Mean WTP (RMB) – Shanghai (N = 221)
Food Safety	32.01	32.32
Animal Welfare	7.65	13.11
Environmental Standards	11.81	20.73
Country of Origin: United States	4.31	9.61
Country of Origin: China	13.26	30.11

Source: Lai et al., 2018.

Wang et al. (2018) used a discrete choice experiment to determine urban Chinese consumers' WTP for pork products with certified attributes. Specifically, this included certified labels for organic production, green food production, food safety, location of origin, and free from veterinary drug residues. Choice experiments were carried out in two Chinese provinces (Jiangsu and Anhui) with results reported for each – these are shown in Table A29 below. This shows a greater WTP for all attributes by Jiangsu consumers, with generally higher WTP for *organic food*, followed by *green food* and *free from veterinary drug residues* across both provinces.

Table A29. Willingness-to-pay for pork certification attributes, Jiangsu and Anhui provinces,China (2018) (Yuan/550g)

Attribute	Jiangsu (N = 475)	Anhui (N = 369)
Safe Food	8.10	7.21
Green Food	20.22	17.63
Organic Food	26.78	18.94
Location of Origin shown	12.77	10.99
Free from veterinary drug residues	23.18	15.40

Source: Wang et al., 2018.

Ortega et al. (2015) explored consumer preferences and WTP for chicken, pork and egg product attributes across various retail channels in China. Retail channel types included wet markets, domestic supermarkets, and international supermarkets, wherein the products may vary in terms of food safety and other attributes such as animal welfare, organic, "green" foods and price. Three hundred consumers were interviewed for each food product (pork, chicken and eggs) with an equal number of participants from each retail channel. Results presented in Table A30 show that while consumer WTP for food safety was mostly similar across the different retail channels, with premiums from 165 per cent to 267 per cent compared to the base price, these varied across product types. "Green food" certification was valued higher (up to 20 RMB/product or 195% premium) than organic certification across all products and retailers. Some differences across retail types can be observed for the WTP for the animal welfare attribute as this was significant only for pork and chicken products and not for wet markets.

		F	Pork	Chi	icken	E	ggs
		WTP RMB/prod		duct Prer		nium (%)**	
Enhanced food	Wet market	27.73	(213%)	19.94	(199%)	9.93	(199%)
safety claim (vs.	Domestic supermarket	23.68	(182%)	26.69	(267%)	9.58	(192%)
no claim)	International supermarket	25.50	(196%)	21.45	(215%)	8.23	(165%)
Animal welfare	Wet market	-	-	-	-	-	-
claim (vs. no	Domestic supermarket	7.36	(57%)	-	-	-	-
claim)	International supermarket	-	-	-	-	2.28	(46%)
Organic	Wet market	-	-	-	-	3.28	(66%)
certification (vs.	Domestic supermarket	11.48	(88%)	15.44	(154%)	5.37	(107%)
no claim)	International supermarket	12.11	(93%)	-	-	3.89	(78%)
Green food	Wet market	-		-		5.07	(191%)
claim (vs. no	Domestic supermarket	11.79	(91%)	19.69	(197%)	6.76	(135%)
claim)	International supermarket	19.29	(148%)	16.27	(163%)	6.63	(133%)

Table A30: Willingness-to-pay for chicken, pork and eggs attributes, China (N= 300/product*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* In-store (at the point of purchase) interviews in Beijing, 2013.

**Compared to average of the applied price vector (pork: RMB 13/jin, chicken: 10 RMB 10/jin and eggs:, and RMB 5/jin

Source: Ortega et al. (2015)

Chung et al. (2012) focused on heterogeneity in WTP for beef attributes. Countries-of-origin of interest included Korea (i.e. domestic), USA and other exporting countries (e.g. New Zealand). They conducted 1,000 interviews amongst Korean consumers, with heterogeneity of preferences and WTP explored using a consumer segment-based approach. As Table A31 shows, the analysis resulted in three consumer segments based on the respondent's choices regarding concerns in relation to GM-beef and the use of antibiotics in production. These segments were labelled as 'very concerned' (59% of the sample), 'moderately concerned' (32%) and the smallest group of 'not too concerned' (9%). Thus, over half of the sample were very concerned about the use of GM and antibiotics with WTP around \$4.4/lb (20 per cent premium), and about product's origin with WTP around -\$8/lb (37 per cent premium) for imported meat. This 'very concerned' segment held generally higher WTP values than other segments, and generally these were higher than the weighted averages. Overall, these results suggest that there exists major heterogeneity in Korean (Seoul) consumer preferences towards meat choices, in particular, regarding the use of GM ingredients and antibiotics in production.

		Very	Moderately	Not too	
		Concerned	Concerned	Concerned	
Class probability		59%	32%	9%	
					Weighted
			WTP \$/lb		Average WTP
			Premium (%)**		US\$/lb
					Premium (%)**
	Extra promium	3.01	1.58	0.88	2.35
Marbling Grade	Extra premium	(13%)	(7%)	(4%)	(7%)
(vs. C)	Promium	2.13	1.05	0.93	1.67
	Freihlum	(9%)	(5%)	(4%)	(7%)
Marbling Grade	^	2.04	0.91	0.62	1.55
(vs. not A)	A	(9%)	(4%)	(3%)	(7%)
Marbling Grade		0.92	0.39	-	0.66
(vs. not B)	В	(4%)	(2%)		(3%)
	High	2.94	1.69	1.14	2.37
Freshness (vs.		(13%)	(8%)	(5%)	(11%)
low)	Madium	1.09	0.76	0.56	0.93
	Wedium	(5%)	(3%)	(2%)	(4%)
Chilled versus	No - freshly	0.63	0.53	0.24	0.56
frozen (vs. yes)	chilled	(3%)	(2%)	(1%)	(2%)
Free of		4.39	1.06	0.81	3.00
antibiotics (vs.	Yes	(20%)	(E9/)	(40/)	(120/)
no)		(20%)	(3%)	(4%)	(15%)
Free of GM-		4.35	0.95	0.59	2.92
feed ingredients	Yes	(19%)	(1%)	(3%)	(13%)
(vs. no)		(1578)	(470)	(370)	(1370)
Country-of-	United States	-8.38	-3.74	-2.85	-6.39
origin (vs	United States	(-37%)	(-17%)	(-13%)	(-28%)
Korea)	Other exporting	-7.25	-3.47	-2.19	-5.57
Koreaj	countries	(-32%)	(-15%)	(-10%)	(-25%)

Table A31: Willingness-to-pay for beef attributes, Korea (N = 1,000*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* In-store intercept interviews in Seoul, 2007.

**Compared to the average of the applied price vector: US\$ 22.50/lb

Source: Chung et al. (2012)

Uchida et al. (2014) examined Japanese consumer preferences for salmon, taking into account two-way interactions motivated by consumer valuations of different product attributes in relation to ecolabel characteristics. The study included a split-sample CE across three types of information effects regarding fisheries (specifically overfishing and the decline of fish-stock): (1) minimal information without the source of the claim; (2) Food and Agriculture Organization (FAO) based information with charts and graphics; and (3) scientific information accompanied by a diagram. Hence, instead of using a conventional approach of "no information" vs. "some information", the authors applied minimum information as the baseline. Likert-scales were used to understand general attitudes, information credibility, and the respondents' level of interest. A nationwide survey included in total 3,370 responses. As shown in Table A32, Japanese consumers were willing to pay a 27 per cent premium (90 yen/package) for the domestic fish compared to imported fish, with a similar premium found for the ecolabel. Considering these attributes together, the WTP was 149 yen/package which is slightly less than sum of the independent WTP values (90 + 89 = 179). Overall, the interaction effects revealed that the value of eco-labels increased value for the wild product, in particular for the domestic product. The findings from the information effect testing revealed that compared to baseline, added information increased the value of the eco-label, although marginally, when the FAO or science based information were considered credible and interesting.

		<i>Premium (%)**</i>
	Hokkaido (domestic)	(26%)
Product origin (vs. Chile)	Alaska	(8%)
	Norway	(7%)
Production (vs. farmed)	Wild	(10%)
Ecolabel (vs. no label)	Labeled	(26%)
	Ecolabel x Hokkaido	(44%)
	Ecolabel x Alaska	(27%)
Country of origin	Ecolabel x Norway	(28%)
x	Ecolabel x Wild	(37%)
Wild***	Hokkaido x Wild	(52%)
	Alaska x Wild	(36%)
	Norway x Wild	(37%)
	Ecolabel x FAO	22%
	Ecolabel x Science	20%
	Ecolabel x FAO x Credible	30%
Information treatments x	Ecolabel x Science x Credible	28%
Perceptions***	Ecolabel x FAO x Interesting	29%
	Ecolabel x science x Interesting	27%
	Ecolabel x FAO x Interesting	36%
	Ecolabel x Science x Interesting	34%

Table A32: Willingness-to-pay for salmon attributes, Japan (N = 3,370*: "minimal information" n = 1,122, "FAO information", n = 1,118, and "Science information" n = 1,130)

* A nationwide online survey in 2009.

**Reported in the study

***Base levels: Country of origin and wild: "Chilean farmed salmon with no ecolabel"; and Treatments and perceptions: "Minimal information perceived neither credible nor interesting" Source: Uchida et al. (2014)

Other regions

The current review includes CE and other WTP studies examining the attributes of meat and seafood products in other regions, including Australia and Lebanon. Attributes examined in these studies include animal welfare, local foods, production quality and certification.

Mugera et al. (2017) examined Australian consumers' WTP for chicken and yogurt products based on their preferences for a range of attributes, including local production, free range, product quality and the size of the producer. This was based on whether a product carried a local food label, was certified free range, or contained other information relating to the attributes listed. The authors examined WTP for a combination of the above attributes, as shown in Table A33. This also shows a range of additional premiums for each of the product types and attributes based on a range of demographic variables, including gender and type of area.

Table A33: Willingness-to-pay for chicken and yoghurt products based on local production, free range, size of producer (relative to medium) and demographic variables, Australia (N=333)

Attribute 1				WTP for pı (\$A	oduct type UD)
	Attribute 2	variable 1	variable 2	Skinless chicken breast	Fruit yoghurt
Local	Australian firm				5.15
	Overseas firm				3.67
		City		6.16	
		Country		8.32	
Not local	Australian firm				3.84
	Overseas firm				2.36
		City		3.74	
		Country		5.91	
Free range		City	Female	5.86	
			Male	3.77	
		Country	Female	4.27	
			Male	2.17	
Small producer				1.55	2.64
Large producer				-1.84	-2.8

Source: Mugera et al., 2017.

Chalak and Abiad (2012) studied Lebanese consumers' preferences and purchasing behaviour in context of shawarma sandwiches², a Lebanese fast food, which is considered to contain a high potential for food safety risk. The study attributes included food safety certification (International Organization for Standardization [ISO] and "ServSafe" food handling program), and contextual factors such as location, serving size and price. The sample included 284 respondents, wherein the information-effect was tested in a split-sampling approach by providing half of the sample with additional descriptions of each type of safety certification. WTP results, as summarised in Table A34, suggest that, overall, consumers appreciated the convenience in buying sandwich from "around the corner", and that they also preferred to pay extra 46 per cent for larger sandwich size (around US\$1.12 (LBP 1,677)). The information effect was apparent in this study, as this increased the average WTP for food safety certification from a 282 to 314 per cent premium to a 320-431 per cent premium compared with the average price of a small sandwich. WTP for certification was highest for the ISO 22000 type.

^{2 &}quot;Shawarma is a Middle Eastern beef, lamb or chicken-based fast food" (Chalak and Abiad 2012 p. 82).

Table A34: Willingness-to-pay for sandwich attributes, Lebanon (N = 284*: informed n =
145, uninformed n = 139)

	Levels		WTP LBP/sandwich	Premium (%)**
Location/	Within walking distance		445	(12%)
Convenience (vs.	(5+ min walk)		-445	(-1270)
Round the corner <	Need to go there by car		-4,181	(-115%)
5 min walk)	Delivery order		-1,009	(-28%)
Certification (vs. none)	150,0001	Uninformed	10,278	(282%)
	150 9001	Informed	11,667	(320%)
	150 22000	Uninformed	11,466	(314%)
	150 22000	Informed	15,719	(431%)
	Com Cofe	Uninformed	1 0,372	(284%)
	Servsare	Informed	14,366	(394%)
Portion size (vs. Typical small-sized sandwich)	Medium-sized sandwich		1,677	(46%)

LBP = Lebanese pounds; US\$1 = LBP1,515

* The survey was conducted in Beirut, 2011, excluding participants who had never purchased shawarma sandwiches.

** Compared to an average of LBP3,650 (USD2.41) for a small-sized shawarma sandwich

Source: Chalak and Abiad (2012)

Cross-regional studies

Tait et al. (2016) conducted a cross-country analysis between developed and developing economies (UK vs. China and India). The authors explored preferences across certified environmental attributes (GHG, biodiversity, and water quality), animal welfare, food safety, country-of-origin (COO) label and price in relation to lamb products. A generic framing on the product, including a percentage price increase, was used to make the cross-country comparison more straightforward. Results reported in Table A35 show that food safety, followed by animal welfare, appeared to be the most valued attributes with WTP values of between 9% and 49% more for a certified product. Another similarity across the countries was that of different environmental attributes, the GHG certification was valued most, although not by much. Key differences included that while UK consumers preferred domestic products, consumers in developing markets were not likely to choose the domestic product or pay for it. Another difference was that the Indian respondents had higher WTP for environmental attributes compared with UK and Chinese consumers. Overall, this study shows there can be cross-country differences when looking into food attribute preferences but also that similarities might exist, for example, in terms of which attributes are valued the highest.

Table A35: Willingness-to-pay for lamb attributes, China, India, UK (N = 2,067*: China *n* = 686, India *n* = 695 and UK *n* = 686)

		WTP (in %)**		
		China	India	UK
Food safety (vs. not certified)	Certified	34%	49%	15%
Farm animal welfare (vs. not certified)	Certified	9%	29%	18%
Water management (vs. not certified)	Certified	7%	21%	6%
Greenhouse Gas (GHG) minimisation (vs. not certified)	Certified	8%	28%	6%
Biodiversity enhancement (vs. not certified)	Certified	5%	26%	4%
Country of origin (vs. no label)	Domestic	-27%	-	5%
	Foreign	-	13%	-5%

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Online survey in in 2012 with regular grocery shoppers who had purchased lamb at least once recently (last month).

** Reported in the study Source: Tait et al. (2016)

Source. Tait et al. (2010)

A1.2 Dairy products

The current review includes 6 CE and other WTP studies examining the attributes of dairy products in Europe, North America and Asia. Attributes examined in these studies include country-of-origin, environmental condition, carbon/GHG emissions associated with production, local foods, organic, functional foods, product health claims, brand and food safety.

European studies

The current review includes 3 CE and other WTP studies examining the attributes of dairy products in Europe, including studies conducted in Germany, France, Italy, Norway, Spain and the UK. Attributes examined in these studies include country-of-origin, environmental condition, carbon/GHG emissions associated with production, local foods, organic, functional foods and product health claims.

Aichner et al. (2017) examined German consumers' WTP for ice cream and tea products based on their associated country-of-origin. The researchers selected an ice cream product from the USA with a Scandinavian name (Häagen-Dasz) as well as a German tea product with an English name (Milford) in order to gauge German consumers' WTP for the product(s) before and after their country-of-origin was revealed. Table A36 shows reductions in WTP for both product types following the reveal of the products' respective country-of-origin, including minimum, maximum and mean WTP ranges (Aichner et al., 2017).

	Häagen-Dasz (ice cream)			Milner (tea)		
	Minimum (€)	Maximum (€)	Mean (€)	Minimum (€)	Maximum (€)	Mean (€)
Actual product price	4.99	5.99	5.05	1.85	2.39	1.89
WTP before COO was revealed	4.99	10.00	5.35	1.85	3.00	1.98
WTP after COO was revealed	2.00	6.50	4.48	0.90	2.50	1.74

Table A36: Willingness-to-pay for ice cream and tea products before and after COO information provided, Germany (N=100)

Source: Aichner et al., 2017.

Feucht and Zander (2017) examined European consumers' (France, Germany, Italy, Norway, Spain and the UK) WTP for "climate-friendly" milk products (i.e. products with a lower carbon footprint), including products that displayed two types of CO₂ label, as well as product claims relating to "climate-friendliness", local production and organic production (EU organic label). Table A37 shows participants WTP for the inclusion of each of the above in relation to milk products, showing the highest indicated WTP for local production and organic production.

Table A37: Willingness-to-pay for milk products, environmental attributes, European countries (Euro per 1-litre UHT milk product)

	France (N=1,000)	Germany (N=1,001)	Italy (N=1,003)	Norway (N=1,001)	Spain (N=1,002)	UK (N=1,000)
CO ₂ Label 1	0.11	0.13	0.24	0.14	0.14	0.10
CO ₂ Label 2	0.03	0.03	0.09	0.00	0.11	0.06
"Climate friendly"	0.06	0.05	0.14	0.09	0.15	0.04
Local	0.19	0.20	0.27	0.27	0.15	0.15
Organic	0.12	0.10	0.23	0.14	0.16	0.09

Source: Feucht and Zander, 2017.

In Germany, Bechtold and Abdulai (2014) estimated consumer WTP for functional dairy products (yoghurt and cream cheese) by linking the choice data with demographics and general attitudes information. The choice alternatives were described as bundles of functional ingredients, health claims and product prices. The data included 1,309 responses where each respondent answering a CE for both yoghurt and cheese products. The data was analysed using the consumer segment based approach with the class determinants including the socioeconomic and attitudinal variables, the latter generated from principal component analysis (PCA). The results in Tables A38 and A39 show evidence for the class-specific preference heterogeneity when taking into account respondent attitude and respondent-type associated determinants in relation to the reference group. For example, it was confirmed that "functional food skeptics" preferred non-functional dairy products, and vice versa by the "functional food advocates". Furthermore, the majority of consumers valued dairy products with functional ingredients, such as omega-3, highly. These WTP varied from €0.13 to

€0.31/serving of yoghurt and €0.35/serving of cream cheese, or premiums of between 10 and 23 per cent.

		Class 1*** Functional food sceptics	Class 2*** Functional food advocates	Class 3*** Functional food neutrals (reference group)
Class probability		(21.5%) (40.5%)		(38%)
			WTP €/200g Premium (%)**	
	Omega-3 fatty acids	0.31 (24%)	0.24 (19%)	0.13 (10%)
Functional Food ingredient	Oligosaccharides	-	0.10 (8%)	0.11 (9%)
	Bioactive	-	-0.10 (-8%)	-0.11 (-9%)
	Polyphenols			
Non-functional alternative		0.47 (36%)	-1.77 (-137%)	-
	Healthy blood vessels.	-	-0.41 (-32%)	-0.13 (-10%)
Health claim	Healthy blood vessels and metabolism	-	0.23 (18%)	-0.08 (-6%)
	One property depending on the ingredient	-	-0.18 (-14%)	0.11 (9%)
	Two properties depending on the ingredient	-	-	-

Table A38: Willingness-to-pay for yoghurt attributes, Germany (N = 1,309*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Nationwide mail survey, 2010-2011.

**Compared to the base price for conventional non-functional food as provided in the study: €1.29/500g

***Class determinants: **Class 1** Reward from using Functional Foods (FF), Safety of FF, General health interest, Natural product interest, Hysteria; **Class 2** Age, Education, Reward from using FF, General health interest, Natural product interest, Hysteria, Necessity for FF, Specific health interest

Source: Bechtold and Abdulai (2014)

		Class 1*** Functional food sceptics	Class 2*** Functional food advocates	Class 3*** Functional food neutrals (reference group)
Class probability		(24.8%)	(33.9%)	(41.3%)
			WTP €/200g Premium (%)**	
	Omega-3 fatty acids	0.35 (23%)	0.35 (23%)	-
Functional Food ingredient	Oligosaccharides	-	0.05 (3%)	-
	Bioactive	-	-0.18 (-12%)	-
	Polyphenols			
Non-functional alternative		0.97 (65%)	-1.86 (-125%)	-0.02 (-1%)
Health claim	Healthy blood vessels.	-	-0.38 (-26%)	-
	Healthy blood vessels and metabolism	-	0.24 (16%)	-
	One property depending on the ingredient	-	-0.24 (-16%)	-
	Two properties depending on the ingredient			

Table A39: Willingness-to-pay for cream cheese attributes, Germany (N = 1,309*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Nationwide mail survey, 2010-2011.

Compared to the base price for conventional non-functional food as provided in the study: €1.49/200g *Class determinants: **Class 1** Children aged < 12, General health interest, Natural product interest, Hysteria, Necessity for Functional Food (FF), Confidence in FF, Safety of FF; **Class 2** Gender, Children < 12years, Reward from using FF, General health interest, Natural product interest, Hysteria, Necessity for FF, Specific health interest, Confidence in FF

Source: Bechtold and Abdulai (2014)

North American studies

Zou and Hobbs (2010) explored consumers' functional food choices and a labelling effect in a context of Omega-3 enriched milk in Canada. The different health claims included heart health, generic health claims and more specific risk reduction claims (RRC) and disease prevention claims (DPC). The authors separated these claims from the visual cues (a red heart symbol included in a choice set) and labelled them as full and partial functional food attributes, respectively. The CE also considered certification and product price. The data analysis used two approaches, the standard model (Table A40) and the segmented-based approach (Table A41). These initial results suggest that consumers respond positively to health claim labels, as well as the verification entities for these claims. Consumers were willing to pay, on average, between \$0.12 and \$0.51 for different health claims (or 6% to 26% more of the conventional milk price), being highest for the RRC. They were also willing to pay, on average, around 12 per cent more for verification (vs. none) with little difference on WTP across the type of verification entity. The study also found some sociodemographic influences, such as income, increased WTP for the Omega-3 attribute.

The second analysis confirmed these preferences were consumer group-specific (Table A41). Overall, the full health claims seemed to have a higher absolute WTP (over no claim) when

compared to the WTP value of the visual claim (over none), apart from the *"health claim challengers"* group, who were minority of the sample (7%). Looking specifically at the functional ingredient attribute, people were willing to pay, on average, \$0.20/litre premium for Omega-3 enriched milk over regular milk, and this WTP was even higher for people with higher income and those with positive attitudes toward functional food in general.

		WTP \$/2 Litres	Premium (%)**
Omega-3 (vs. regular milk)	Contains Omega-3	0.20	(10%)
	Function Claim: "Good for your heart health"	0.19	(10%)
Health Claims (full labelling) (vs. none)	RRC: "Reduces the risk of heart disease and cancer"	0.51	(26%)
	DPC: "Helps to prevent Coronary Heart Disease and Cancer"	0.33	(17%)
Symbol (partial labelling) (vs. none)	Heart Symbol	0.12	(6%)
Verification	Government	0.24	12%
Organization (vs. none)	Third party	0.23	12%

Table A40: Willingness-to-pay	for milk attributes,	Canada (N = 740	*)
-------------------------------	----------------------	-----------------	----

* Online survey conducted in 2009.

** Compared to the lowest price in the given price vector: \$1.99/2 litres of conventional milk.

Source: Zou and Hobbs (2010)

		WTP \$/2 Litres Premium (%)**				
		Conventional milk consumers	Functional food believers	Functional milk lovers	Health claim challengers	
Class probabilities		48.9%	21.7%	22.1%	7.3%	
	Contains Omega-3	-	0.25 (13%)	1.64 (82%)	0.29 (15%)	
	Omega3 x Factor1	0.11 (6%)	4.84 (243%)	0.48 (24%)	0.74 (37%)	
Omega-3 (vs. regular milk)	Omega3 x Factor2	-	-0.25 (-13%)	-	-0.23 (-12%)	
	Omega3 x Income	1.39 (70%)	3.85 (193%)	8.94 (449%)	-4.37 (-220%)	
	Omega3 x Gender	0.12 (6%)	3.09 (155%)	0.96 (48%)	0.96 (48%)	
	Function Claim	-	0.16 (8%)	0.49 (25%)	-	
	RRC	-	0.37 (19%)	1.83 (92%)	-	
	RRC x Factor1	-	-0.14 (-7%)	0.36 (18%)	0.26 (13%)	
(full labelling)	RRC x Factor3	-	-	0.36 (18%)	-	
(vs. none)	RRC x Heart disease	-	-	-0.58 (-29%)	-	
	RRC x Education	-	-	-0.29 (-15%)	-	
	DPC	-	0.46 (23%)	1.74 (87%)	-	
Symbol (partial labelling) (vs. none)	Heart Symbol	-	-	0.31	0.27	
	Government	-	0.17 (9%)	0.98 (49%)	0.37 (19%)	
Verification Organization	Government x Factor3	-	0.09	0.25	0.33	
(vs. none)	Third party	-	0.33	0.70	-	

Table A41: Willingness-to-pay for milk attributes: The latent class approach, Germany (N = 740*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Online survey in 2009.

** Compared to the lowest price in the given price vector: \$1.99/2 litres of conventional milk.

*** Heart disease: "respondent self-reports having heart disease"; Factor 1 "positive attitudes toward and experience consuming functional food"; Factor 2 "more awareness of health and healthy diet behaviours"; Factor 3 "higher levels of trust in health claims and nutrition labels" (Zou and Hobbs 2010 p. 10 Table 2). Source: Zou and Hobbs (2010)

(17%)

(35%)

Grashuis and Magnier (2018) used two choice experiments to assess US consumers' WTP for a range of attributes associated with cheese and cereal products. Specifically, this included type of company ownership (cooperative, firm), product origin (local, Wisconsin/lowa), and family ownership status. The researchers assessed consumers' WTP using three models for each choice experiment, rendering different sets of results for each (i.e. Model 1 includes the main attributes, Model 2 includes more detailed analysis of ownership type in combination with origin, and Model 3 includes more detailed analysis of ownership type in combination with family-owned status). Results are shown in Table A42 and A43 below. In general, WTP for both products was shown to be higher for firm-owned production, with generic *local* production favoured over specified locations (Grashuis and Magnier, 2018).

Attribute	Mean WTP (USD/8oz) (Model 1)	Mean WTP (USD/8oz) (Model 2)	Mean WTP (USD/8oz) (Model 3)
Ownership: Cooperative	0.766	1.374	0.655
Ownership: Firm	1.453	1.908	1.365
Origin: Local	0.728	1.156	0.735
Origin: Local (Cooperative)		-0.672	
Origin: Local (Firm)		-0.489	
Origin: Wisconsin	0.406	1.186	0.410
Origin: Wisconsin (Cooperative)		-1.289	
Origin: Wisconsin (Firm)		-0.916	
Family-Owned	0.501	0.504	0.385
Family-Owned (Cooperative)			0.211
Family-Owned (Firm)			0.170

Table A42. Willingness-to-pay for cheese attributes, US (N = 298)

Source: Grashuis and Magnier, 2018.

Table A43. Willingness-to	-pay for cereal attribu	tes, US (N = 394)
---------------------------	-------------------------	-------------------

Attribute	Mean WTP (USD/12oz) (Model 1)	Mean WTP (USD/12oz) (Model 2)	Mean WTP (USD/12oz) (Model 3)
Ownership: Cooperative	1.001	1.014	1.400
Ownership: Firm	1.153	1.099	1.257
Origin: Local	0.411	0.335	0.404
Origin: Local (Cooperative)		0.253	
Origin: Local (Firm)		-0.139	
Origin: Iowa	0.067	-0.255	0.063
Origin: Iowa (Cooperative)		0.092	
Origin: Iowa (Firm)		0.648	
Family-Owned	0.513	0.580	0.885
Family-Owned (Cooperative)			-0.894
Family-Owned (Firm)			-0.327

Source: Grashuis and Magnier, 2018.

Asian studies

In China, Wu et al. (2014) assessed consumers' WTP for organic infant formula, as well as respondents' food safety risk perceptions and level of knowledge. The CE attributes included organic label, COO brand (including two Chinese ("unknown" *Dele*, and well-known *Yili*) and two foreign brands (European *Topfer*, and North American *Enfamil*)) and product price. The design also included two-way interaction effects between the attributes in order to explain variance in preferences. The study was conducted in Shandong province (China's third most populous province), resulting in 1,254 completed responses. The result show, firstly, that the respondents' knowledge and understanding of organic food were relatively low while the perception regarding the food safety risk were relatively high. The CE results in Table A44 show that consumers had a higher average WTP of \$5-\$10 (or 36-69 per cent of the base price) for the EU and US-based organic labels than for the Chinese label (vs no label). These WTP

estimates increased if the level of knowledge and the level of perceived food safety risk were higher, up to 112 per cent and 86 per cent, respectively. Furthermore, Chinese consumers preferred imported products and brands over domestic ones which is consistent with previous studies (Saunders et al. 2013). Lastly, the study highlighted two of the significant and positive findings from the attribute interactions (between the US organic label and China-COO, and between *Enfamil* and China-COO), which imply a potential complementary relationship whereby adding these labels/brands to formula produced in China could improve their value.

Full sample					By le knov	evel of vledge	By lev perc	el of risk eption
		WTP US\$/40 0g	Premium (%)**		WTP U	S\$/400g	Premiu	ım (%)**
				Low	3.49	(23%)	3.84	(26%)
	Chinese	3.23	(22%)	Medium	3.84	(26%)	4.28	(29%)
				High	1.95	(13%)	4.20	(28%)
Organic				Low	3.81	(25%)	3.75	(25%)
label (vs. no	EU	5.36	(36%)	Medium	6.93	(46%)	6.02	(40%)
label)				High	6.04	(40%)	6.25	(42%)
				Low	10.66	(71%)	9.93	(66%)
	US	10.40	(69%)	Medium	16.87	(112%)	12.58	(84%)
				High	16.55	(110%)	12.89	(86%)
Durana di (ura	Yili	4.40	(29%)					
Brand (VS.	Topfer	6.17	(41%)					
Dele)	Enfamil	7.08	(47%)					
Country of	China	-2.42	(-16%)					
origin (vs. Germany)	the US	3.53	(24%)					

* In-store interviews, in 2012.

** Compared to the average of the applied price vector: US\$ 15/400g

Source: Wu et al. (2014)

A1.3 Fruit & vegetable products

The current review includes 8 CE and other WTP studies examining the attributes of fruit and vegetable products in Europe, Asia and other regions. Attributes examined in these studies include organic, local foods, country-of-origin, social responsibility, carbon/GHG emissions associated with production, food safety, production methods and product quality.

European studies

The current review includes 3 CE and other WTP studies examining the attributes of fruit and vegetable products in Europe, including the markets of Denmark, France, UK and the Netherlands. Attributes examined in these studies include organic, local foods, country-of-origin, social responsibility and carbon/GHG emissions associated with production.

Denver and Jensen (2014) focused on the organic and local food (apples) preferences in Denmark. The study combined CE and PCA, where the latter was used to aggregate attitudinal Likert-scale responses. The CE included attributes of food origin ranging from domestic (local or domestic) to imported apples (within or outside of the EU); production method (organic vs. conventional); alongside colour and taste/texture. The survey included in total 637 respondents. The PCA show two components - one related to organic products and the other

to locally produced products. While no WTP was calculated, the authors provided an indication of WTP for these two attributes (Table A45). The participants were willing to pay 5.40 DKK/kg premium for organic apples and 19 DKK/kg for local food. These numbers increased by 97 percentage points if the respondents hold "maximum perception" of the organic attributes based on the PCA. This suggests that, in the case of apples, consumers with positive perceptions of organic food can also have relatively strong preferences for local food but not necessarily vice versa. The authors suggest that this asymmetry needs to be explored further.

	Full sample Those with maximum perception of the organic attributes State			timum perception nic attributes	
		WTP DKK/kg	Premium (%)**	WTP DKK/kg	Premium (%)**
Production method (vs. conventional)	Organic	5.40	77%	12.20	174%
Origin (vs. outside EU)	Local	19.00	(271%)	22.60	(323%)

* Online survey in 2010.

**Compared to current price (status quo option) of a conventional apple 7 DKK/kg Source: Denver and Jensen (2014)

In another European study, Akaichi et al. (2015) assessed consumers WTP for fair-trade (FT), organic and carbon footprint attributes (collectively known as ethical attributes) in bananas. A particular objective was to identify if these attributes compete in different markets. For the study, in total 247 consumers were interviewed in three countries. The CE results (Table A46) show that consumers were willing to pay between €0.08 and €0.14 for fair trade and organic bananas with French participants indicating a slightly higher, and statistically significant, WTP compared to Scottish and Dutch participants. All respondents were also willing to pay, on average, €0.10 (77% premium of the lowest price) to reduce carbon footprint (1kg on the transport). These WTP values were statistically significantly higher by Dutch over Scottish participants. In order to explore these trade-offs, a within-sample test of WTP differences was applied. These results show that, in Scotland, consumers were willing to pay significantly more for fair trade bananas compared to other attributes, but also that they would choose organic bananas if the FT price too high. In the Netherlands sample, there was no evidence for different WTP for attributes; thus these attributes are competing and the price of attribute determines choices. Lastly, French participants were willing to pay significantly more for organic bananas than fair trade bananas, if the price is not too high. Overall, consumers in all countries show positive WTP for all claims/labels, and although generally these ethical claims may not be competing, this study identified that under some circumstances this may change.

Table A46: Willingness-to-pay for the banana attributes, Scotland, France and the Netherlands (N = 247*: 100 in Edinburgh, 95 in Clermont-Ferrand and 52 in Amsterdam)

	WTP by all respondents		WTP by Country		
	€/banana	Premium (%)**		€/banana	Premium (%)**
Fairtrade	0.10		Scotland	0.14	108%
Label (vs. no	0.10	77%	Netherland	0.13	100%
label)			France	0.09	69%
Organic Label	0.00	60%	Scotland	0.08	62%
(vs. no label)	0.09	69%	Netherland	0.09	69%

			France	0.13	100%
Carbon			Scotland	0.09	69%
footprint/	0.10	770/	Netherland	0.12	92%
reduction per kg	0.10	7776	France	0.12	92%

* Intercept survey at public places and retail stores with occasional buyers, at minimum, of bananas ** Compared to the lowest amount of the price vector: €0.13/banana

Source: Akaichi et al. (2015)

Ceschi et al. (2018) used a choice experiment to analyse Italian consumers' WTP for apple attributes, specifically their variety, production method(s) and region(s) of production. As shown in Table A47 below, the authors found a range of premiums associated with specific regions of production, with consumers willing to pay a higher premium for apples produced in Trentino-Adige (+ \in 1.44 per kg) and Emilia-Romagna (+ \in 1.41 per kg) over imported apples (- \in 2.12 per kg). Similarly, the organic attribute was shown to have only marginal increased WTP relative to conventional apples (+ \in 0.18 per kg) (Ceschi et al., 2018).

Table A47. Willingness-to-pay for apple attributes, Italy (N = 301)

Attribute	WTP (€/kg)
Organic	+0.18
Bicolour	-0.34
Green	-1.00
Red	-0.94
Trentino-Alto Adige	+1.49
Emilia-Romagna	+1.44
Imported	-2.12

Source: Ceschi et al., 2018

North American studies

There have been some, but limited, studies of consumer WTP for attributes of fruit and vegetable products in North American countries.

Grebitus et al. (2018) used a series of online choice experiments to determine US consumers' WTP for Medjool dates with associated GMO, pesticide use and region of origin credentials. In general, this showed that participants were willing to pay positive premiums for all attributes, particularly those with GMO- and pesticide-free status, as well as a preference for dates grown in the state of Arizona over California. Table A48 below shows the range of premiums associated with the above attributes.

Attribute	Mean WTP (US\$/ounce)
Arizona grown	+0.14
California grown	+0.03
Pesticide-free	+0.55
GMO-free	+0.17
GMO- and pesticide-free	+0.53

Table A48. Willingness-to-pay for date attributes, US (N = 1,411)

Source: Grebitus et al., 2018.

Asian studies

In a developing economy context, Wongprawmas and Canavari (2017) examined Thai consumers' WTP for fresh produce with associated food safety credentials, including a product's freshness, brand and food safety information. For product freshness, a range between 0 and 2 days post-harvest was indicated. Food safety labels used in the CE included

a generic "safe produce" claim, the well-recognised Q Mark label, as well as well-known and trusted produce brands "Royal Project" and "Doctor's Vegetables", both of which may also use the Q Mark label. Table A49 shows a range of WTP for different brand and food safety information credentials in relation to Chinese cabbages among Thai consumers, with trusted private brands Royal Project and Doctor's Vegetables receiving the highest WTP.

Table A49: Willingness-to-pay for Chinese cabbage with food safety credentials, Thailand (N=350)

Attribute	WTP (Thai Baht/kg)
Claim "safe produce"	39.23
Q mark	68.44
Royal Project and Q mark	74.56
Doctor's Vegetables and Q mark	79.06

Source: Wongprawmas and Canavari, 2017.

While not strictly a fruit and vegetable product, Gao et al. (2019) used a series of choice experiments to examine urban Chinese consumers' WTP for country of origin and genetically modified organism status of different orange juice products (*orange juice drink* (OJD), *orange juice from concentrate* (FCOJ) and *orange juice not from concentrate* (NFC)). As shown in Table A50 below, WTP estimates were produced against alternatives (e.g. a series of origins versus Chinese origin), producing a range of premiums associated with different orange juice product attributes. In particular, the results show a range of discounts associated with country of origin and GM status, with price premiums associated only with changes in product types.

Table A50. Willingness-to-pay for orange juice products by type, production method and country of origin, China (N = 646)

Category	Attribute	Mean WTP (RMB)
Draduct Type (vs 10% Orange	50% Orange Juice Drink	5.38
luico Drink)	Orange Juice From Concentrate	9.81
Julee Driftk)	Orange Juice Not From Concentrate	13.27
	US	-4.61
Conventional Juice (Imported	Brazil	-1.55
vs China)	Israel	-2.13
	Australia	-3.70
	US	-4.87
	Brazil	-13.60
iuico from the same country)	Israel	-14.52
Juice from the same country)	Australia	-4.59
	China	-12.12
	US brand, made in United States	-3.05
Brand and Manufacturer	US brand, made in China	-4.66
Country of Origin (other vs	US brand, made in Florida	-5.47
Chinese brand, made in China)	Taiwanese brand (China), made in China	-1.45
	Australian brand, made in Australia	-0.05

Source: Gao et al., 2019.

Other regions

The current review includes 2 CE and other WTP studies examining the attributes of fruit and vegetable products in other regions, including Peru and West African nations (Benin, Ghana

and Burkina Faso). Attributes examined in these studies include organic, local foods, food safety and production methods.

Blare et al. (2017) conducted a CE to determine Peruvian consumers' WTP for locally grown tree fruits (avocadoes, apples and pears). Table A51 shows the percentage of participants willing to pay a range of premiums (0%, 10%, 20%, 30%, 40% and 50% more) for locally-produced apples, avocadoes and pears, with highest overall premiums shown for local apples, followed by pears and avocadoes.

Table A51: Percentage of participants willing-to-pay for locally-grown tree fruits, Per	L
(N=300)	

	WTP range					
	0%	10%	20%	30%	40%	50%
Apples (%)	26	17	24	16	6	11
Avocadoes (%)	24	29	30	12	1	4
Pears (%)	25	21	26	16	8	4

Source: Blare et al., 2017

Probst et al. (2012) explored the potential for marketing certified organic vegetables in three West African cities (Cotonou in Benin, Accra in Ghana and Ouagadougou in Burkina Faso). In particular, certified organic production was examined as a potential strategy to improve food safety. Two separate CEs were developed - one for the food vendors' choices of tomatoes (a common ingredient in meals) and another for consumer meal choices of (continental or traditional) when eating out. The vendor CE included trade-offs across appearance (freshness, colour and neatness), production method and price attributes, while the consumer CE included trade-offs across taste, production method and price attributes. Both CEs targeted different types of retailers ranging from street food vendors to restaurants, where the interviews resulted in 180 vendor responses and 360 consumer responses. There were some differences in sample demographics between vendors and consumers, such as consumer sample being predominantly female whereas the vendors were mostly male. In both CEs, the WTP was only reported for the organic production attribute. As shown in Table A52, the vendors were willing to pay, at median, US\$0.85 for organic certification of the fresh tomatoes, which equals to a premium between 12 and 53 per cent of typical retail price. These WTP across the cities vary depending on the season. Next, Table A53 shows they consumers were willing to pay, at median, just over US\$1 per meal if the food served contained only certified organic vegetables. This equates to around a 19 per cent premium on average meal price for restaurants, 75 per cent premium for small food businesses, and 177 per cent premium on average meal price for street food vendors.

Table A52: Willingness-to-pay for basket of tomatoes attributes (by vendors), Benin, Ghana and Burkina Faso (N = 180*, n = 60/city)

			By City	Lean season	Peak season
		WTP US\$/3 kg		(promium %)**	(premium
		basket		(premum %)	%)**
			Benin	(16%)	(39.9%)
How vegetables	Cortified organia	¢0.949	Burkina	(26.70/)	(52.40/)
were grown (vs. not	Certified organic	ŞU.848	Faso	(20.7%)	(53.4%)
organic)			Ghana	(12.1%)	(23.9%)

Note: The WTP values were not estimated for all attributes.

* Intercept interviews, in 2009, with street food vendors, small food businesses and restaurants.

** Reported in the study.

Source: Probst et al. (2012)

Table A53: Willingness-to-pay for meal attributes (by consumers), Benin, Ghana and Burkina Faso (N = 360*)

		WTP US\$/plate	By retailer	(% premium)**
How vegetables	Contified		Street food vendor	177%
added to the meal	Certified	\$1.044	Small food business	75%
were grown (vs. not organic)	vegetables		Restaurant	19%

* Intercept interviews, in 2009, with customers of the street food vendors, small food businesses and restaurants.

** Reported in the study. Source: Probst et al. (2012)

A1.4 Wine products

The current review includes 8 CE and other WTP studies examining the attributes of wine products in Europe, North America, Asia and other regions. Attributes examined in these studies include sustainability (generic), country- and region-of-origin, grape variety, vintage, brand, social responsibility, organic, carbon/GHG emissions associated with production, environmental condition, reduced packaging and taste.

General studies

Schaufele and Hamm (2017) conducted a review of international WTP literature regarding WTP for the inclusion of a range of sustainability credentials in wine products. The authors found that consumers across different countries showed a willingness to pay a premium for wine products with associated sustainable production methods, including environmental friendly, local and organic production methods (Schaufele and Hamm, 2017).

European studies

The current review includes 3 CE and other WTP studies examining the attributes of wine products in Europe, including the markets of Spain, France, Germany and the UK. Attributes examined in these studies include sustainability (generic), region-of-origin, grape variety, social responsibility, organic, carbon/GHG emissions associated with production and reduced packaging.

Sellers (2016) examined Spanish consumers' WTP for sustainable wine products based on their market segment and levels of knowledge of wine culture. As shown in Table A54, premiums that Spanish consumers are willing to pay may be based on their level of knowledge

of wine culture, with less participants with higher levels of knowledge of wine culture willing to pay a premium as well as a generally lower average percentage of premium price paid. In addition, Table A55 shows that Spanish consumers in different segments may be willing to pay higher premiums than others. For example, a higher percentage of urban-based consumers may be willing to pay a higher premium than consumers in the 'traditional segment'. This study shows that relative levels of expertise as well as socio-demographic segmentation may affect WTP for sustainability wine products in Spain.

Table A54: Willingness-to-pay (€) for sustainable wine by level of knowledge of wine culture
Spain (N = 553)

	(1) Beginner	(2)	(3)	(4)	(5) Expert	Global
% of consumers willing to pay a premium price	87.2	76.5	81.2	75	61.6	77.9
Average % of premium price	18.72	15.02	10.97	8.1	5.08	12.87

Source: Sellers, 2016

Table A55: Willingness-to-pay (€) for sustainable wine by market segment, Spain (N = 553)

	Traditional	Urban	Trendy	Routine	Occasional	Social	Global
% of consumers willing to pay a premium price	76.9	84.6	80.2	70.2	74.3	84.1	77.9
Average % of premium price	9.75	13.11	14.41	13.25	11.92	12.97	12.87

Source: Sellers, 2016

In a wine context, Kallas et al. (2013) focused on elements involved in wine choices for a special occasion, such as origin, people's experience and knowledge of wine ("wine references"), grape type and price. In the survey, the respondents were asked to complete two separate wine CEs. The first being a so-called "forced choice task" (with no opt-out option), and the second being "non-forced choice task" (with an added opt-out alternative). Four hundred wine consumers participated in the study. The results, shown in Table A56, indicate that the most preferred origins were non-imported wines, particularly the regional Catalonian wine with WTP around 2.60-3.10 \notin /bottle (or around 30% of the base price). Also experience and type of wine influenced consumers' wine choices, as indicated by the relatively higher WTP estimates. The main differences between forced and non-forced choices involved the significantly higher premium for regional wine and Cabernet Sauvignon wine when allowing opting-out. However, the forced choices resulted in higher WTP for national wines as well as lower discount or compensation (negative WTP) for prestigious wines and imported wines. Overall, the results from the non-forced CE suggest an increasing tendency of statistically significantly higher WTP for most preferred type and origin levels.

		Average WTP €/bottle (Premium %)**		
		"Forced choices"	"Non-forced choices"	
		2.65	3.07	
	Catalonia (regional)	(27%)	(31%)	
Origin	Crain (national) ***	0.50	0.39	
Origin	Spain (national)	(5%)	(4%)	
	Imported (international) ***	-3.15	-3.46	
	Imported (International)	(-32%)	(-35%)	
Wine references	proviously known (ovporionced	0.81	0.73	
	previously known/experienced	(8%)	(7%)	
	Decommended wine	-0.17	0.04	
	Recommended wine	(-2%)	(0.4%)	
	Prostigious wino***	-0.64	-0.78	
	Prestigious wine	(-6%)	(-8%)	
	Cabernet Sauvignon (French	1.77	2.29	
	variety) ***	(18%)	(23%)	
a	Cranacha (Cnanich variatu)	-1.18	-1.33	
Grape variety	Grenache (Spanish variety)	(-12%)	(-13%)	
	Marlet (French variety) ***	-0.60	-0.96	
	wenut (French variety)	(-6%)	(-10%)	

Table A56: Willingness-to-pay for wine attributes, Spain (N = 400*)

* Face-to-face interviews in supermarkets and streets (central city) of Barcelona.

** Compared to average of the applied price vector: 10 €/bottle

*** Statistically significant different between the forced and non-forced choices (p < 0.01 or p < 0.10) Source: Kallas et al. (2013)

Pomarici et al. (2018) used an experimental auction method to assess younger Italian consumers' (n = 200) WTP for a range of water-related attributes of wine products. Specifically, this included three different wine products – a conventional wine product (i.e. no water saving), a water saving front-of-pack labelled product, and a water saving back-of-pack labelled product. The authors showed that participants bid a median price of \notin 4.16 for the conventional wine product, and a median price of \notin 4.32 (\notin 0.16 premium) for the front-of-pack and back-of-pack labelled wine products respectively (Pomarici et al., 2018).

Asian studies

The current review includes 2 CE and other WTP studies examining the attributes of wine products in Asia (namely China). Attributes examined in these studies include country- and region-of-origin, vintage and brand.

Xu et al. (2014) used a mixed Logit model to examine Chinese consumers' WTP for countryof-origin, vintage and brand attributes in relation to red wine for personal consumption and gifting purposes. Table A57 shows that Chinese consumer WTP for red wine attributes differ depending on context (e.g. for personal consumption or gifting), with negative WTP shown for Chinese wines for gifting, as well as unanimously for non-branded wine products.

	Personal consumption	Gift purchase
USA to China	36.07	-63.3
USA to France	83.53	101.53
2- to 5-year old	57.42	36.81
2- to 10-year old	64.51	38.82
Branded to no brand	-91.32	-118.61

Table A57: Willingness-to-pay (Yuan) for red wine attributes for own consumption and gifting, China (N=540)

Source: Xu et al., 2014

Using the same dataset from the previous study, Xu and Zeng (2014) compared results using conditional logit and mixed logit models to examine Chinese consumers' WTP for red wine attributes. Table A58 shows differences in WTP estimates produced through the use of each method.

Table A58: Willingness-to-pay (Yuan) for red wine attributes for own consumption andgifting, China (N=540)

	Conditional logit	Mixed logit
California to China	-45.19	61.89
California to France	35.13	144.40
2- to 5-year old	35.77	39.36
2- to 10-year old	63.28	67.58
Branded to no brand	-115.36	-120.69

Source: Xu and Zeng, 2014

Other regions

The current review includes 2 CE and other WTP studies examining the attributes of wine products in other regions, including Australia and Russia. Attributes examined in these studies include country-of-origin and taste.

In another special occasion wine study by Mueller et al. (2010), the objective was to understand the importance of different wine label statements for regular wine consumers in Australia, not calculate WTP. The CE included a relatively large number of attributes, with ten different statements (history of the winery; local grape sources; production method; taste descriptor; elaborate taste descriptor; food pairing between wine and type of meal; consumption advice; environmental consciousness; website; and ingredients) either present or not on the label, plus price. Each alternative was represented with an undefined Australian wine with the same alcohol level to enhance the use of extrinsic cues in the choices. A sociodemographic comparison indicates that the sample for this study is mostly aligned with the general Australian wine consumer population based on a wine consumer survey from Roy Morgan in 2007 (as cited in Mueller et al. 2010). The data was analysed with a consumer class segmentation approach which resulted in five distinct classes that varied in terms of preferences for certain label information and price, but not in terms of respondents' characteristics. Overall, the most influential label attributes associated with the wine choices were price, history, taste descriptors and food pairing. In contrast, environmental information, ingredients and website information on the labels had a relatively smaller, or negative, impact on choices. An additional analysis revealed that just over half of the participants, generally, read the wine labels and found them interesting as well as helpful.
In a Russian case study, Cicia et al. (2013) explored consumer preferences and WTP for red wine. Their CE included seven wine types varying by their geographical origin and quality-dependent price. Based on the estimated WTP (Table A59), three distinct segments were found: (1) high-quality-high-price Italian and French wines with WTP varying between ξ 4.8-5.7/bottle, or 96-113 per cent of the base price; (2) a medium-quality wines (WTP of ξ 2.96/bottle, or 54%); and (3) lower quality wines with WTP less than one Euro per bottle. Moreover, the non-CE results showed that wine consumption was generally described as occasional and that certification of origin was considered as a proxy for quality, which was also reflected in respondents' WTP.

		WTP €/bottle	Premium (%)**
	Italy-Tuscany (Chianti)	5.66	(113%)
Geographical	France (Bordeaux)	4.81	(96%)
origin (vs. Chile	Spain (Rioja)	2.69	(54%)
Cabernet)	Italy-Sicily (Cabernet)	0.97	(19%)
	Russia (Krasnodar Grenache dry)	0.92	(18%)
	Georgia (Saperavi dry)	0.06	(1%)

Table A59: Willingness-to-pay for wine attributes, Russia (N = 388*)

* Sample included Russian households located in Moscow, Saint Petersburg and Novosibirsk.

**Compared to the lowest value of the applied price vector including Chilean wine, approximately €5/bottle. Source: Cicia et al. (2013)

Cross-regional studies

Lastly, Mueller Loose and Remaud (2013) explored North American and European consumer preferences for wine choices which involve corporate social responsibility claims (an umbrella term for ethical and social attributes) alongside product price. Prior to the CE, participants were also asked about their awareness and trust of different claims in food and wine products. The survey targeting wine consumers resulted in between 982 and 2,027 respondents in different countries. The results show, firstly, that overall awareness, purchase penetration and trust with regards to social and environment claims were similar across for each claim but different across the markets. For example, compared to European markets, North American consumers seemed to have a higher level of trust and claim awareness. As shown in Table A60, WTP results support differences across markets, but also across the different label claims. Over all markets, the average WTP was highest for organic claims at around €1.20/bottle (or 14% premium) - twice as much than the WTP for the environmental claims. Across the markets, not all attributes were statistically significant in all countries, such as for social and environmental responsibility. In most of these markets, the organic attribute had the highest WTP, particularly in France and Germany. Negative WTP can interpreted as a consumer demand for a discount, or consumer dislike, if such labels exist for wine products, such as socially responsibility in French markets or the reduced glass weight of wine bottles. Overall, this cross-country study illustrates that differences might exist between different developed markets.

Table A60: Willingness-to-pay for wine attributes, USA, Canada, France, Germany and UK (N=11,322*: US n = 1,617 and n = 1,614, Canada n = 1,036 and n = 982, France n = 2,027, Germany n = 2,025, UK n = 2,021)

	Average all countries	By country	
	Premium (%)**		Premium (%)**
Social responsibility logo (vs. no logo)	2.3%	France	-3.4%
Environmental		US East coast	10.4%
responsibility logo (vs. no	6.6%	US Midwest	7.3%
logo)		CAN Anglo	8.8%
		UK	3.8%
		France	26.1%
		Germany	27%
Organic logo (vs. no logo)	14.4%	US East coast	17.6%
		US Midwest	10.7%
		CAN Anglo	12.8%
		CAN Franco	2.9%
	3.2%	UK	3.4%
		France	-3.1%
		Germany	-0.3%
Carbon zero logo (vs. no		US East coast	9.6%
logo)		US Midwest	5.2%
		CAN Anglo	4.0%
		CAN Franco	3.3%
		UK	-1.4%
		France	-4.3%
10		Germany	-8.1%
10 per cent less	-2.9%	US East coast	1.2%
		US Midwest	1.7%
		CAN Anglo	-4.6%
		CAN Franco	-4.3%

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Online survey, in 2009Samples in US included New York metropolitan area (Northeast) and Chicago metropolitan area (Midwest); samples in Canada included Anglophone and Francophone Canada ** reported in the study.

Source: Mueller Loose and Remaud, (2013)

A1.5 Other product categories

There has also been a number of CE and other WTP studies conducted for products that do not strictly fit in the previous categories (meat and seafood, dairy, fruit and vegetables, and wine) or include multiple types of food products. The current review includes 8 CE and other WTP studies examining the attributes of other types of food products in Europe and North America. Attributes examined in these studies include organic, local foods, GM production, country-of-origin, product quality, landscape of the place of origin, social responsibility, functional foods, environmental condition and carbon/GHG emissions associated with production.

European studies

The current review includes CE and other WTP studies examining the attributes of other types of food products (almonds, lamb, strawberries, olive oil, honey and chocolate) in Europe, including the markets of Belgium, Italy, Spain and the UK. Attributes examined in these studies include organic, local foods, GM production, country-of-origin, product quality, landscape of the place of origin and social responsibility.

de-Magritis and Gracia (2016) examined Spanish consumers' WTP for almonds with organic and local attributes, including the inclusion of an EU organic label, as well as product labels indicating a series of distances between the production and consumption areas (i.e. food miles) (100km, 800km and 2,000km). Based on a series of preference questions, the authors placed participants in one of three segments: Segment 1 consisted of mostly male and younger participants who positively valued the organic and 100km labels and negatively valued the 2,000km label; Segment 2 consisted of mostly female and older participants who positively valued the organic and 100km labels and negatively valued both the 800km and 2,000km label; Segment 3 consisted of mostly female and older participants who positively valued both the organic and 100km label but negatively valued only the 2,000km label. Average WTP (\notin /package) for each of these attributes across the three segments are presented in Table A61 below. Results show participants in Segment 2 have the highest negative WTP for higher food miles, while participants in Segment 3 have the highest positive WTP for organic and local foods (de-Magritis and Gracia, 2016).

Table A61: Willingness-to-pay for almonds with associated organic and local attributes, Spain (N=171), €/package

	Segment 1	Segment 2	Segment 3
Organic	0.27	0.85	1.22
100km label	0.21	1.18	1.40
800km label	-0.04	-1.01	0.23
2,000km label	-0.32	-1.68	-1.33

Source: de-Magritis and Gracia, 2016.

Arnoult et al. (2010) conducted a cross-product CE, focussing on UK consumers' WTP for COO and related attributes, including origin, season, type (GM or organic) alongside price. The sample size were just under 200 for both products. The WTP results reported in Table A62 indicate strong preferences for local products and an aversion to EU imports for both product types. WTP values were just under £1.94/kilo (or 37%-60% premium of the base price) and approximately -£1.10/kg (-22% and -34%). However, some seasonality differences were observed between product types as the WTP for lamb increased in spring whereas WTP for strawberries increased in summer. Another difference was observed was that while organic strawberries had higher WTP than GM-free berries, WTP was higher for GM-free lamb than organic lamb. Finally, a number of socio-demographic influences were tested, finding that the locality of product was valued higher by higher income people, higher weekly spending influenced WTP for lamb, whereas gender influenced WTP for strawberries over different seasons.

		Lamb		Strawberries	
		WTP £/kg	<i>Premium (%)**</i>	WTP £/kg	Premium (%)**
Location (us	Local	1.75	37%	1.94	60%
Location (vs.	National	-	-	-	-
world)	European Union	-1.06	-22%	-1.11	-34%
Concernelity (up	Summer			0.58	18%
Seasonality (vs.	Autumn	-0.52	-11%	-0.49	-15%
winter season)	Spring	0.31	7%		
Type 1 (vs. nothing stated)	GM-free	0.59	12%	0.40	12%
Type 2 (vs. nothing stated)	Organic	0.29	6%	0.64	20%

Table A62: Willingness-to-pay for lamb and strawberry attributes, UK (N = 185 lamb CE and N = 187 strawberry CE*)

* Face-to-face interviews in 2005.

** Compared to average of the applied price vectors (lamb: £4.74/kg and strawberries: £3.24/kg) Source: Arnoult et al. (2010)

In a Spanish study, de-Magistris and Gracia (2014) used the "food miles" concept as part of the CE where alternatives vary across almonds produced between 100km and 2000km distances, versus no such labelling at all. The survey participants completed two sets of choice sets, where the second one was used for validity checking. In addition, at the end of this process each participant were offered €10 with a *hold-out set* including a purchase option. The estimated WTP values are described in Table A63, which shows positive preferences with WTP of €0.62-€0.68/100g, or a 30-33 per cent premium, towards an organic label and a 100km label. WTP values towards longer distances were negative and increased according to total distance travelled, hence indicating preferences towards more local products.

Table A63: Willingness-to-pay	for almond attributes,	Spain (N = 171*)
-------------------------------	------------------------	------------------

		Avera €/100 g (Premi	ge WTP ; package um %)**
Production method (vs. No label: conventional)	EU organic label	0.62	(30%)
Origin of	100-km label: almonds were produced within 100km (i.e., within province)	0.68	(33%)
information of distance)	800-km label: almonds were produced around 800km (i.e., within Spanish or neighbour regions)	-0.25	(-12%)
	2000-km label: almonds were produced around 2000km (i.e., outside Spain but in Europe)	-1.03	(-49%)

* Random sample of respondents across the capital area of Spain.

** Compared to average of the applied price vector (€2.085/100g) based on the prices in supermarkets at the time.

Source: de-Magistris and Gracia (2014)

Aprile et al. (2012) assessed Italian consumer values for geographical and quality labels in olive oil products. These labels provide a tool to communicate sustainable production or products' value-added qualities. The labels included Protected Designation of Origin (PDO), Protected Geographical Indications (PGI) and organic farming (OF). The results suggested that all of these attributes affected consumer preferences with regards to olive oil product choices. Consumer

WTP, as summarised in Table A64, ranged from ≤ 1.52 up to ≤ 5.60 per litre, being highest for the PDO label with an 86 per cent premium compared with the base price. The second highest WTP was found for the PF label. The authors commented higher WTP for the PDO label than the PGI label may be due to the fact that olive oil produced in the study location is typically PDO-certified.

		WTP €/litre	(Premium %)**
Type of olive oil/quality (vs.	Extra virgin	4.44	(68%)
Furopean OF label (vs. label			
absent)	Present	4.78	(74%)
European geographical	PDO label	5.60	(86%)
indication (vs. label absent)	PGI label	1.52	(23%)

Table A04. Winnighess-to-pay for onve on attributes, italy ($N = 200$
--

* In-store interviews in grocery stores, 2010 in Naples.

** Compared to average of the applied price vector (€6.5/litre).

Source: Aprile et al. (2012)

In another Italian study, Cosmina et al. (2015) assessed consumer preference for honey attributes including product origin, product type, landscape of the place of origin and price. Most respondents (over 90% of the sample) were honey consumers – however, they typically consumed honey products only occasionally. The place of purchase varies between "buying directly from producer" and supermarkets. The result presented in Table A65 are based on the use of a consumer segmentation approach resulting in four consumer classes with similar choice patterns. People in the first class considered only the origin attribute in their choices. The other three classes were labelled as 'environmentally friendly' consumers (35% of the sample), 'pro-intensive production' consumers and 'organic' consumers. As Table A65 shows, environmentally friendly consumers had a WTP of between €4.76 and €3.99 (84 and 70 per cent) for organic and local honey respectively while indicating negative WTP for other attributes, whereas pro-intensive production and organic consumers were willing to pay between €2.54 and €8.30 (45 and 146 per cent respectively) for most attributes, with the type of honey valued the highest in both classes. Overall these WTP values indicate strong preferences towards local and organic attributes in honey with some differences in WTP between consumer segments. Only a small section of respondents (in Class 1) were not willing to pay any premium for any product other than the local product.

Table A65: Willingness-to-pay for honey attributes, Italy (N = 427	*)
--	----

		Class 1 N/A	Class 2 Environmentally friendly	Class 3 Pro-intensive production	Class 4 Organic	
Class probability		19%	35%	19%	27%	
			WTP € (premiun	WTP €/jar (premium %) **		
Geographic origin	Friuli Venezia Giulia	2.88	3.99	4.53	5.41	
(vs. other Italian	(local) Region	(51%)	(70%)	(80%)	(95%)	
regions)	Other countries	-	-6.45 (-114%)	-	-2.54 (-45%)	
Honey crystallisation (vs. semi-solid state)	Liquid (runny) state	-	-4.84 (-85%)	8.30 (146%)	6.70 (118%)	
Organic (vs. no)	Yes	-	4.76 (84%)	6.57 (116%)	6.33 (112%)	
Landscape (vs. Skyscraper hives)	Evocative landscape	-	-	3.69 (65%)	2.54 (45%)	
	Beehives near industrial buildings	-	-1.59 (-28%)	6.74 (119%)	5.23 (92%)	

* Face-to-face interviews, in 2014

** Compared to average of the applied price vector (€5.67/jar).

Source: Cosmina et al. (2015)

Social responsibility attributes have been included in some, but not many, food and beverage choice studies. Vlaeminck et al. (2016) assessed consumer WTP for a Fair Trade (FT) chocolate product in Belgium. This was done using a within-sample test with two separate CEs: a "FTlabel experiment" including the label (FT and Bio-FT), quality & taste, origin of cocoa and price attributes; and a "FT-characteristics experiment" with sub-attributes of FT covering environmental standards, price paid to producers, community investment, working conditions and product price. Half of the sample saw the FT-label CE first, with the other half seeing a reversed order. In this sample, the general purchase habits of FT products in general, if available, was split across (almost) never (approximately 50% of sample), regularly (42%) and always (5%); and only quarter of respondents defined a FT-product correctly. These general results also show that while most people (70%) believed the FT-statement, not everyone care about these issues personally. A summary of the WTP results from the CE analysis is provided in Tables A66 and A67. As shown in Table A66, the results of the FT-label experiment show that consumers valued the FT-label with a positive WTP of €0.84/100g for the standard FT label and \$1.22 for the Bio-FT label. This equates to 207 per cent and 301 per cent premiums, respectively, relative to the standard supermarket price. Average WTP for the FT-label was then compared with different combinations of the FT-characteristics (FT-high, FT-low, BioFThigh and BioFT-low). As shown in Table A67, WTP values for different FT-sub-attributes were between €2.25 and €3.76 (up to 928% premium); hence consumers valued the bundle of FT attributes more than the plain FT labels. The results of the plain FT-label valuation are comparable to the price premium operated in supermarkets indicating that consumer surplus is effectively captured.

Table A66: Willingness-to-pay for chocolate attributes, Belgium (N= 144*)

		CE with a Fair Trade label	
		WTP €/100g	Premium (%)**
Label presence (vs. no	Fair trade label	0.84	(207%)
label)	Bio-Fair trade label	1.22	(301%)

* Face-to-face intercept survey, in 2013.

** Compared to supermarket price of FT chocolate (€0.81/200g or €0.45/100g)

Source: Vlaeminck et al. (2016)

Table A67: Willingness-to-pay for chocolate attributes, Belgium (N= 144*)

	CE with Fair Trade characteristics	
Attribute bundles	WTP (€/200g)	Premium (%)**
FT highest outcomes: EU Environmental standard, price paid to producer, high community investment and frequent controls in working conditions	3.76	(928%)
FT lowest outcomes: EU Environmental standard, average price paid to producer, average community investment and infrequent controls in working conditions	2.54	(627%)
Bio-FT highest outcomes: Organic Environmental standard, fair price paid to producer, high community investment and frequent controls in working conditions	3.47	(857%)
Bio-FT lowest outcomes: Organic Environmental standard, average price paid to producer, average community investment and infrequent controls in working conditions	2.25	(556%)

* Face-to-face intercept survey, in 2013.

** Compared to supermarket price of FT chocolate (€0.81/200g or €0.45/100g)

Source: Vlaeminck et al. (2016)

Boccia et al. (2019) conducted a number of choice experiments to examine Italian consumer preferences and WTP for brand, corporate environmental and social responsibility programme participation in relation to ready-meal products. Results indicated approximate WTP for the inclusion of these attributes, with participants willing to pay a \pounds 2.46 premium for products with recognisable brand names that also participate in the above programmes. In addition, participants were willing to pay a \pounds 1.53 premium for products participating in environmentally friendly social responsibility programmes, while they were only willing to pay a \pounds 0.19 premium for only social responsibility programme participation (Boccia et al., 2019). A summary of these results is shown in Table A68 below.

Table A68. Willingness-to-pay for ready meal attributes, Italy (N = 1,083)

Attribute bundles	WTP (€/product)
Brand (well-known/recognised vs unknown);	
environmental programme participation; social	2.45895
responsibility participation programme	
Environmental programme participation; social	1 52960
responsibility participation programme	1.52800
Social responsibility participation programme	0.19325

Source: Boccia et al., 2019

North American studies

The current review includes CE and other WTP studies examining the attributes of other types of food products (canola oil and coffee) in North America (US and Canada). Attributes examined in these studies include organic, GM production, country-of-origin, social responsibility, functional foods, environmental condition and carbon/GHG emissions associated with production.

A comparison of GM (or genetically engineered (GE)) products and associated healthenhancing (or functional food) benefits were explored by Ding et al. (2015) in Canada. In this study, consumer preferences for GM-food were linked with consumer trust (generalized trust and trust in the food system) and health-related beliefs. In the context of canola oil products, the selected attributes covered GM or GE information, omega-3 content, COO and price. Consumer trust and health beliefs (i.e. health locus of control (HLC)) were measured in Likertscale statements. The results in Table A69 show that consumers were willing to pay a premium of between 12 and 29 per cent of the base price for domestic and/or regular/enhanced omega-3 levels over no label. However, this WTP was relatively lower compared to the perceived disutility, or required compensation, from the negative WTP associated with GM products. A further analysis with the interactions show (WTP not reported here) that stronger health concerns will increase WTP for enhanced omega-3, and that negative preferences of GM food can be offset or linked to trust. Some additional findings included that men valued GM products more than women, older people and those with higher education were less likely to prefer GM products, and that people with higher income valued health benefits more.

		WTP CAN \$/1 litre	Premium (%)**
Omega-3 content (vs. no	Contains omega-3	0.95	19%
label)	Enhanced omega-3	0.86	17%
Country of origin (vs. USA)	of origin (vs. USA) Canada		29%
GM (vs. no label	label Non-GM		12%
information)	Contains GM/GE	-1.82	-36%

Table A69: Willingness-to-pay for canola oil attributes, Canada (N = 1,009*)

* Nationwide online survey

** Compared to average of the applied price vector (\$5 per 1 liter)

Source: Ding et al. (2015)

Van Loo et al. (2015) focused on consumer preferences for sustainability certification of coffee products. The sustainability labels considered were Fair Trade (FT), Rainforest Alliance, USDA Organic and carbon footprint, the latter of which is less common in the US coffee market. A novelty in the study was a focus on visual attention on the choice sets (coffee packages) by respondents. This was done by an eye-tracking exercise on areas of interest (AOI) using a tracking device connected to the computer used to complete the surveys. From this, two measures were calculated - time and count of total fixation. In addition, Likert-scales were used to explore participants' attitudes to and perceived importance of the sustainability concepts. Three consumer segments were discovered based on the cluster analysis³: 'indifferent', 'sustainability and price conscious' and "price-oriented" consumers. Relative WTP values presented in Table A70 show that respondents, on average, were willing to pay the most (\$1.16/12oz, or 16% premium) for USDA certified coffee, and up to a 19 per cent premium for 'sustainability and price conscious' consumers, which included most of the

³ Using the variables from the Likert scale questions and eye-tracking attention scores.

sample. The results also showed that visual attention to attributes is related to preferences for attributes whereby taking more time and fixating more attention on a particular attribute related to higher WTP. Significant interactions with participants' attention included USDA organic, Fair Trade and price attributes. Hence this study illustrated that *sustainability-motivated* consumers are also likely to seek information about sustainability credentials.

	By consumer segment: Full sample Sustainability and price conscious (n = 47) WTP \$/12 oz Premium (%)** WTP \$/12 oz Premium (%)**		By consumer segments***		
			Sustainability and price conscious (n = 47)		Price-oriented (n = 26)
Fair Trade – label (vs. label not present)	0.68	(9%)	0.71	(10%)	-
Rainforest Alliance – label (vs. label not present)	0.84	(12%)	0.99	(14%)	-
USDA Organic – label (vs. label not present)	1.16	(16%)	1.41	(19%)	-
Carbon Footprint – label (vs. label not present)	-		0.51	(7%)	-

Table A70: Willingness-to-pay for coffee attributes, USA (N = 81*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Participants were recruited from a University database, in 2013.

** Compared to average of the applied price vector (\$7.30/12 oz)

*** Since the "Indifferent consumer" segment consisted of only 8 participants, no WTP was calculated. Source: Van Loo et al. (2015)

A1.6 Products adopting new technology

Finally, some studies have considered the opportunities provided by technological advancements in relation to food choices. The current review includes CE and other WTP studies examining the attributes of food products adopting new technology in Europe (UK) and North America (US and Canada). Attributes examined in these studies include nanotechnology, animal welfare, food safety, traceability, country-of-origin, GM production, functional foods, environmental condition and taste.

European studies

Erdem (2015) explored UK consumers' preferences for reduced food safety risk in chicken products. The authors tested the impact of incorporating nanotechnology into food product packaging by including this attribute (as a symbol) in one CE and not in the other. Other attributes of consideration were risk of food poisoning and animal welfare level (based on the Welfare Quality index). Each subsample was further split into "welfare-improved" chicken consumers and "conventional" chicken consumers according to their reported purchasing behaviour⁴. Other than the nanotech attribute, the levels used in the status quo option varied according to purchasing behaviour. As Table A71 shows, consumers on average preferred chicken with a lower food safety risk and improved animal welfare, regardless of the presence

⁴ Approximately 30% of the respondents in both samples were welfare-improved chicken consumers.

of nanotechnology. WTP values were found to be higher for the "welfare-improved" consumers compared with "conventional" consumers. It also appeared that the presence of nanotechnology could increase WTP for food safety and chicken welfare. A choice debriefing question revealed that around half of the respondents considered the inclusion of such nanotechnology to be "a good idea", with the remaining responses varying from "not bothered" to "more than concerned".

		Nano treatment (n = 225)		Non-nano treatment (n = 224)	
	Consumer type	WTP (£/chicken)	Premium (%) **	WTP (£/chicken)	Premium (%) **
Food poisoning risk:	Conventional	-0.30	(-10%)	-0.30	(-3%)
Reduction from a baseline	Welfare-improved	-0.59	(-20%)	-0.52	(-5%)
Chicken welfare level (scale	Conventional	0.09	(3%)	0.08	(1%)
0-100)	Welfare-improved	0.67	(22%)	0.51	(5%)

Table A71: Willingness-to-pay for chicken attributes, UK (N = 449*)

* Online survey, in 2010

** Compared to average price (around £3/chicken). Source: Erdem (2015)

North American studies

Lilavanichakul and Boecker (2013) explored Canadian consumer acceptance of traceability technology in ginseng products. This was explored amongst trade-offs with the products origin and manufacturer attributes. As summarised in Table A72, estimated WTP values implied a 16 per cent premium of the base price (\$2.78/bottle) for having an internal tag for traceability/quality assurance. However, this WTP was relatively lower than for the inclusion of a Guarantee label or Canadian Ginseng product. The negative interaction term with a WTP of -\$1.67/bottle for the simultaneous use of the 'Canadian Guaranteed' and 'Product of Canada' labels suggest that these attributes could be seen as substitutes.

Table A72: Willingness-to-pay for ginseng product attributes, Canada (N = 1,647*)

		WTP (\$/bottle with	Premium
		60 capsules)	(%)**
Internal tag (vs. no)	Yes	2.78	(16%)
Manufacturer (vs. Ontario Association of	National Manufacturer	2.24	(140/)
Ginseng Producers)	Brand	-2.34	(-14%)
Canadian Ginseng Guaranteed (vs. no)	Yes	9.52	(56%)
Product of Canada (vs. no)	Yes	5.74	(34%)
Canadian Ginseng Guaranteed* Product		1.67	(100/)
of Canada		-1.07	(-10%)

* Nationwide online survey

** Compared to average of the applied price vector (\$16.99/bottle)

Source: Lilavanichakul and Boecker (2013)

In the third new-technology orientated CE, Yue et al. (2015) explored US consumer preferences for nano- and GM-food in the context of a rice product. The CE considered the possible benefits (e.g. better food safety) that these technologies could provide. The data was analysed using a class based approach from which four distinct consumer groups, based on their choices and characteristics (gender, income, education, race/ethnicity, and political and religious associations), were identified (see Table A73). Most respondents were in the *'benefit orientated group'* with a likelihood of 40 per cent for participants to belong to this group.

Across all groups, new technologies had a negative WTP, varying between -2 and -89 percent of the base price, thus the conventional production method was preferred. The most valued benefits varies across consumer groups. *'Price oriented'* consumers were willing to pay the most for the enhanced nutritional elements (an approximate 10 per cent premium) and no extra for improved taste or environmental impacts when compared to the provision of no additional benefits. The remaining three groups were willing to pay most for improved food safety, (premiums of between 9 and 136 per cent), with the *'benefit oriented'* group indicating the highest WTP. These results imply that consumers express highly heterogeneous preferences when distinguished by their choices and consumer characteristics. While new technologies had negative WTP values, the attached benefits were valued differently across the groups. Thus consumer preferences towards nanotechnology can include a complex set of trade-offs.

		Class 1***	Class 2***	Class 3***	Class 4***
		Price	Technology	Benefit	New
		oriented	averse	oriented	technology
					rejecters
Class	probability	18%	17%	40%	25%
			WTP (S	\$/lb)	
			premium	ı (%)**	
Production	Nanotechnology	-0.09	-0.70	-0.94	-3.39
technology		(-2%)	(-16%)	(-21%)	(-77%)
(vs.	GM	-0.1	-0.78	-1.06	-3.9
conventional)		(-2%)	(-18%)	(-24%)	(-89%)
	Enhanced nutrition	0.42	0.21	5.16	0.56
Benefit		(10%)	(5%)	(118%)	(13%)
from using the	Improved taste	-	0.33	2.99	0.56
given			(8%)	(68%)	(13%)
technology	Improved food	0.22	0.39	5.96	1.10
(vs. no	safety	(5%)	(9%)	(136%)	(25%)
additional	Less harmful	-	-	4.08	0.37
benefit)	environmental			(93%)	(8%)
	impact during				
	production				

Table A73: Willingness-to-pay for (a bag of) white rice attributes: The latent class approach, USA (N = 1,117*)

Note: In this adapted Table, WTP was included only if the attribute was statistically significant.

* Online survey, in 2013

** Compared to average of the applied price vector (\$\$4.375/lb)

***Statistically significant class determinants: Class 1 reference group; Class 2 Gender; Class 3 Education, Gender, Income, Religion, Politics; Class 4 Gender, Religion

Source: Yue et al. (2015)

A1.7 Summary

In conclusion, this review included 69 international CE and other WTP studies regarding food and beverage choices and associated credence attributes from 2010 to 2019. This complements and updates previous reviews (Miller et al., 2014; Saunders et al., 2016) with the inclusion of more recent studies. Most of the studies reviewed pertained to meat and seafood products (35), following by wine (8), fruit and vegetable (8) and dairy products (6). Another 12 studies were reviewed in other product contexts (e.g. coffee and chocolate) or food products adopting new technology to communicate food safety or traceability. Most studies examined consumer preferences, typically targeting regular purchasers of the type of product examined; although one study included a comparison between food retailers and food consumers about their preferences towards the use of organic ingredients (Probst et al., 2012).

Appendix A: References

Aichner, T., Forza, C. and Trentin, A. (2017). The country-of-origin lie: Impact of foreign branding on customers' willingness to buy and willingness to pay when the product's actual origin is disclosed. The International Review of Retail, Distribution and Consumer Research, 27(1); 43-60.

Akaichi, F., de Grauw, S. and Darmon, P. (2015). Are Fair Trade, carbon footprint and organic attributes competing? Some evidences from Scotland, Netherland and France. Paper presented at the International Association of Agricultural Economists Conference, August 9-14, 2015, Milan, Italy.

Arnoult, M., Lobb, A., and Tiffin, R. (2010) Willingness to Pay for Imported and Seasonal Foods: A UK Survey. Journal of International Food & Agribusiness Marketing, 22(3-4); 234-251.

Balcombe, K., Bradley, D., Fraser, I. and Hussein, M. (2016). Consumer preferences regarding country of origin for multiple meat products. Food Policy, 64; 49-62.

Bechtold, K.-B. and Abdulai, A. (2014). Combining attitudinal statements with choice experiments to analyse preference heterogeneity for functional dairy products. Food Policy, 47; 97-106.

Blare, T., Donovan, J. and del Pozo, C. (2017). Estimates of the willingness to pay for locally grown tree fruits in Cusco, Peru. Renewable Agriculture and Food Systems, 2017; 1-12.

Boccia, F., Manzo, R.M. and Covino, D. (2019). Consumer behaviour and corporate social responsibility: An evaluation by a choice experiment. Corporate Social Responsibility and Environmental Management, 26 (2019); 97-105.

Byrd, E.S., Widmar, N.J.O. and Wilcox, M.D. (2017). Are consumer willing to pay for local chicken breasts and pork chops? Journal of Food Products Marketing, 0(2017); 1-14.

Calvo Dopico, D., Mendes, R., Silva, H.A., Verrez-Bagnis, V., Perez-Martin, R. and Sotelo, C.G. (2016). Evaluation, signalling and willingness to pay for traceability: A cross-national comparison. Spanish Journal of Marketing – ESIC, 20(2); 93-103.

Carlsson, F., Frykblom, P., and Lagerkvist, C.J. (2005). Consumer preferences for food product quality attributes from Swedish agriculture. AMBIO: A Journal of the Human Environment, 34(4); 366-370.

Ceschi, S., Canavari, M. and Castellini, A. (2018). Consumer's Preference and Willingness to Pay for Apple Attributes: A Choice Experiment in Large Retail Outlets in Bologna (Italy). Journal of International Food and Agribusiness Marketing, 30(4); 305-322.

Chalak, A. and Abiad, M. (2012). How effective is information provision in shaping food safety related purchasing decisions? Evidence from a choice experiment in Lebanon. Food Quality and Preference, 26(1); 81-92.

Chung, C., Briggeman, B.C. and Han, S. (2012). Willingness-to-pay for beef quality attributes: A latent segmentation analysis of Korean grocery shoppers. Journal of Agricultural and Applied Economics, 44(4); 447-459.

Cicia, G., Cembalo, L., Del Giudice, T. and Scarpa, R. (2013). Country-of-Origin Effects on Russian Wine Consumers. Journal of Food Products Marketing, 19(4); 247-260.

Clark, B., Stewart, G.B., Panzone, L.A., Kyriazakis, I. and Frewer, L.J. (2017). Citizens, consumers and farm animal welfare: A meta-analysis of willingness-to-pay studies. Food Policy, 68; 112-127.

Cosmina, M., Gallenti, G., Marangon, F. and Troiano, S. (2015). Attitudes towards honey among Italian consumers: A choice experiment approach. Paper prepared for presentation at the EAAE-AAEA Joint Seminar 'Consumer Behavior in a Changing World: Food, Culture, Society" March 25 to 27, 2015 Naples, Italy.

de-Magistris, T. and Gracia, A. (2014). Do consumers care about organic and distance labels? An empirical analysis in Spain. International Journal of Consumer Studies, 38(6); 660-669.

de-Magritis, T. and Gracia, A. (2016). Consumers' willingness-to-pay for sustainable food products: The case of organically and locally grown almonds in Spain. Journal of Cleaner Production, 118; 97-104.

Denver, S. and Jensen, J.D. (2014). Consumer preferences for organically and locally produced apples. Food Quality and Preference, 31; 129-134.

Denver, S., Sandoe, P. and Christensen, T. (2017). Consumer preferences for pig welfare – can the market accommodate more than one level of welfare pork? Meat Science, 129; 140-146.

Ding, Y., Veeman, M.M. and Adamowicz, W.L. (2015). Functional food choices: Impacts of trust and health control beliefs on Canadian consumers' choices of canola oil. Food Policy, 52; 92-98.

Erdem, S. (2015). Consumers' Preferences for Nanotechnology in Food Packaging: A Discrete Choice Experiment. Journal of Agricultural Economics, 66(2); 259–279.

Feucht, Y. and Zander, K. (2017). Consumers' willingness to pay for climate-friendly labelled food in Europe. 11th Igls-Forum, 13-17 February.

Gao, Z., Yu, X., Li, C., McFadden, B.R. (2019). The interaction between country of origin and genetically modified orange juice in urban China. Food Quality and Preference, 71 (2019); 475-484.

Gracia, A. (2014). Consumers' preferences for a local food product: A real choice experiment. Empirical Economics, 47; 111-128.

Grashuis, J. and Magnier, A. (2018). Product differentiation by marketing and processing cooperatives: A choice experiment with cheese and cereal products. Agribusiness, 34 (2018); 813-830.

Grebitus, C., Peschel, A.O. and Hughner, R.S. (2018). Voluntary food labelling: The additive effect of "free from" labels and region of origin. Agribusiness, 34 (2018); 714-727.

Hanley, N., Shogren, J.F. and White, B. (2013). Introduction to environmental economics. Oxford: Oxford University Press.

Hempel, C. and Hamm, U. (2016). How important is local food to organic-minded consumers? Appetite, 96; 309-318.

Hensher, D.A., Rose, J.M. and Greene, W.H. (2015). Applied choice analysis (2nd Ed.). Cambridge University Press: UK.

Kallas, Z., Borrisser-Pairó, F., Martínez, B., Vieira, C., Rubio, B., Panella, N., ..., Gil, J.M. (2015). The impact of the sensory experience on scale and preference heterogeneity: The GMNL model approach applied to pig castration and meat quality. Paper prepared for presentation at the EAAE-AAEA Joint Seminar 'Consumer Behavior in a Changing World: Food, Culture, Society. March 25-27, Naples, Italy.

Kallas, Z., Escobar, C. and Gil, J.M. (2013). Analysis of consumers' preferences for a specialoccasion red wine: A dual response choice experiment approach. Food Quality and Preference, 30(2); 156-168.

Kallas, Z., Vitale, M. and Gil, J.M. (2019). Health Innovation in Patty Products: The Role of Food Neophobia in Consumers' Non-Hypothetical Willingness to Pay, Purchase Intention and Hedonic Evaluation. Nutrients, 11 (2019); doi:10.3390/nu11020444.

Lagerkvist, C.J., Berthelsen, T., Sundström, K. and Johansson, H. (2014).Country of origin or EU/non-EU labelling of beef? Comparing structural reliability and validity of discrete choice experiments for measurement of consumer preferences for origin and extrinsic quality cues. Food Quality and Preference, 34; 50-61.

Lagerkvist, C.J., Hess, S. and Johansson, H. (2017). How Much Credence Does It Take? Evidence on the Trade-Off between Country-Of-Origin Information and Credence Attributes for Beef from a Choice Experiment in Sweden. Foods, 84(6); 1-19.

Lai, J., Wang, H.W, Ortega, D.L. and Widmar, N.J.O. (2018). Factoring Chinese consumers' risk perceptions into their willingness to pay for pork safety, environmental stewardship, and animal welfare. Food Control, 85 (2018); 423-431.

Li, X., Jensen, K.L., Clark, C.D. and Lambert, D.M. (2016). Consumer willingness to pay for beef grown using climate friendly production practices. Food Policy, 64; 93-106.

Lilavanichakul, A. and Boecker, A. (2013). Consumer Acceptance of a New Traceability Technology: A Discrete Choice Application to Ontario Ginseng. International Food and Agribusiness Management Review, 16(4); 25-50.

Lim, K.H., Hu, W., Maynard, L.J. and Goddard, E. (2014). A taste for safer beef? How much does consumers' perceived risk influence willingness to pay for country-of-origin labeled beef. Agribusiness, 30(1); 17-30.

Merritt, M.G., Delong, K.L., Griffith, A.P. and Jensen, K.L. (2018). Consumer willingness to pay for Tennessee Certified Beef. Journal of Agricultural and Applied Economics, 50(2); 233-254.

Miller, S., Driver, T., Velasquez, N. and Saunders, C. (2014). Maximising Export Returns (MER): Consumer behaviour and trends for credence attributes in key markets and a review of how these may be communicated. AERU Research Report No. 332, July 2014. Agribusiness & Economics Research Unit (AERU), Lincoln University: Lincoln, New Zealand.

Mueller, S., Lockshin, L., Saltman, Y. and Blanford, J. (2010). Message on a bottle: The relative influence of wine back label information on wine choice. Food Quality and Preference, 21(1); 22-32.

Mueller Loose, S. and Remaud, H. (2013). Impact of corporate social responsibility claims on consumer food choice: A cross-cultural comparison. British Food Journal, 115(1); 142-166.

Mugera, A., Burton, M. and Downsborough, E. (2017). Consumer Preference and Willingness to Pay for a Local Label Attribute in Western Australian Fresh and Processed Food Products. Journal of Food Products Marketing, 23(4); 452-472.

O'Brien, K.A. and Teisl, M.F. (2004). Eco-information and its effect on consumer values for environmentally certified forest products. Journal of Forest Economics, 10; 75-96.

Ortega, D.L., Wang, H.H., Wu, L., and Hong, S.J. (2015). Retail channel and consumer demand for food quality in China. China Economic Review, 36; 359-366.

Ortega, D.L., Wang, H.H. and Olynk Widmar, N.J. (2014). Aquaculture imports from Asia: an analysis of U.S. consumer demand for select food quality attributes. Agricultural Economics, 45(5); 625-634.

Paci, F., Danza, A., Del Nobile, M.A. and Conte, A. (2018). Consumer acceptance and willingness to pay for fresh fish-burger: A choice experiment. Journal of Cleaner Production, 172 (2018); 3128-3137.

Poelmans, E. and Rousseau, S. (2016). How do chocolate lovers balance taste and ethical considerations? British Food Journal, 118(2); 343-361.

Pomarici, E., Asioli, D., Vecchio, R. and Naes, T. (2018). Young consumers' preferences for water-saving wines: An experimental study. Wine Economics and Policy, 7 (2018); 65-76.

Probst, L., Houedjofonon, E., Ayerakwa, H.M. and Haas, R. (2012). Will they buy it? The potential for marketing organic vegetables in the food vending sector to strengthen vegetable safety: A choice experiment study in three West African cities. Food Policy, 37(3); 296-308.

Risius, A. and Hamm, U. (2017). The effect of information on beef husbandry systems on consumers' preferences and willingness to pay. Meat Science, 124; 9-14.

Saunders, C., Guenther, M., Tait, P. and Saunders, J. (2013). Assessing consumer preferences and willingness to pay for NZ food attributes in China, India and the UK. Contributed Paper prepared for presentation at the 87th Annual Conference of the Agricultural Economics Society, University of Warwick, United Kingdom.

Saunders, C., Driver, T., Mowat, A., Kaye-Blake, W., Payn, T., Bayne, K., Saunders, J. ... Tait, P. (2016). Driving better programme investment and accelerating challenge impact through a prioritisation matrix of international and national perspectives. Retrieved 7th November 2017 from http://www.ourlandandwater.nz/the-challenge/greater-value-in-global-markets.

Schaufele, I. and Hamm, U. (2017). Consumers' perceptions, preferences and willingness-topay for wine with sustainability characteristics: A review. Journal of Cleaner Production, 147; 379-394. Sellers, R. (2016). Would you pay a price premium for a sustainable wine? The voice of the Spanish consumer. Agriculture and Agricultural Science Procedia, 8; 10-16. Florence "Sustainability of Well-Being International Forum". 2015: Food for Sustainability and not just food, FlorenceSWIF2015.

Tait, P., Saunders, C., Guenther, M. and Rutherford, P. (2016). Emerging versus developed economy consumer willingness to pay for environmentally sustainable food production: A choice experiment approach comparing Indian, Chinese and United Kingdom lamb consumers. Journal of Cleaner Production, 124; 65-72.

Teratanavat, R., and Hooker, N.H. (2006). Consumer valuations and preference heterogeneity for a novel functional food. Food Science, 71(7); S533–S541.

Ubilava, D., Foster, K.A., Lusk, J.L. and Nilsson, T. (2011). Differences in consumer preferences when facing branded versus non-branded choices. Journal of Consumer Behaviour, 10(2), 61-70.

Uchida, H., Onozaka, Y., Morita, T. and Managi, S. (2014). Demand for ecolabeled seafood in the Japanese market: A conjoint analysis of the impact of information and interaction with other labels. Food Policy, 44; 68–76.

Van Loo, E.J., Caputo, V., Nayga Jr., R.M., Meullenet, J.-F. and Ricke, S.C. (2011). Consumers' willingness to pay for organic chicken breast: Evidence from choice experiment. Food Quality and Preference, 22(7); 603-613.

Van Loo, E.J., Caputo, V., Nayga Jr., R.M., Seo, H.-S., Baoyue Zhang, B. and Verbeke, W. (2015). Sustainability labels on coffee: Consumer preferences, willingness-to-pay and visual attention to attributes. Ecological Economics, 118; 215-225.

Van Loo, E.J., Caputo, V., Nayga Jr., R.M. and Verbeke, W. (2014). Consumers' valuation of sustainability labels on meat. Food Policy, 49; 137-150.

Van Wezemael, L., Caputo, V., Nayga Jr, R.M., Chryssochoidis, G. and Verbeke, W. (2014). European consumer preferences for beef with nutrition and health claims: A multi-country investigation using discrete choice experiments. Food Policy, 44; 167-176.

Viegas, I., Nunes, L.C., Madureira, L., Fontes, M.A. and Santos, J.M. (2014). Beef Credence Attributes: Implications of Substitution Effects on Consumers' WTP. Journal of Agricultural Economics, 65(3); 600-615.

Vlaeminck, P., Vandoren, J. and Vranken, L. (2016). Consumers' willingness to pay for Fair Trade chocolate. In The Economics of Chocolate by M.P. Squicciarini and J. Swinnen (Eds.) pp. 180-191. Oxford, UK: Oxford University Press.

Wang, J., Ge, J. and Ma, Y. (2018). Urban Chinese Consumers' Willingness to Pay for Pork with Certified Labels: A Discrete Choice Experiment. Sustainability, 10 (2018); doi:10.3390/su10030603.

Wongprawmas, R. and Canavari, M. (2017). Consumers' willingness-to-pay for food safety labels in an emerging market: The case of fresh produce in Thailand. Food Policy, 69; 25-34.

Wu, L., Yin, S., Xu, Y. and Zhu, D. (2014). Effectiveness of China's organic food certification policy: Consumer preferences for infant milk formula with different organic certification labels. Canadian Journal of Agricultural Economics, 62(4); 545-568.

Wu, L., Wang, S., Zhu, D., Hu, W. and Wang, H. (2015). Chinese consumers' preferences and willingness to pay for traceable food quality and safety attributes: The case of pork. China Economic Review, 35; 121-136.

Wu, L., Wang, H., Zhu, D. Hu, W. and Wang, S. (2016). Chinese consumers' willingness to pay for pork traceability information – the case of Wuxi. Agricultural Economics, 47(1); 71-79.

Xu, P and Zeng, Y.C. (2014). Factors that affect willingness to pay for red wines in China. Journal of International Consumer Marketing, 26(5); 426-439.

Xu, P., Zeng, Y.C., Song, S. and Lone, T. (2014). Willingness to pay for red wines in China. Journal of Wine Research, 25(4); 265-280.

Yue, C., Zhao, S. and Kuzma, J. (2015). Heterogeneous Consumer preferences for nanotechnology and genetic-modification technology in food products. Journal of Agricultural Economics, 66(2); 308–328.

Zanoli, R., Scarpa, R., Napolitano, F., Piasentier, E. Naspetti, S. and Bruschi, V. (2013). Organic label as an identifier of environmentally related quality: A consumer choice experiment on beef in Italy. Renewable Agriculture and Food Systems, 28(1); 70-79.

Zhang, C., Bai, J. and Wahl, T.I. (2012). Consumers' willingness to pay for traceable pork, milk, and cooking oil in Nanjing, China. Food Control, 27(1); 21-28.

Zou, N.-N. and Hobbs, J.E. (2010). The role of labelling in consumers' functional food choices. Paper prepared for presentation at the 1st Joint EAAE/AAEA Seminar "The Economics of Food, Food Choice and Health" Freising, Germany, September 15–17, 2010. Appendix B: Survey Instrument

Our Land and Water Science Challenge - Survey

Our Land and Water Science Challenge

The Drivers Project

Q1

Welcome to Our Land and Water Science Challenge survey.

We would really welcome your opinion on the international and domestic issues that have the potential to influence land use change/practice in New Zealand. The results you provide will feed into the research planning for the next phase of the Our Land and Water National Science Challenge.

This survey takes about 5 minutes to complete. You are free to skip any question or stop the survey at any time. If you do stop the survey before the end, the information you have provided will not be used. This survey is being conducted by the Agribusiness and Economics Research Unit (AERU) at Lincoln University in New Zealand.

The lead researcher is Prof Caroline Saunders. If you have any questions or concerns about the research, you may contact her at:Caroline.Saunders@lincoln.ac.nz

To begin the survey, begin by clicking on the >> button below.

Regards,

Caroline

Page Break

Q2 Key issues: What do you see as the three most critical *international* issues which have the potential to influence New Zealand land use change/practice or land management practices?

О	1	(Most critical)
О	2	
Ο	3	

Q3: **Key Issues:** What do you see as the three most critical *domestic* issues which have the potential to influence **New Zealand land use change/practice or land management practices?**

0	1	(Most critical)
Ο	2	
Ο	3	

Page Break —

Q4: International Issues

Below are some key issues that stakeholders and the team have previously identified.

Please indicate whether you think the following **international** issues and drivers will have a high, medium or low impact on **New Zealand land use change/practice or land management practices** over the coming decade.

	High (1)	Medium (2)	Low (3)	Don't know (4)
Agricultural policy (1)	О	О	О	Ο
Air quality (2)	О	О	О	Ο
Animal health and welfare (3)	0	О	0	0
Authentication/traceability (4)	О	О	0	0
Biodiversity (5)	О	О	О	О
Biosecurity (6)	О	О	О	O
Brand (7)	О	О	О	Ο
Chemical residues (8)	Ο	О	О	О
Condition of the environment (9)	0	О	0	0
Country-of-Origin (10)	О	О	О	Ο
Cultural values (11)	О	О	О	О
Demographics (12)	О	О	О	Ο
Digital communications systems (13)	Ο	Ο	O	0

	High (1)	Medium (2)	Low (3)	Don't know (4)
Emissions trading (14)	O	0	O	O
Extreme weather events (15)	0	О	0	0
Fair trade (16)	0	0	0	0
Family and community values (17)	O	0	O	O
Food safety (18)	0	Ο	0	0
Functional foods (19)	0	0	O	0
GM and nanotechnology (20)	0	0	0	0
Greenhouse gas emissions (21)	0	O	O	0
Health and safety (22)	O	O	O	O
Innovative products and services (23)	0	0	0	0
Local foods/food miles (24)	0	О	0	O

Page Break

	High (1)	Medium (2)	Low (3)	Don't know (4)
Māori values (25)	О	О	О	О
Nitrate limits (26)	0	0	0	0
Organic production (27)	О	О	О	О
Pasture-based production (28)	0	0	0	0
Product quality (29)	0	0	0	0
Religion (30)	0	0	0	0
Soil quality (31)	О	0	0	0
Sustainable supply (32)	0	0	0	0
Trade agreements (33)	0	0	0	0
Trade effects (34)	0	0	0	0
Waste/recycling (35)	0	0	0	0
Water footprinting/use (36)	0	0	0	0
Water quality (37)	0	0	0	0

Page Break -

Q5: Domestic

Please indicate whether you think the following **domestic** issues and drivers will have a high, medium or low impact on **New Zealand land use change/practice or land management practices** over the coming decade.

	High (1)	Medium (2)	Low (3)	Don't know (4)
Agricultural policy (1)	О	О	О	О
Air quality (2)	Ο	О	О	О
Animal health and welfare (3)	О	О	О	О
Authentication/traceability (4)	0	О	0	О
Biodiversity (5)	О	О	Ο	О
Biosecurity (6)	Ο	О	O	0
Brand (7)	О	О	Ο	О
Chemical residues (8)	Ο	О	О	О
Climate change (9)	О	О	O	О
Condition of the environment (10)	О	О	О	О
Country-of-Origin (11)	Ο	О	Ο	О
Cultural values (12)	О	О	О	О
Demographics (13)	Ο	О	O	Ο
Digital communications systems (14)	О	О	О	О

	High (1)	Medium (2)	Low (3)	Don't know (4)
Emissions trading scheme (ETS) (15)	O	0	0	0
Extreme weather events (16)	0	0	0	0
Fair trade (17)	0	0	0	0
Family and community values (18)	0	0	0	0
Food safety (19)	0	0	0	0
Functional foods (20)	0	0	0	0
GM and nanotechnology (21)	0	0	0	0
Greenhouse gas emissions (22)	0	O	0	0
Health and safety (23)	O	O	O	O
Innovative products and services (24)	0	0	0	0
Local foods/food miles (25)	О	О	O	О

Page Break

	High (1)	Medium (2)	Low (3)	Don't know (4)
Māori values (26)	О	О	О	О
Nitrate limits (27)	0	0	0	О
Organic production (28)	0	0	0	0
Pasture-based production (29)	0	0	0	О
Product quality (30)	O	O	0	О
Religion (31)	0	0	0	О
Soil quality (32)	О	Ο	О	О
Sustainable supply (33)	0	0	0	0
Trade agreements (34)	O	O	O	О
Trade effects (35)	0	0	0	О
Waste/recycling (36)	0	Ο	O	0
Water footprinting/use (37)	0	0	0	0
Water quality (38)	О	Ο	0	Ο

Page Break -

Q6: Which primary sector are you most aligned with:

- **O** Meat (1)
- O Dairy (2)
- \bigcirc Wool (3)
- O Viticulture/Wine (4)
- **O** Horticulture (5)
- O Forestry (6)
- O Aquaculture (7)
- O Government (8)
- O Māori enterprise (9)
- O Science/Research (10)
- **O** Extension work (11)
- Smart agriculture (12)
- O Other (please specify) (13)

Page Break

	Very knowledgeable (1)	knowledgeable (2)	Some knowledge (3)	Little knowledge (4)	No knowledge (5)
North America (Canada, USA, Mexico) (1)	O	0	0	0	O
China (2)	0	0	Ο	О	O
South East Asia (Vietnam, Thailand, Cambodia, Indonesia, Malaysia, Myanmar) (3)	0	O	0	0	0
Japan (4)	0	0	О	О	О
South Korea (5)	0	Ο	0	О	0
European Union (6)	O	0	0	0	O
Other European countries (7)	0	0	0	0	0
United Kingdom (8)	0	О	0	0	0
Other (Please specify): (9)	O	О	O	О	O

Q7: What level of knowledge do you have concerning the following markets/regions:

Q8: Please indicate the extent of your experience in the following areas:

	Extensive (1)	High (2)	Moderate (3)	Some (4)	None (5)
International markets (1)	O	0	О	О	0
Environmental policy (2)	0	0	О	О	O
R&D/innovation (3)	0	O	0	О	0
Trade policy (4)	O	О	О	О	О
Other domestic (5)	0	0	О	О	O

Thank you!

Thank you for your contribution to our research!

We value the time and contribution you have made to setting the direction of this National Science Challenge. If you have any queries, please contact:

Professor Caroline Saunders Caroline.Saunders@lincoln.ac.nz

End of Block: Default Question Block